

## **Durham E-Theses**

Arabic grammatical studies in the late 9th and early 10th centuries, with particular reference to the Usūl al-Nahw of Ibn al-Sarrāj

Cassels, David Andrew

### How to cite:

Cassels, David Andrew (1979) Arabic grammatical studies in the late 9th and early 10th centuries, with particular reference to the  $Us\bar{u}l$  al-Nahw of Ibn al-Sarr $\bar{a}j$ , Durham theses, Durham University. Available at Durham E-Theses Online: http://etheses.dur.ac.uk/7911/

#### Use policy

The full-text may be used and/or reproduced, and given to third parties in any format or medium, without prior permission or charge, for personal research or study, educational, or not-for-profit purposes provided that:

- ullet a full bibliographic reference is made to the original source
- a link is made to the metadata record in Durham E-Theses
- the full-text is not changed in any way

The full-text must not be sold in any format or medium without the formal permission of the copyright holders.

Please consult the full Durham E-Theses policy for further details.

# ARABIC GRAMMATICAL STUDIES IN THE LATE 9TH AND EARLY 10TH CENTURIES, WITH PARTICULAR REFERENCE TO THE USUL AL-MAHU OF IBN AL-SARRĀJ

BY

DAVID ANDRES CASSELS

Thosis submitted to the Peculty of Arts
in the University of Durham
for the Degree of Dector of Philosophy

The copyright of this thesis rests with the author.

No quotation from it should be published without his prior written consent and information derived from it should be acknowledged.

School of Oriental Studies,
Elvet Hill,
Durham

May, 1979

#### Abstract

This thesis gives particular attention to the <u>Ugul</u> of the noted grammarian, Ibn al-Sarrāj, but some attention is also given to the <u>Muqtadab</u> of al-Mubarrad, his teacher, although this latter work is less significant. This dissertation also provides a more general discussion of grammatical thought as relevant material from the works of earlier and later scholars has also been introduced.

Chapter I consists of an account of the history of Arabic grammatical thought up until the early 10th, century and special attention is given to al-Mubarrad and Ibn al-Sarraj and their writings on grammar.

Chapter II deals with the methodical and systematic approach of The al-Sarraj to his subject matter and assesses the significance of this.

Chapter III deals with Ibn al-Sarraj's discussion of the regent (amil) and considers related questions.

Chapter IV is a discussion of aspects of <u>divas</u> and <u>tagdir</u>, two important concepts in the methodology of the Arab grammarians.

Chapter V looks specifically at how gives determines the relationship of asl and far 6.

Chapter VI examines how al-Mubarrad and Ibn al-Sarraj approach two specified topics, the <u>tamyiz</u> and the verb of wonder (fi<sup>c</sup>l al-ta<sup>c</sup>ajjub).

Chapter VII deals with Ibn al-Sarraj's treatment of Kufan grammatical thought, and his use of the expression "the Baghdadis" is considered. In addition, the question of Kufan influence on Ibn al-Sarraj is discussed.

Chapter VIII consists of an examination of material taken from the <u>Usul</u> by Ibn Ya<sup>c</sup>ish and incorporated into his commentary on the <u>Mufaggal</u> of al-Zamakhshari.

## CONTENTS

Titlep		i
Abstra		ii
Table	of contents	iii
List o	f tables	iv
Acknow	<b>ledguments</b>	v
Transliteration		
	•	
Chapte	r ·	
~	TAMENONIA	1
I.	INTRODUCTION	ı
II.	CLASSIFICATION, ANALYSIS, AND DEFINITION	27
III.	REGENTS, THEIR CLASSIFICATION, AND RELATED THEORY	55
IV.	THE USE OF QIYAS AND TAQDIR	79
٧.	QIYAS AND THE RELATIONSHIP OF ASL AND FAR	102
VI.	THE TAMYIZ AND THE VERB OF WONDER	117
VII.	THE USUL AND THE KUFAN SCHOOL OF GRAWMAR	139
VIII.	IBN YACISH AND THE UŞUL OF IBN AL-SARRAJ	166
e o o		e o
ਵਜ਼ਾ ਜ਼ਵਾ	TWD BTRITOCDADHY	182

# List of tables

1.	The division of the parts of speech into regents	
	according to Ibn al-Sarraj	56
2.	The "hundred regente" according to al-Jurjani	57

## Acknowledgments

In connection with this thesis I would like to acknowledge the help and encouragement of my supervisor, J. A. Haywood, Esq., Reader in Arabic in the University of Durham. I would also like to make mention of the assistance given by Mrs. J. Butterworth of the library staff at Elvet Hill who made every effort to obtain necessary works.

## Transliteration

## Consonants

## Vowels and diphthongs

k

h

у

#### CHAPTER I

#### INTRODUCTION

Comparatively little attention has been given by scholars in the West to the detailed study of the development of Arabic grammatical thought. While they have been interested in the grammatical tradition of the Arabs, they have largely directed their efforts to preparing editions of the classics of Arabic grammar. The first detailed European study of the Arabic grammatical tradition is Gustav Flugel's work of 1862, Die grammatischen Schulon der Araber, but this is essentially an account of the lives and works of the grammarians based on the then available biographical sources and no attention is given in it to the study of grammatical thought. The first proper study of grammatical thought by a Western scholar is the essay by Gotthold Weil which appears in the Introduction to his odition of the Kitab al-ingaf of Ibn al-Antari which was published in 1913. This essay remains important and other later European scholars who have given attention to Arabic grammatical thought can be seen to have based themselves firmly on the work of Weil. 2 In the Middle East usoful studies have been made of the lives and works of individual grammarians, 3 but more general works on grammatical thought have tended to be biographical in nature and little emphasis has been put on critically examining the development of the ideas and techniques of the grammarians. 4

Although there is considerable scope for undertaking research on the development of Arabic grammatical thought there are certain limits on what can be done. This is because there are some quite considerable gaps in the works readily available of grammarians from the period between the time of Sibarayh in the late 8th century and that of al-Zamakhahari in the late 11th and early 12th centuries. Many works of this period appear to be irrevocably lost although manuscripts of works thought to be lost are still being discovered and catalogued. Although research into Arabic grammatical thought is hindered by the loss of much valuable material it has been greatly assisted in recent times by the publication of further grammatical texts based both on long-known and newly discovered manuscripts.

The publication of editions of the Mugtadab of al-Mubarrad and the Usul of his pupil. Ibn al-Sarraj, is an important contribution to making easily available the works of the early grammarians. ' Togsther these two works provide such information about grammatical studies in the later 9th and early 10th centuries which is a period to which up until recently little attention could really be given. Because of the availability of these two works it is possible to show how the grammatical scholarship of this particular period relates to that of the later pariod of the famous classics of Arabic grammer and to show how grammatical studies had progressed since the time of Sibawayh. However, in this present thesis attention is restricted to the grammarians work on syntax and their work on norphology and phonology has not been taken into consideration. This step has been taken to put necessary limitations on the scope of the thesis and can be justified because syntex, morphology, and phonology are rather different branches of linguistic study.

In the history of Arabic grammatical studies the later 9th and early 10th centuries constitute the period of al-Mubarrad and his immediate followers amongst whom was Ibn al-Sarrāj. While there are positive reasons for taking the later 9th century as a point of departure there is an important, but negative reason for so doing. Because of the action of time the works of scholars who flourished in the period

between Siberryh and el-Huberrad seed largely to have been lost and their contribution to grammatical thereint can only be imperfectly known through leter occandary sources. There execute works such as the <u>Theriff</u> of al-Husini which was the subject of a commentary by Ibn Jinai and has survived through the commentary, but dajor works of grammar from this period do not appear to have survived.

Probably the cost important work of grander from the ported between the <u>Kitāb</u> of Sībewayh and the <u>Magtadab</u> of al-Kabewred was the Mrs 11 of al-Akhiash al-Aver; (d. 830) who was a scholar whose views and froquently quoted by later grammarians. The manuscripts of this work are yet known end, indeed, there is no evidence that late scholare 1110 al-Savati (1445-1505) and 'Abd al-Qadir al-Beadcai (1621-82) bad at thoir disposed copies of this work. This would tend to confirm that this vork emorget others of el-Akhsach's vas lost at a correctatively ourly dato, although one of the pupile of Non al-Sarraj, al-Russinj, troto a commontary on it. 8 It is possible by consulting secondary Downson to dicoover the views of a scholar like al-Athlesh on many polato and even to find quotations from his works, but this cannot be copy coubotitate for being able to consult his works at first band which is vital for obtaining a truly accurate vior of the dovelopment of grammatical thought. Because of this loss of works of scholars like el-Althiach, it is necessary in attempting to take up in detail from primary sources the history of grammatical thought after Sibanaph to rate a start with the work of al-Maderred.

Although the point just made constitutes a consulat negative exgress for taking the later part of the 9th century as a point of departure for the present study, there are now positive considerations for so doing. The Basers school of granter, which becaus the school of Arabic granter, reached an important stage in its development with al-Haberral for, when he noved from Basersh to take up residence in Basersh the Basers school itself was virtually removed from its town

of origin to Regaded. Up wattl the exrivel of el-Maderred in Regaded the green tion cohool which bed flowriched there were that of Kufch and it had long been established in the setropolis. Indeed, the ourliest Kufen greenwiens of note, el-Kisä'ī (d. 804) and el-Farra' (d. 822), both left Indah to take up residence in Regaded.

By the later 9th contury the Kufan coheel appears to have become consulate of the first order. The lab, the great Kufan contemperary of al-Maderral, was without doubt a distinguished scholar but he cannot be considered a gravitation of the same stature as his famous predecessors, al-Kieā'I and al-Farrā'. When al-Maderral settled in Baghdad an alternative tradition of grammar and philology became readily accessible. Students were attracted to the circle of al-Maderral and the Bastran school took firm root in Baghdad and eventually completely ousted the Kufan school.

Another reason for taking the later 9th century as a point of departure is that the leading grammarian of the period, al-Mubarrad, is chronologically the most distinguished Basran scholar after Sībawayh who gave particular attention to grammar. Sībawayh's pupil, al-Akhfash, was a noted and often quoted grammarian but he cannot be said even to approach the rank of al-Mubarrad as a philologist. Even if al-Mubarrad was a noted grammarian his reputation was established by his great work of adab, the Kāmil, and by his general standing as a philologist the position of the Basran school of grammar in particular must have been immeasurably strengthened.

While there are particularly clear reasons for beginning a study of grammatical thought with the later 9th century, the period of al-Mubarrad, terminating it with the early 10th century, the period of his students, is somewhat more arbitrary but is justified simply by the need to limit the scope of the study. However, there are other considerations for setting such a limit. From among the pupils of

al-Nubarrad this study gives attention to the work of Ibn al-Sarrāj and, if al-Nubarrad can be said to have inaugurated a particular phase in the development of the Basran school, then his pupil, Ibn al-Sarrāj, brought it to a particular highpoint with his renowned work, the Ugul.

Al-Nubarrad's most outstanding pupils were al-Zajjāj, whose work has largely been lost, and Ibn al-Sarrāj, whose major work on grammar is extant. The grammatical works of lesser grammarian pupils of al-Nubarrad are also largely lost like those of al-Zajjāj and in any case did not attract much attention. In short, this means that the work of al-Pubarrad and Ibn al-Sarrāj form together a convenient subject for a study which covers the work on grammar of two generations of scholars.

Having explained the period which this present study intends to cover, it is necessary to give some account of the prior history of the study of grammar. In undertaking such a review of the history of Arabic grammatical thought the work of Sibawayh, which is contained in the <u>Kitab</u>, forms a convenient and suitable starting point. Firstly, the <u>Kitab</u>, which was acclaimed as the "Qur'an of grammar", 9 is the earliest monument of the Basran grammatical tradition and was a work which was universally praised for its excellence and had a decisive influence on the subsequent study of grammar. Secondly, knowledge of the grammarians of the Basran school who preceded Sibawayh is to be derived from what is recorded of their work in the <u>Kitab</u> because this book seems to have superceded at an early date the works of preceding Basran schoolars.

In considering the further development of the Basran school in the period between Sibawayh and al-Mubarrad an important point which emerges is that the Basran study of grammar was continued and developed by comparatively few specialists. Sibawayh was succeeded by his older contemporary and pupil, al-Akhfash al-Awsat (d. 830), and he was succeeded by his pupils, al-Jarmi (d. 840) and al-Mazini (d. 862), of

thom the latter was also a pupil of the former. 10 These two scholars were succeeded in turn by their pupil, al-Mubarrad. Al-Akhfash was undoubtably a more distinguished scholar than his two pupils, al-Jarmi and al-Mazini, and his views are very frequently quoted by later authors. His most famous work, which is often mentioned by name, is the Masa'il. It is apparent that he wrote two works of that name, the Kitab al-masa'il al-kabir and the Kitab al-masa'il al-saghir, but often the two titles are not clearly distinguished and reference is made simply to the Masa'il. 11 It is recorded that Ibn al-Sarraj made particular use of the Masa'll which is an indication of the esteem in which the. work was held. 12 However, the Masz'il does not seem to have been given the attention in later times which a work like the Ugul of Ibn al-Sarraj was given and after a period of popularity it fell into disuse. Al-Akhfash's pupils, al-Jarmi and al-Mazini, were scholars of note and their views are referred to by later scholars but they had the stature neither of their teacher, al-Akhfash, nor far less of their pupil, al-Mubarrad, and no further mention of their work is necessary here.

Abu 1- 'Abbas Muḥammad b. Yazid al-Mubarrad (826-98) was born in Basrah and as was natural his training in the linguistic sciences was in the Basran tradition. <sup>13</sup> He began his grammatical training by studying the <u>Kitab</u> of Sibawayh under al-Jarmi and on this scholar's death he continued his study of the <u>Kitab</u> with al-Mazini. An eye-witness account indicates that al-Mubarrad distinguished himself even as a pupil studying under al-Mazini: <sup>14</sup>

رأينا محد بن يزيد وهو هديث السن متصدرا في هلقة أبى عثمان المازني يقرأ عليه كتاب سيويه, وأبو عثمان في تلك الحلقة كأهد من فيها. Al-Mubarrad did not remain in Basrah but at some time moved to Samarra which was then the seat of the caliphal government. It is said that the reason for his coming was that he was summoned to settle a dispute between the caliph, al-Mutawakkil (847-61), and his well-known intimate, al-Fath b. Khaqan, over a point of Qur'an vowelling. Be that as it may, al-Mubarrad did definitely move in court circles in Samarra but on the murder of the caliph, al-Mubarrad moved to Baghdad and appears to have resided there until his death in 898.

As a result of his arrival in Baghdad al-Mubarrad's depth of knowledge became widely known there and he began to attract students somewhat to the detriment of the Kufan scholar, Tha'lab, 15 who was at that time the leading philologist in Baghdad. On this matter there are, in particular, accounts of how al-Zajjaj, who became an important Basran scholar and who had up until then studied under Tha'lab, was attracted to al-Mubarrad and abandoned his studies with his old teacher: 16

وقال الزهاج: لها قدم المبرد بفداد هنت لأناظره وكنت أقرأ على أبى العباس نقلب. فعزمت على إعناته فلما باهنته ألجمنى بالجئ وطالبنى بالعلة وألزمنى إلزامات لم أهتد إليها فأستيقنت فضله واسترهمت عقله وأهندت في ملازمته.

This account makes clear the immediate impression which al-Mubarrad made on many who heard him in discussion right from his earliest days in Baghdad and it was from such auspicious beginnings that his career in Baghdad developed.

Al-Mubarrad's most important work on grammar is the <u>Muqtadab</u> which is discussed below. Apart from this work it is also worth mentioning his <u>Radd 'ala Sibawayh</u> in which he criticised certain views taken by Sibawayh. This work has not survived but its contents are known from the <u>Intisar</u> of Ibn Wallad (d. 943) which is a refutation

of al-Mubarrad and a vindication of Sibavayh. This work has survived and "Adimah, the editor of the Muqtadab, makes use of it in his marginal notes. 17 Al-Mubarrad also wrote quite a number of works on grammar apart from the two mentioned here and many of them were directly concerned with the study and elucidation of the Kitab of Sibavayh. However, these works have not survived and do not seem to have attracted much attention on the part of later scholars. They are only known as titles and these can be ascertained from the biographical sources.

After the Kamil al-Mubarrad's most important work was his major treatise on grammar, the Mugtadab, which is in fact an earlier work than the Kamil as it is alluded to on several occasions in it. It is clear that the Muqtadab did not have the same high reputation of a work like the Usul of Ibn al-Sarraj and it was not a grammar which later scholars often had recourse to. However, it did have some popularity for a period, which is indicated by the fact that commentaries were written on it. 20 Ibn al-Sarraj's pupil, Abu 'Ali al-Farisi, wrote a commentary and al-Mubarrad's own pupil, Ibn Durustawayh, wrote a commentary which he did not complete. 21 A commentary was also done by Ibn Badhish (1055-1133) who was a Spanish scholar of Granada but a comparatively minor scholar, although an assiduous commentator on the classics of grammar. 22 It is clear that the Magtadab continued to be used in Spain after the time of Ibn Badhish because the Spanish scholar, Ibn Khayr (1108-79), lists it in his Fahrasah as a work which had been transmitted down to his time. 23

A specialised commentary on the <u>Muqtadab</u> has survived and that is the <u>Tafsīr al-masā'il al-mushkilah fī avwal al-muqtadab</u> by Sa'īd b. Sa'īd al-Fariqī (d. 1001). <sup>24</sup> In the <u>Muqtadab</u> there are a number of very complicated, but artificially constructed sentences which were used to train students and in his commentary al-Fariqī undertakes to analyse and explain such sentences. <sup>4</sup>Adīmah has incorporated into his

marginal notes those parts of this commentary which are strictly relevant to the text of the <u>Muqtadab</u>. It is clear from what al-Farique writes in the introduction to his commentary that the <u>Muqtadab</u> enjoyed considerable popularity in the later 10th century amongst beginners and those who had gone a little deeper into grammar.

However, the sort of commendatory remark about the <u>Usul</u> of Ibn al-Sarraj which can be found in the biographical sources cannot be found in the case of the <u>Muqtadab</u>; indeed, references to it tend to be unfavourable. There is a story that on one occasion, in the presence of Ibn al-Sarraj, one of his pupils compared al-Mubarrad's <u>Muqtadab</u> unfavourably with the <u>Usul</u> and Ibn al-Sarraj felt obliged to defend his old teacher's reputation:

جرى بحضرة ابن السرام ذكر كتابه الأصول الذي مسنفه فقال قائل: هو أحسن من المقتضب فقال أبو بكر (ابن السرام): لا تقل هكذا، وأنشد:

وَنَكِنْ بَكَتْ قَبْلِي فَهَيْجَ لِي ٱلْبُكَا فَمُلْتُ الْفَضْلُ لِلْمُنْتَقَدِّم.

There is also a rather derogatory remark about the <u>Muqtadab</u> which Yaqut records Abū 'Alī al-Fārisī as having made: 27

قال ابر على: نظرت في المقتضب فها انتفعت منه بنشى الا بمسألة واهدة وهي وقوع إذا جوابا للشرط في قوله تعالى: وَإِنْ نُصِبْهُمْ سَيِّنَةٌ بِهَا قَدَّمَتْ أَيْدِيهِمْ إِذَا هُمْ يَقْنِطُونَ.

This remark is nothing more than an attempt to belittle the value of the <u>Muqtadab</u> and is really a rather fatuous comment because, as 'Adimah notes, with respect to the point on which Abu 'Ali al-Farisi did reputedly

benefit from the <u>Muqtadab</u>, there is in fact no more information given there than in the Kitab. 28

After recounting this remark Yaqut goes on to mention a general reason why the Muqtadab was not a work from which profit was derived: 29

ویزعمون أن سبب عدم الانتفاع به أن هذا الکتاب أذنه ابن الراوندی الزندیق عن المبرد و تناوله الناس من ید ابن الراوندی . فكأنه عاد علیه شؤمه فلا یكاد بنتنم به.

This is again simply a rather abusive remark at al-Mubarrad's expense because he had the evident misfortune of being a teacher of the notorious heretic Ibn al-Rāwandī. The actual fact, although Ibn al-Rāwandī may have studied the Mugtadab under al-Mubarrad there is no reason to suppose that he had any great part in continuing the study of this work which would rather have been done by those scholars who were known as grammarians.

Although al-Mubarrad was recognised as a grammarian this did not result in a strong continuing interest in his works which was to last for many centuries. He did continue to be remembered as a grammarian but later scholars contented themselves with merely citing his views which they seem to have been aware of largely through secondary sources. It is comparatively rare for the later scholars to cite al-Mubarrad's major work on grammar, the Muqtadab, although specific references to it can be found.

As a grammar, the <u>Muqtadab</u> is quite a large-scale work and in the printed edition occupies four volumes. One of the very obvious characteristics of this work is the lack of any systematic arrangement of its chapters. In the <u>Kitāb</u> the chapters on accidence and syntax are to a large extent kept separate even if beyond this there is no clear

scheme for arranging the material. However, in the <u>Mugtadab</u> there is not even any separation of the chapters on accidence and syntax. On this point it is worth noting that Ibn al-Sarrāj is specifically described as giving the material in the <u>Ugul</u> an excellent arrangement and this may well indicate that previous works like the <u>Mugtadab</u> and the <u>Kitab</u> were felt to be somewhat deficient in this respect. <sup>32</sup> This would tend to suggest that there is no reason for supposing that the rather haphazard order of the chapters in the <u>Mugtadab</u> does not represent very much the original order in which al-Mubarrad composed the work.

Although 'Adimah takes the view that the manuscript on which he bases his printed edition is complete, or very nearly so, it is worth noting that al-Mubarrad seems to neglect certain topics. There are, for example, no chapters on the maf'ul labu or the maf'ul ma'ahu, but apart from this, and disregarding the haphazard arrangement of material, the Muqtadab does cover at some point most major topics usually discussed in a grammar.

The most important continuator of the work of al-Nubarrad was his pupil, Abū Bakr Muḥammad b. al-Sarī, generally known as Ibn al-Sarrāj. 34 Although the date of the death of Ibn al-Sarrāj is given as 316 A.H. (929 A.D.), there is no information on when he was born. The editors of his brief work on grammar, the Mujaz fi l-nahu, El-Chouemi and Damerdji, place his birth between the years 260-5 A.H. and this is based on facts known about him. 35 Firstly, according to an anecdote, Ibn al-Sarrāj was present at the entry of the caliph, al-Muktafī, into Baghdad in 289 A.H. At that time Ibn al-Sarraj was in love with a slave girl and in some verses drew a comparison between her and the pomp of the caliphal procession. Secondly, and this seems a stronger argument, he was a pupil of al-Mubarrad who died in 285 A.H. and before the death of his teacher he had made a reputation for himself as a pupil of

distinction. As to the places where he resided, it appears that Ibn al-Sarrāj, unlike al-Mubarrad, spent all his life in Baghdad and the editors of the <u>Mujaz</u> point out in this connection that Yaqut styles him the "Baghdadi" and that anecdotes about him are set in Baghdad. 36

Ibn al-Sarraj received his philological training under al-Mubarrad and he became an outstanding pupil, particularly favoured by his teacher: 37

كان ابن السراج من أهدت غلمان المبرد سنا مع ذكائه ومطلته وكان المبرد يسيل إليه ويقربه ويشرع ل ويجتمع معه في الخلوات والرعوات ويأنس به .

There is no record that Ibn al-Sarraj had any teacher other than al-Mubarrad but the latter was such a distinguished philologist that a pupil would no doubt have received all the training he needed from him alone. Ibn al-Sarraj does not seem to have studied under Tha lab, the great Kufan contemporary of al-Mubarrad, although he probably could have if he had so wanted.

If al-Mubarrad was his only attested teacher, Ibn al-Sarraj was, at least, an associate of al-Zajjaj, a somewhat older pupil of al-Mubarrad. There is a story recorded of an occasion when Ibn al-Sarraj and al-Zajjaj were together which throws light on the career of the former as a grammarian: 39

ورأیت ابن السراج یوما وقد مضر عند الزهاج مسلما علیه بعد موت المبرد، فسأل رجل الزجاج عن مسانة فقال لابن السراج: أجبه یأبا بکر. فأجاب فأخطأ فانتهزه الزجاج وقال: لوکنت فی منزلی فنریتل ولکن المجلس لا مجمل هذا. وقد کنا نشبهلا فی الزکاد والفطنة لابی هسن بن رهاه و أنت تخطی فی منل هذا. فقال: قد فرتنی

یأبا اساق وأدبتنی وأنا تارك ما درست مذ قرأت هذا الکتاب سیبویه - لأن تشاغلت منه بالمنطق والموسیتی والدن أنا أفاود.

According to the tailpiece to this story Ibn al-Sarraj was as good as his word and went on to become the leading grammarian after the death of al-Zajjāj.

Ibn al-Sarrāj's reputation as a scholar is firmly based on his major work of grammar, the Usul. Although this work did not become one of the great classics of grammar that were in wide use in the later Middle Ages, nevertheless, it was apparently quite widely used in the centuries after its composition and remained a work well-known to specialists in grammar. The biographical sources make clear that the Usul was a work that was well thought of although, because such sources tend to take over much material directly from earlier works, it is not easy to date from them the period when such commendatory remarks were first made. With perhaps a shade of hyperbole, Yaqut records the remark in praise of Ibn al-Sarrāj's scholarship in the Usul:

وله من المصنفات كناب الأصول وهو أهسنها وأكبرها واليه المرجع عند اضطراب النقل واختلافه. وجمع فيه أصول علم الوبية وأخذ مسائل سيبويه ورتبها أهسن ترتيب.

Apart from such commendations of Ibn al-Sarraj and his scholarship, the continued use of the <u>Usul</u> itself testifies to the high esteem in which its author was held. However, it is not possible, due to the lack of the necessary evidence, to give a complete account of the later

history of the use of the Usul but certain details can be given. First of all, a number of commentaries on it were written. 42 The earliest commentary was composed by Ibn al-Sarraj's own pupil, al-Rusmani (909-94). 43 The next was done by Ibn Babshadh (d. 1077), an Egyptian scholar of distinction, whose works had some popularity. 44 The next two commentators are scholars of the Islamic West. The first is Ibn Badhish (1055-1133) who has already been mentioned as having written a commentary on the Muntadab. The second, who is the last attested commentator, was the notable North African grammarian, al-Jazuli (d. ca. 1209-19). 45 His teacher was another eminent North African scholar, Ibn Barri (1106-87), and it is recorded that in an hour of need al-Jazuli pawned his copy of the Usul which he had made himself while studying the work with Ibn Barri. 46

Further information on the use of the <u>Ugal</u> in the Islamic West comes from the <u>Fabrasah</u> of Ibn Khayr (1108-79) in which the work is listed as handed down from generation to generation. <sup>47</sup> Ibn Khayr gives two chains of transmission back to Ibn al-Sarraj, one through Abu 'Ali al-Farisi and the other through al-Sirafi, the author of the famous commentary on the <u>Kitab</u>. <sup>48</sup> He also mentions al-Rummani's commentary on the <u>Ugul</u> but does not provide a chain of transmission.

From the evidence of the commentaries written on the Usul and of the Fahrasah of Ion Khayr it is clear that the Usul was in general use down to the 12th century and in particular in the Islamic West. This last point is supported by the fact that two of the four texts used by al-Fatli in preparing his printed edition of the Usul were found in North Africa. 49 After the time of Ion al-Sarraj's pupil, al-Rummani, it is difficult to trace the history of the use of the Usul in the Eastern Islamic world because the Kashf al-zunun of Hajji Khalifah does not mention any further commentaries from that region, and there are no sources of information for the East comparable with the Fahrasah

of Ton Khayr for Spain.

The Usul, however, did remain an important work and was one consulted by scholars undertaking very detailed studies of grammar.

Al-Suyūţī (1445-1505) frequently cites Ibn al-Sarrāj in the Ashbāh

wa-l-nagā ir and in it quotes passages from the Usūl. 50 The Usūl is

also the earliest work which al-Suyūţī draws on in his treatise on

grammatical methodology, the Iqtirāh. 51 'Abd al-Qādir al-Baghdādī

(1621-82) also found the Usūl valuable and in his introduction to the

Khizānat al-adab lists the Usūl as one of the works which he particularly

consulted. 52 It is of interest that it is the earliest grammar after

the Kitāb in al-Baghdādī's list of works consulted. It is also worth

noting here that Ibn Ya'īch also found the Usūl as it stands and works

it into his commentary on the Mufagsal of al-Zamakhsharī. This is a

matter which will be dealt with in detail in a later chapter. 53

fortunately, the published parts include all the sections of the work dealing with syntax with which this thesis is particularly concerned. The Usul is a work in which the various chapters are presented in a coherent and well thought out manner and the significance of this will be discussed in a later chapter. <sup>54</sup> A particular point about the method of presentation of the material in the Usul is that Ibn al-Sarraj tends to treat the various topics by having a section which constitutes a basic discussion and a section constituting an additional discussion of further points which he calls masa'll. Thus, for instance, there is the points which is followed by the particular in the Muqtaqab.

Among other works on grammar by Ibn al-Sarraj was a commentary on the <u>Kitab</u> but this has not survived and never had the popularity of

the later commentary on the <u>Kitab</u> by al-Sīrāfī. 55 Another work on grammar by Ibn al-Sarrāj is the <u>Jumal al-ugul</u> which was also known as the <u>Kitab al-ugul al-gaghīr</u> and is clearly an abridgment of the <u>Ugul</u>. 56 There is also mention in the biographical sources of another work by Ibn al-Sarrāj whose title has the consonantal skeleton ..., but what this work is or what the correct vocalisation of this word is cannot be said. 57 There is no reason to suppose that this is a work on grammar called the <u>Jumal</u> which is to be differentiated from the abovementioned <u>Jumal al-ugul</u>. References to a work on grammar by Ibn al-Sarrāj called the <u>Jumal</u> would simply involve an abbreviation of the fuller title <u>Jumal al-ugul</u>.

In the <u>Unul</u> Ibn al-Sarraj mentions a work called simply the <u>Jumal</u> and relates its contents and format to that of the <u>Usul</u>: 58

قد انتهینا إلى الموضع الذى يتساوى فيه كتاب الاصول وكتاب الجمل بعد ذكر الذى والألف والألف واللام. أم لا فرق بينهما إلا أن بعد التصريف زيادة المسائل فيه و الجمل ليس فيه ذلك.

It is recorded that al-Rummani wrote a commentary on the <u>Jumal</u>, <sup>59</sup> and a rather minor scholar called Ibn Humaydah (1076-1155) wrote a commentary on the verses cited in the <u>Jumal</u>. <sup>60</sup> Although little is known about this scholar, the commentary by him is at least evidence that the <u>Jumal</u> was in use up until the middle of the 12th century.

Apart from the <u>Ugul</u>, another work on grammar by Ibn al-Sarraj has survived and been edited and this is called the <u>Mujaz</u>. <sup>61</sup> This is a brief resume of Arabic grammar and al-Ma<sup>a</sup>arri provides some information on the writing of this work:

وذكر المعرى في رسالة الففران أن أبا على الفارسي كان يذكر أن أبا بكر ابن السراج

عمل من الموجز النصف الأول لرجل بزاز نم تقدم إلى أبى على الفارس باتهامه، قال. وهذا لا يقال أنه من إنشاء أبى على لأن الموضوع في الموجز هو منقول من كلام ابن السراج في الأصول والجمل فكأن أبا على هاء به على سبيل النسخ ، لا أن ابتدع شيئا من عنده.

This work had a certain popularity and is mentioned by Ibn Khayr in his <u>Fahrasah</u> and he gives the same two chains of transmission back to its author which he gives for the <u>Usul</u>. 63 The <u>Mujaz</u> was also the subject of a commentary by al-Rummani and this work is again mentioned by Ibn Khayr. 64

The only other Basran pupil of al-Nubarrad who can be ranked with Ibn al-Sarrāj as a grammarian is al-Zajjāj (d. 923) but he did not have the lasting reputation of Ibn al-Sarrāj nor were his writings held in such high regard. 65 However, his Ma'anī l-Qur'ān did have a certain popularity and was a work which Ibn Khayr studied with his teachers and whose transmission chain he could trace back to al-Zajjāj. 66 He was the first pupil of al-Mubarrad after the latter arrived in Baghdad, and he seems to have been a particular intimate of his teacher because anyons interested in becoming a pupil of al-Mubarrad first took up the matter with him. He was a much older man than Ibn al-Sarrāj and, according to al-Zubaydī, he was over eighty when he died which would make him probably some twenty years older. After the death of al-Mubarrad it was he who became the senior Basran scholar and he was recognised as the head of the Basran school. 67

The next most distinguished grammarian from among the pupils of al-Mubarrad was Ibn Durustawayh (872-958) who wrote a number of grammatical works. 68 He was reckoned to hold very firmly to the Basran

school and wrote a work called the Radd 'alo Tha'lab fi khtilaf al-mahwiyin which was a reply to Tha'lab's exposition of the differences between the grammatical schools, the Uthtilaf al-mahwiyin. <sup>69</sup> Among lesser Basran grammatians of this generation it is worth mentioning the names of Mabraman (d. 956) who was a pupil of both al-Mubarrad and al-Zajjāj, and 'Ali b. Sulaymān al-Akhfash al-Şaghir (d. 928) who studied both under the Basran, al-Mubarrad, and the Kufan, Tha'lab. <sup>70</sup> Then the generation of Basran scholars who were pupils of al-Mubarrad are taken into consideration, it is clear that they produced no other work of grammar which can compare in reputation with the Usul of Ibn al-Sarraj, and this is confirmed by the fact that no such work has survived. For this reason an assessment of Basran grammatical thought in the early 10th century must be firmly based on a study of the Usul.

This thesis is particularly concerned with Basran grammatical thought because in the period under consideration it reached an important stage in its development into the dominant school of Arabic grammar. On the other hand, the rival school of Kufah was in decline and was making no significant advances in the field of grammatical thought. Kufan grammatical study consisted largely of giving attention to the works of past scholars and there was little new work of any vitality. However, for the sake of completeness, it is necessary to give some consideration to the history of the Kufan school of grammar and to take particular account of its state during the period considered in the present study.

Although considerable work has still to be done on the history of Kufan grammatical thought, with particular need for a study of the Ma'ani 1-Qur'an of al-Farra', certain general judgments can be made on the development of the school. 71 It is clear that the only Kufan scholars who were grammarians of note were al-Kisa'i (d. 804) and

al-Farra' (d. ca. 822). 72 When individual Kufen gramarians are particularly mentioned it is these two scholars who are singled out on most occasions. Outside the field of grammar, al-Kisa'i is best remembered as the scholar who established one of the three sets of Kufan canonical Qur'an readings. Al-Farra' is best remembered as the author of the still extant and partly published Ma'ani l-Qur'an. This is a verse-by-verse, surah-by-surah commentary on the Qur'an which is predominantly concerned with grammar. This work was held in high repute and was used by al-Baghdadi in the Khizanat al-adab. 73

After al-Farra' there did not emerge any Kufan grammarian of great note and this had the corresponding result that Kufan grammatical thought did not develop any further. This placed it in an unfavourable position to compete with the Basran school of grammar which continued to be developed by able scholars. The Kufan school also had a great weakness in that there was no Kufan work of grammar which had the undisputed authority of the <u>Kitab</u> of Sibawayh among the Basrans. This last work was instrumental in firmly establishing the position of the Basran school.

In the history of the Basran school, the later 9th and early 10th centuries constitute the period of al-Mubarrad and his pupils, and correspondingly in the history of the Kufan school, the same time-span constitutes the period of the outstanding scholar, Tha'lab, and his pupils. 74 While it cannot be denied that Tha'lab was a distinguished philologist, as a grammarian he was much loss than an equal of al-Mubarrad. It appears that the main weakness of Tha'lab as a grammarian was that he had learned his grammar by studying the works of his Kufan predecessors and did not have much aptitude for reasoning things out on his own. This is made clear by an account of his teaching style as it appeared to his contemporaries: 75

Part of the reason why Thatlab did not become a good grammarian was that he had not been trained up to the level of the best Basran grammarians. Although he received a philological training from Kufan teachers, Tha'lab himself draws attention to his own personal study of the works of al-Farra, 76 No doubt much of his grammatical knowledge was acquired in precisely this same way and he seemed to lack the advantage of a really thorough grammatical training with a teacher. There is evidence indicating that Tha lab studied grammar under Salmah b. 'Asim but this scholar cannot be ranked with al-Mubarrad's grammar teachers, al-Jarmi and al-Mazini. 77 Tha lab s lack of grammatical training with particular regard to the Kitab, which he did study although it was a Basran work, is made clear by an anecdote. Tha 'lab's son-in-law, who to his annoyance used to go to al-Mubarrad to study the Kitab, was asked why al-Mubarrad was more knowledgeable on the Kitab than Tha lab, and he answered that the former had studied it under scholars whereas the latter had studied it under himself. 78 It also seems to be the case that in the main the Kufans regarded grammar as a subject which was to be studied as an introduction to a general philological training and they did not lay the emphasis on studying grammar for its own sake which the Basrans did.

Although Tha lab wrote several works on grammar these have not survived. 79 He does cover grammatical questions in his Majalis which has survived, but this provides no real basis for making meaningful comparisons with the output of the Basran school. 80 It is clear that neither Tha lab nor those of his pupils who remained within the Kufan tradition produced works that were the equal of contemporary Basran works, and this is confirmed by the fact that no such works have survived.

For this reason the present study is centred on the Basran school, although in a later chapter attention will be given to the Basran approach to Kufan scholarship at this period.

After dealing with the Basran and Kufan schools during the period under study, it is worth noting that there were a number of scholars who were pupils of both al-Mubarrad and Tha'lab and in their work were reckoned to have drawn both upon the grammatical traditions of the Basrans and of the Kufans. What this "mixing the two schools" meant in practice is difficult to assess because there are no relevant works extent to form the basis for a judgment. Such scholars never constituted a separate school but were seen as inclining more towards one school than the other. The most important of these scholars were Ibn Khayyat (d. 932) and Ibn Kaysan (d. 911), the latter of whom is reckoned by al-Sirafi to have been with al-Zajjaj the leading Basran scholars after the death of al-Mubarrad, although al-Sirafi points out that Ibn Kaysan did "mix the two schools". 82 This eclectic approach seems to have been a short-lived phenomenon and did not survive long beyond the generation of the pupils of the pupils of al-Mubarrad and Tha lab, and there is no evidence that it had any profound effect on the development of the mainstream Basran school.

A study of any Arab grammarian inevitably involves reference to the works of other grammarians and some indication is now given of the principle works referred to in the present study which were composed outside of the period considered in this present study. In a detailed study of grammatical thought it is necessary to refer chiefly to the more compendious works of grammar and for this reason little attention is given to such small-scale works as the well-known treatises of Ibn Hishām. 83 The main earlier work which is consulted in this present study is, of course, the <u>Kitāb</u> of Sibawayh. Of later works particular reference is made to the Sharp al-mufaşşal of Ibn Ya'ish. Although

the profundity and originality of this work is open to question, it is one which had been widely used in the Islamic world. As evidence of this may be cited the very frequent references to it in the Ashbah wall-naga'ir of al-Suyūtī. <sup>84</sup> Another work consulted is the commentary by al-Radi al-Astarabadhi on the Kafiyah of Ibn Rajib. <sup>85</sup> A compendious work to which reference has also been made but which is not so well-known is the Manhaj al-salik of the famous Spanish scholar, Abū Hayyan Athir al-Din. <sup>86</sup> This work is a commentary on the Alfiyah of Ibn Malik and is particularly useful because of the attention given to making clear the views of Sibawayh and to relating the views of the Basran and Kufan scholars as well as those of later scholars of the Islamic West.

## Notes to Chapter I

- 1 G. Flügel, Die grammatischen Schulen der Araber, Leipzig, 1862; Ibn al-Anbari, <u>Kitab al-insäf</u>, edited and with an Introduction by G. Weil, Leiden, 1913.
- See H. Fleisch, <u>Traité de philologie arabe</u>, 1 vol. published (Beirut, 1961), pp. 1-36; G. Lecomte, <u>Ibn Qutaybah</u>, <u>l'homme</u>, son oeuvre, ses idées (Damascus, 1965), pp. 377-96.
- Z. E.g. A. I. Shalabi, Abu 'Ali al-Farisi, Cairo, 1958; M. Mubarak, Al-Rummani al-nahwi, Damascus, 1963.
- 4 E.g. H. Awn, <u>Tatawwur al-dars al-nahwi</u>, Cairo, 1970; Shawqi Dayf, <u>Madaris al-nahwiyah</u>, Cairo, 1968; A. al-Sayyid, <u>Madrasat al-Başrah al-nahwiyah</u>, Cairo, 1968.
- 5 Al-Mubarrad, Mugtadab, 4 vols., edited and with an Introduction by M. A. 'Adimah ( عضيمة ), Cairo, 1965-8; Ibn al-Sarraj, Usul fil-nahw, 2 vols. published, edited and with an Introduction by A. al-Fatli, vol. i, Najaf, 1973, and vol. ii, Baghdad, 1973.
- 6 Ibn Jinni, Munşif, sharh kitab al-taşrif li-l-Mazini, Cairo, 1954.
- 7 For further mention of al-Akhfash and his work see p. 6 above and n. 10 below.
- 8 Al-Qifti, Inbah al-ruwah, 3 vols. (Cairo, 1950-5), vol. ii, p. 295.
- 9 Abu Tayyib al-Lughawi, Maratib al-nahwiyin (Cairo, 1955), p. 65.
- For al-Akhfash al-Awsat see Encyclopaedia of Islam, 2nd ed., s.v. al-Akhfash. (Hereafter this work is abbreviated to E.I<sup>2</sup>); for al-Jarmi see U. R. Kaḥḥalah, Mu'jam al-mu'allifin, 15 vols. (Damascus, 1957-61), vol. v, p. 3; for al-Mazini see Kaḥḥalah, vol. iii, p. 71.
- 11 See for instance the passages cited in n. 12 below.
- 12 Al-Qifti, vol. iii, p. 145; Yaqut, Mu'jam al-udaba', 7 vols. (Leyden and London, 1907-31), vol. vii, p. 10.
- For a comprehensive list of biographical sources on al-Mubarrad see Kahhalah, vol. xii, pp. 114-5. In particular see al-Qifti, vol. iii, pp. 241-53; Yaqut, vol. vii, pp. 137-44; al-Zubaydi, <u>Tabaqat</u> al-nahwiyin (Cairo, 1954), pp. 108-20.
- Al-Zubaydi, p. 108. In quoting Arabic passages from unpunctuated texts some punctuation has been added for convenience, based on standard Arabic practice. In punctuated texts, particularly in the Usul of Ton al-Sarraj, some changes in punctuation have been made for the sake of uniformity and convenience.
- 15 For Tha'lab see Kahhalah, vol. ii, pp. 203-4.
- 16 Yaqut, vol. vii, p. 14.

- 17 'Adimah, Introduction to the <u>Muqtadab</u>, pp. 89-95. For Ibn Wallad see Kahhalah, vol. ii, p. 167.
- 18 Ibn al-Nadim, Fihrist (Leipzig, 1872), p. 59; al-Qifţi, vol. iii, pp. 251-2; Yaqut, vol. vii, pp. 143-4.
- 19 See 'Adimah, pp. 78-80.
- 20 Hajji Khalifah, <u>Kashf al-zunun</u>, 2 vols. (Istanbul, 1941-3), vol. ii, p. 1793.
- 21 For Abu 'Ali al-Farisi see E.I.<sup>2</sup>, s.v. Abu 'Ali al-Farisi; for Ibn Durustawayh see E.I<sup>2</sup>, s.v. Ibn Durustawayh.
- 22 For Ibn Badhish see Kahhalah, vol. vii, p. 15.
- 23 Ibn Khayr, Fahrasah (Saragossa, 1894-5), p. 307.
- 24 See 'Adimah, pp. 83-6. For al-Fariqi see Kahhalah, vol. iv, p. 224.
- 25 'Adimah, p. 84.
- 26 Ibn al-Nadim, p. 62.
- 27 Yaqut, vol. vii, p. 143.
- 28 'Adimah, p. 70. See al-Mubarrad, vol. ii, pp. 57-8 and vol. iii, p. 178; Sibawayh, <u>Kitab</u>, 2 vols. (Bulaq, 1316-7 A.H.), vol. i, p. 435; W. Wright, <u>A grammar of the Arabic language</u>, 3rd ed., 2 vols. (reprinted Beirut, 1974), vol. ii, p. 345C.
- 29 Yaqut, vol. vii, p. 143.
- 30 See E.I.<sup>2</sup>, s.v. Ibn al-Rawandi.
- 31 See 'Adimah, pp. 77-8. See also M. A. Maiman, <u>Iqlid al-khizanah</u> (Lahore, 1927), s.v. <u>Muqtadab</u>.
- 32 See pp. 27-9 below.
- 33 See 'Adimah, pp. 74-81.
- 34 For a comprehensive list of biographical sources on Ibn al-Sarraj see Kahhalah, vol. x, p. 19. In particular see al-Qifti, vol. iii, pp. 145-9; Yaqut, vol. vii, pp. 9-12; al-Zubaydi, pp. 122-5.
- J5 Ibn al-Sarraj, <u>Mujaz fi l-nahw</u>, ed. M. El-Chouemi and B. Damerdji, (Beirut, 1965), pp. 5-6. This particular work is subsequently referred to by its author's name and title together to distinguish it from the <u>Usul</u>.
- 36 Ibid., p. 5.
- 37 Ibn al-Nadīm, p. 32. Also al-Qiftī, vol. iii, p. 148. In l. 4 of this quotation the تأنس of the <u>Fihrist</u> has been changed to the of the <u>Inbah</u>.
- 38 For al-Zajjāji see Kaḥḥālah, vol. i, p. 33.

- Ibn al-Nadim, p. 62. Also al-Qifti, vol. iii, p. 149. In l. 4 of this quotation the شهر بالركاء of the Fihrist has been changed to the نشبهائ في الذكاء
- 40 Yaqut, vol. vii, p. 10.
- 41 Ibid., pp. 10-1.
- 42 Hajji Khalifah, vol. i, p. 111.
- 43 For al-Rummani see Kahhalah, vol. vii, p. 162.
- 44 For Ibn Babshadh see Kahhalah, vol. v, p. 32.
- 45 See E.I.<sup>2</sup>, s.v. al-Djazuli.
- 46 See ibid., s.v. Ibn Barri.
- 47 Ibn Khayr, pp. 307-8.
- 48 For al-Sirafi see Kahhalah, vol. iii, pp. 242-3.
- 49 Ibn al-Sarraj, vol. i, pp. 33-4.
- 50 Al-Suyuti, Ashbah wa-l-naza'ir, 4 vols. (Hyderabad, 1359-61 A.H.), vol. i, pp. 13, 28, 90, 143-4, 241, 268, 274, 322: vol. ii, pp. 11, 51, 79, 87, 90, 91, 131.
- 51 Idem, Iqtirah (Hyderabad, 1359 A.H.), pp. 6, 49.
- 52 'Abd al-Qadir al-Baghdadi, Khizanat al-adab, 4 vols. (Beirut, n.d.), vol. i, p. 8. See Maiman, s.v. Usul.
- 53 See pp. 166-79 below. Ibn Ya'ish, Sharh al-mufassal, 10 vols., Cairo, n.d.
- 54 See pp. 27-9 below.
- 55 Ibn al-Nadim, p. 62.
- 56 Yaqut, vol. vii, p. 11.
- 57 Ibid.
- 58 Ibn al-Sarraj, vol. ii, p. 271.
- 59 Al-Qifti, vol. ii, p. 295.
- 60 Yaqut, vol. vii, p. 40.
- 61 See n. 35 above.
- 62 Yaqut, vol. iii, pp. 13-4. See also al-Ma'arri, Risalat al-ghufran, ed. Bint al-Shati' (Cairo, 1950), pp. 357-8.
- 63 Ibn Khayr, p. 310.
- 64 Tbid., p. 316. See also al-Qifti, vol. ii, p. 295; Yaqut, vol. v, p. 281.

- 65 See n. 38 above.
- 66 Ibn Khayr, pp. 64-5.
- 67 Ibn al-Nadim, p. 60; al-Sirafi, Akhbar al-nahwiyin al-Başriyin (Cairo, 1955), pp. 80-1.
- 68 See n. 21 above.
- 69 See Fleisch, p. 19.
- 70 For Mabraman see Kahhalah, vol. x, p. 307; for al-Akhfash al-Saghir see E.I., s.v. al-Akhfash.
- 71 Al-Farra', Ma'ani l-Qur'an, 2 vols. published, Cairo, 1955-
- 72 For al-Kisa'i see Kahhalah, vol. vii, p. 84; for al-Farra' see E.I.<sup>2</sup>, s.v. al-Farra'.
- 73 See Maiman, s.v. Ma'ani l-Qur'an.
- 74 For Tha lab see Kahhalah, vol. ii, pp. 203-4.
- 75 Al-Qifti, vol. i, p. 144; Yaqut, vol. i, p. 141.
- 76 Yaqut, vol. i, pp. 135, 140.
- 77 Al-Zubaydi, p. 150. See also Abū Tayyib, p. 96. For Salmah b. 'Asim see Kahhalah, vol. iv, pp. 240-1.
- 78 Al-Qifti, vol. i, pp. 144-5; al-Zubaydi, p. 156.
- 79 See Ibn al-Nadim, p. 74; al-Qifti, vol. i, pp. 150-1; Yaqut, vol. ii, pp. 152-3.
- 80 Tha'lab, Majalis Tha'lab, 2 vols., Cairo, 1948-9.
- 81 See pp. 139-62 below.
- 82 Al-Sirafi, pp. 80-1. For Ibn Khayyat see E.I.<sup>2</sup>, s.v. Ibn Khayyat; for Ibn Kaysan see E.I<sup>1</sup>, s.v. Ibn Kaysan.
- 83 See ibid., s.v. Ibn Hisham.
- 84 See p. 180, n. 2 below.
- 85 Al-Astarabadhi, Sharh al-kafiyah, 2 vols., Turkey (Istanbul?), n.d.
- 86 Abu Hayyan, Manhaj al-salik, New Haven, Connecticut, 1947.

#### CHAPTER II

## CLASSIFICATION, ANALYSIS, AND DEFINITION

In the <u>Kitāb</u> of Sībawayh little attention is given to arranging the various chapters in a logical and systematic manner. There are, of course, many instances of chapters on related but separate topics being grouped in a sequence, but there is nothing truly comparable with the ordered presentation of subject matter to be found in later works. This may well be due to the fact that the chapters of the <u>Kitāb</u> are largely arranged in the order in which Sībawayh happened to deal with them. In addition, the <u>Kitāb</u> is described in the biographical sources as a work unlike any previously written and its author would probably have been working without a convenient model for planning the lay—out of his work. 

More importantly, it is recorded that Sībawayh never taught the <u>Kitāb</u> to any of his students and this may well indicate that by the time of his death Sībawayh may not have completed revising and arranging the work.

The next major extant work of a Basran grammarian is the <u>Muqtadab</u> of al-Mubarrad and, as has been mentioned in the introduction, <sup>3</sup> this grammar also lacks a logical and systematic arrangement of its various chapters. However, the <u>Usul</u> of Ibn al-Sarraj, unlike the <u>Kitab</u> and the <u>Muqtadab</u>, is a work which is arranged in a careful and planned manner. It seems safe to assume that it is a characteristic which differentiates the <u>Usul</u> from other preceding major works of grammar and this would apply even in the case of works which are no longer extant. An indication that this assumption is correct is provided by a passage quoted by al-Qifti: <sup>4</sup>

قال أبر عبدالله المرزاني: صنف - يعني ابر السراع - كتابا في النحر محاه الأمول انتزعه من أبواب كتاب سيميه , وهمل أمنافه بالتقاسيم على لفظ المنطقيين فأخيب بهذا اللفظ الفلسفيون. وإنها أدخل فيه لفظ التقاسيم ؛ فأما الممنى فهو كله من كتا ب سيميه على ما قسمه ورتبه ؛ إلا أنه عول فيه على مسائل الدهنفش ومذاهب الكوفيين , وهالف أمول البصريين في أبواب كثيرة لتركه النظر في الموسيق .

With regard to the content of this passage it may also be mentioned that Yaqut writes of Ibn al-Sarraj in a passage previously quoted:

Before discussing further the passage quoted by al-Qifti from al-Marzubani (d. 994), it is worth noting that the tone of it does not appear to be particularly friendly towards Ibn al-Sarraj. The passage seems to belittle his work for it suggests that although he introduced into the <u>Usul</u> certain considerations drawn from logic, the content itself simply consists of material drawn from the <u>Kitāb</u> which has been re-arranged. In addition, there are certain influences from his attention to the <u>Masā'il</u> of al-Akhfash and Kufan grammatical thought, and deviations from Basran norms are attributed, somewhat disparagingly, to the fact that he was distracted from grammar by the study of music.

The passage suggests that Ibn al-Sarrāj introduced into grammar what are called "divisions" (taqāsīm). The meaning of this term is not completely clear but it seems to imply that al-Marzubānī holds that Ibn al-Sarrāj was influenced by classification procedures used in logic. The term <u>lafz</u>, as it is used in this passage, would seem to

mean "phraseology", implying that Ibn al-Sarraj introduced into the Usul the terminology used by the logicians in their classification procedures. Although Ibn al-Sarraj gives considerable attention in the Usul to classifying his material and arranges the work to take this into account, nevertheless, he does not express himself in terms which could be said to belong particularly to the terminology of the classification procedures of logic. The real force of al-Marzubani's remarks about Ibn al-Sarraj's approach would appear to be that in logic there are procedures for classification and analysis, and in the Usul too there is attention given to classification and analysis which was accordingly felt to have come about through the influence of logic. In a passage quoted earlier Ibn al-Sarraj confesses to having been distracted from grammar by the study of music and logic, and Ibn Abi Uşaybi'ah particularly mentions that Ibn al-Sarraj studied logic under the philosopher, al-Farabi. 8 This study of logic may well have contributed to making Ibn al-Sarraj methodical and systematic in his approach to his work as a grammarian.

on the evidence from biographical sources it may be inferred that Ibn al-Sarrāj's work on classification and analysis in the Usul was something that had not been undertaken before and essentially represents an innovation on his part. This, however, is represented in the passage quoted from al-Qifti as having come about through the influence of logic and is not attributed to a desire simply to arrange the Usul in a systematic and coherent manner which would have been a natural advance for some grammarian to make in writing a grammar. That this was an innovation on the part of Ibn al-Sarrāj is supported by the evident lack of systematic arrangement of material in earlier works like the Kitāb and the Muqtadab which has been already mentioned.

At various points in the <u>Usul</u> Ibn al-Sarraj explains the order in which he is dealing with the various topics to show that they are

being treated in a systematic manner. <sup>9</sup> In particular, at the beginning of the <u>Usul</u> Ibn al-Sarraj states that he will present his material in a convenient and well-ordered fashion and in the clearest possible terms: <sup>10</sup>

فقد أعلنت في هذا الكتاب أسرار الني وجمعته جمعا يحضره عفصلته تنصيلا يظهره. ورنيت أنواعه وصنوفه على سراتيه بأحضر ما أمكن من القول وأبينه ليسبق إلى القلوب فهمه ويسهل على متعليه حفظه.

It would be unfair to treat this statement as simply expressing the sort of conventional claim which an author might make in the preface of a work. Rather, it should be taken as a wholly justifiable claim, perhaps implying in itself that previous works are somewhat defective with regard to arrangement and clarity of expression.

In the <u>Usul</u> many examples can be found of Ibn al-Sarraj's attention to questions of classification and definition and a start will be made here with his treatment of the nominative case. Before he discusses the various uses of the nominative case Ibn al-Sarraj sets out what he considers them to be:

الأسماء التى ترتفع هنمسة أصناف:
الأول: مبتدأ له هبر
النانى: هبر المبتدأ بنيته عليه
النالث: فاعل بنى على فعل ودلا الفعل هديث عنه
الرابع: صفعول به بنى على فعل فهو هديث دنه ولم
يذكر من فعل به فقام مقام الفاعل
الخاسس: مشبّه بالفاعل في اللفظ.

The main point of interest about the list itself is that Ibn al-Sarraj

identifies a particular use of the nominative case which he calls the mushabbah bi-l-fa'il fi l-lafz or simply the mushabbah bi-l-fa'il.

This consists of the subject of kans and analogous verbs and of the subject of particles which behave like kans and the verb proper and this includes words like the negative particle ma in the Hijazi usage.

Although other grammarians took account of the uses of the nominative case which Ibn al-Sarraj classifies as mushabbah bi-l-fa'il, the expression itself for this class seems to be particular to Ibn al-Sarraj.

Of the five uses of the nominative which Ibn al-Sarraj lists, the <u>mubtada</u> and the <u>khabar</u> will be considered in the chapter on the regent and need not be discussed here. <sup>13</sup> The other three uses of the nominative case broadly cover what would be called in English the subject of the verb. The first of these three classes is the <u>fa'il</u> of which Ibn al-Sarraj offers the following definition:

الاسم الذي يرتفع بأنه فاعل هم الذي بنيسته على الفعل الذي بني للفاعل، ويجعل الفعل هدينا عنه مقدما قبله, كان فاعلا في الحقيقة أو لم يكن ، كقوللا: هاه زيد , وحات عمد ، وحا أشبه ذلك . وصفى قم لى : منينه على الفعل الذي بني للفاعل أي ذكرت الفعل قبل الدسم لأنك لو أتيت بالفعل بعد الاسم لارتفع قبل الدسم بالاحتداء . وإنها قلت : على الفعل الذي بني للفاعل ، كان وأعلا في كانوا قد فرقوا بينهما ... وإنها قلت : كان فاعلا في الحقيقة أو لم يكن ، لأن الفعل ينقسم قسمين : الحقيقة أو لم يكن ، لأن الفعل ينقسم قسمين : فمنه هقيق ومنه فيرهقيق .

This passage consists of a definition of what a  $\underline{fa}^t$ il is and an explanation of why the definition is worded as it is. The first point

made in the definition is that for a new to be a fa'il it must follow the verb whose subject it is because, if the new precedes the verb, it is not a fa'il but a mubtada'; this is a point commonly made clear by the arrangement. The second point rade in the definition is that for a new to be a fa'il it must be the subject of a verb in the active voice (fi'l alladhi buniya li-l-fa'il). By this condition Ibn al-Sarraj distinguishes between the fa'il (active subject) and the maf'ul alladhi law yusamma fa'ilwhu (passive subject). On this point it may be mentioned that Ibn Ya'ish expressly criticises definitions of the fa'il which are so worded as to differentiate it from the maf'ul alladhi law yusamma fa'ilwhu:

و بعمنهم يقول في وصفه: كل اسم تقدمه فعل فير مفير عن بنيت وأسندت ونسبت ذلا النعل إلى عن بنيته وأبيد بقوله: فير مفير عن بنيته الانفصال من فعل ما لم يسم فاعله و لا عابة إلى الاهتراز من ذللا لأن الفعل إذا أسند إلى صفعول فحد: فرب زيد ، وأكرم بكر . معار ارتفاع من جهة ارتفاع الفاعل إذ ليس من شرط الفاعل أن يكون مع هيدا للفعل أو ليس من شرط الفاعل أن يكون مع هيدا للفعل أو مؤثرا فيه .

The third point which Ibn al-Sarraj makes in his definition of the <u>fa'il</u> is that in defining whether a neum is a <u>fa'il</u> or not, no distinction is to be made as to whether the neum is a true <u>fa'il</u> or not. This stipulation arises out of Ibn al-Sarraj's classification of verbs into what are true verbs and what are not. What Ibn al-Sarraj complians as true verbs will be dealt with later when discussing the question of the division of verbs into transitive and intransitive, but it is necessary to discuss verbs which are not true verbs in dealing with the <u>fa'il</u>. Ibn al-Sarraj's particular classification of

this sort of verb is not one which seems to be followed in later works but it is of interest for illustrating problems which the term  $\underline{fa'il}$  itself presented to the grammarians. The verbs which are not real verbs are of three types: <sup>16</sup>

فالضرب الأول: أفعال مستمارة للاختصار وفيها بيان أن فاعليها في الحقيقة صفعولون نحو: مات زيد، وسقط الحائط، ومرض بكر.

والخرب النان: أنعال في اللفظ وليست بأفعال مقيقية وإنا تعل في الزمان فقط وذلك قولك: كان عبد الله أخاك، وأصبح عبد الله عاقلا. ولست تخبر بنيئل فَمَكَ ، إنها تخبر أن عبد الله أخوك فيها منى وأن الصباح أن عليه وهوعاقل.

والضرب النالث: أنعال منقولة, تراد بها فيرالفاعل الذي جعلت له نحو قولك؛ لا أرينتلاً هاهنا. فانهى إنما هو للمتكلم كأنه ينهى نفسه في اللفظ وهو للمخاطب في المعنى. و تأويله: لا تكونن هاهنا فإن من هضرن رأيته . وسئله قوله تعالى: وَلَا تَمُوتُنَّ إِلَّا وَأُنْتُمْ مُسْلِمُونَ. لم ينهم عن الموت في وقت لأن ذلك ليس لهم ينهم عن الموت في وقت لأن ذلك ليس لهم نقد يمه وتأخيره ولكن معناه : كونوا على الإسلام ، فأن الموت لا يدمنه فمتى صادفكم عليه . فإن الموت لا يدمنه فمتى صادفكم عليه . وهذا تعيير أبي العباس (المبرد) رهمه الله .

The first class of unreal verb is illustrated by examples like من المال , مات ربد , and بمن المال , مات ربد , and بمن المال . It would clearly be incorrect to describe the subject in any of these examples as a fa'il, if

due regard is given to the literal sense of that word. Indeed, Ibn al-Sarrāj observes that the <u>fā'il</u> in such examples is really a <u>maf'ūl</u>. However, it would be incorrect to call the subject in these examples the <u>maf'ūl alladhī lam yusamma fā'iluhu</u> because this would imply that there is a definable, though unspecified <u>fā'il</u>, when in fact the question of the true <u>fā'il</u> is more one for philosophy or theology than one for the grammarians. Although it would be fair to call the <u>fā'il</u> "unreal" for this reason, there does not seem to be a strong case for classifying the verbs themselves in the above examples as "unreal".

The second class of unreal verb consists of kana and analogous verbs. It was generally accepted by the grammarians that these verbs are not true verbs but resemble verbs in outward form and behaviour and because of this they are called af'al nagisah or, less commonly, af'al 'ibarah. 18 Although Ibn al-Sarraj undertakes a classification of verb types to assist in explaining his definition of the fa'il, the subject of a verb like kana is not strictly referred to by the grammarians as a fa'il but rather as the ism kana. Ibn al-Sarraj himself, as has been mentioned, classifies it as a mushabbah bi-l-fa'il not as a true fa'il. Abu Hayyan does point out, however, that the grammarians do loosely refer to the subject and predicate of kana and analogous verbs as the fa'il and the maf'ul. 19

The introduction by Ibn al-Sarraj of a classification of the various types of verb into his discussion of the fa'il reflects the

matical subject of a sentence. In English this problem does not really arise because the single expression "subject" can be used in analysing any type of sentence. The Arab grammarians called the subject of nominal sentences the <u>mubtada</u>, but the term <u>fā'il</u> was only just one possible term for the subject of verbal sentences. The Arabic term <u>fā'il</u> will translate literally into English as "doer" or "agent" or, if an exact technical expression is wanted, as "active subject". The Arab grammarians and, as the <u>Usul</u> shows, Ibn al-Sarraj in particular were always to a greater or lesser degree sensitive to the underlying meaning of the term <u>fā'il</u> and did not use it as a comprehensive term for the subject in verbal sentences.

a <u>fa'il</u> in the strict sense of a "doer", as the sentence in the passive voice "illustrates. The grammarians considered that such a sentence implies that someone struck Zayd, say 'Amr: 'implies that such the object of the action of the verb as it is in the active sentence. Accordingly, the subject of the passive sentence is called the <u>maf'ul</u> alladhi lam yusamma fa'iluhu and this can be translated into English as the "passive subject". Because the Arab grammarians felt that there was a real difference between the active and passive subject, Ibn al-Sarraj is careful to word his definition of when a noun is an active subject in such a way as to differentiate it from the passive subject.

The use of the term <u>fa'il</u>, as has been mentioned, is also unsuitable with regard to its underlying meaning of "doer" for the subject of <u>kana</u> and analogous verbs, and so the term <u>ism</u> is used as in <u>ism kana</u>, for instance. Ibn al-Sarraj recognises that the use of the term <u>fa'il</u> without qualification for the subject of the verb <u>kana</u> and analogous verbs is unsuitable. Accordingly, he calls it the <u>mushabbah bi-l-fa'il</u>

which might be rendered into English as the "quasi active subject".

The subject of kana and analogous verbs is only one type of quasi active subject and a second type is formed by the subject of sentences introduced by the particle ma in the Hijazi use together with other negative particles behaving similarly. On In an example like where the negative particle follows the Tamimi usage, the sentence consists of a mubtada and a khabar and the presence of the negative particle, which here has no power of government, does not affect the construction. However, in an example like which here has no power of government, does not affect the negative particle in the Hijazi usage functions like the verb of negation laysa, and accordingly the subject of the sentence is, in the terminology of Ibn al-Sarraj, a quasi active subject just like the subject of the verb laysa.

In the classification of types of verb which Ibn al-Sarrāj undertakes in his section on the <u>fā'il</u> he divides verbs into two great classes: those which are true verbs and those which are not; and his treatment of the latter class has been dealt with above in discussing the <u>fā'il</u>. His treatment of true verbs consists simply of a classification of verbs according to their transitivity and intransitivity and this is a subject which he discusses further in his chapter on the <u>maf'ūl bihi</u>. <sup>21</sup> The establishment of an exact classification of verbs according to their transitivity and intransitivity is a topic to which Ibn al-Sarrāj gives much more attention than is generally given in later works like the <u>Sharh al-mufassal</u> of Ibn Ya'īsh. Dealing in his section on the <u>fā'il</u> with the further classification of true verbs Ibn al-Sarrāj writes:

والحقيقي ينقسم قسمين : أهدها أن يكون الفعل لا يتعدى الفاعل إلى ما سواه ولا يكون فيه دليل على مفعول نحر: قمت وقعدت ، والآخر أن يكون فعلا واصلا إلى اسم بعد اسم الفاعل.
والفعل الواصل على فنريين: فضرب واصل مؤثر نحو: ضربت زيدا، وقتلت بكرا: والضرب الآخر واصل إلى الاسم فقط غير سؤثر فيه فحو: ذكرت زيدا، وصدهت عمرا، وهجوت بكرا. فإن ذكرت زيدا، وصدهت عمرا، وهجوت بكرا. فإن هذه تتعدى إلى الحي والميت والشاهد والفائب، وإن كنت إنما شدح الذات وتذمها إلا أنها غير مؤثرة.

وسنها الدفعال الداخلة على الدبتداء والخبر والخبر والحاسبة عن الفاعل بها هبس في نفسه أرتيقنا فبر صور نرة بالمفعرل، ولكن أخبار الفاعل بها وقع عنده نحد: فلننت زيدا أطالا، و علمت زيدا فير الناس.

The division of verbs into transitive and intransitive is a very basic one but Ibn al-Sarrāj goes further by dividing the former into those which can be said to have a tangible effect on their direct object and those of which this cannot be said. This distinction is also made briefly by al-Mubarrad in the Muqtadab. 25 In English to distinguish between the fi'l al-mu'aththir and the fi'l ghayr al-mu'aththir it would be neccessary to talk perhaps of "physical" and mental" verbs. Ibn Ya'ish also distinguishes between the two sorts of transitive verb using the term 'ilai rather than mu'aththir. Of the transitive verb he writes:

یکون علاها أونیر علاج فالعلاج ما بفتقر فی ایجاده ای استعمال هارههٔ أو نحوه نحو: ضربت زیدا، وقتلت مکرا و فیرالعلاج ما لم یفتقر الی

ذلا بل يكون عايتملق بالقلب نحو: ذكرت زيدا وفهمت الحديث.

When Ibn al-Sarraj discusses the  $\underline{\text{maf}}^4\underline{\text{ul bihi}}$  he again returns to the question of the classification of transitivity and intransitivity and he goes into further detail:  $^{25}$ 

قد تقدم قدلنا في المفعول على الحقيقة أنه المحدر. ولها كانت هذه تكون على فترين: فترب سنها بلاقى سَسنًا ديوز فيه وفرب منهما لديدق سنا ود يؤنزف فسي الفول الملاقي سمريا وما لا ملافی فرستمد فأما النمل الذي هو فيرستمد فهر الذي لا يلاقي مصدره مفعولا نحو: قام ، والمم ، ... والأنمال التي لا تتمرى في ما كان منها ملقة أو مركة للحسم في ذاته و هيئة له أو نماد من أفعال النفس غير منشبت بشيء هارج عنها. أما الذي هو هناف فنو : الود . واحم .... وأما مرك الجسم بغير ملاظاة لني و آخر فنيو: قام ، وقعد .... ألاترى أن هذه الأنمال معمونة لحركة الجمم وهيئت في ذاته. فإن قال قائل: فلا بد لهذه الأنمال من أن تلاقي المكان وأن تكون فع ، قيل : هذا لا بد سنه لكل فعل والمتمى وغير المتمى في هذا سراء ... لأن الفعل يصنع ليدل على المكان كما صيم ليدل على المصدر والزمان. وأما أفعال النمس التي لا تتمداها فنحو كرم ، وظرف ... وأما الفعل الذي يتمرى فكل هركة للجسم كانت ملاقية لغيرها وما

أشبه ذلا من أفعال النفس وأفعال الحواس من الخسس كلها منعدية ملاقية نحر: نظرت ، و منه و منه من و منه ما في معانيهن فهو منعد وكذلك هرك الجسم إذا لاقت شيئا كان الفيل من ذلك متعديا ، نحر أتيت زيدا ، ووطئت بلدك ودارك .

Ibn al-Sarrāj is, of course, inaccurate when he writes at the beginning of the above passage that transitive verbs beginning of the above passage that transitive verbs because, as he mentions in the passage quoted previously, not all transitive verbs can be said to have an effect on their direct object. The way in which Ibn al-Sarrāj deals with the question of transitivity and intransitivity in verbs is of interest because he considers this to depend on whether or not their respective infinitives are transitive. The Basran grammarians held that verbs are derived from their respective infinitives which are more basic than the verbs themselves and, accordingly, Ibn al-Sarrāj attributes a verb's power to govern a direct object to the power of its infinitive so to do. Ibn Ya'īsh makes this same point clear when he defines transitivity:

ومعنى التعدى أن المصدر الذى هو صدلول الفعل، وهو فعل الفاعل، على ضربين: ضرب منهما يلاقى نشيئا وية ترفيه فيسسى متعديا , وضرب منهما لا يلاتى نشيئا فيسسى فير متعد . فكل هركة للا يلاتى نشيئا فيسسى فير متعد . فكل هركة للحسم كانت ملاقية لغيره سميت متعدية وكل هركة له لم تكن ملاقية لغيره كانت لا زمة أى حى لازمة للفاعل لا تنجا و زه ، نحو : قام وقعد .

In a later chapter the question of Ibn Ya'ish's incorporation of material from the <u>Usul</u> into his <u>Sharh al-mufassal</u> will be dealt with in detail, <sup>27</sup> but it is worth noting here that Ibn Ya'ish does seem to be influenced in this passage by his reading of the <u>Usul</u>. Just like Ibn al-Sarraj he also makes the incorrect statement that all transitive verbs have an effect on their direct object. In addition, Ibn Ya'ish also uses the verb <u>laga</u> as a non-technical term to explain the technical term <u>ta'adda</u> and similarly he uses the idea of movement (harakah) in explaining transitivity. However, Ibn Ya'ish's analysis of types of verb is not as thorough as that of Ibn al-Sarraj.

In the preceding part of this chapter Ibn al-Sarraj's classification of the uses of the nominative case and the varieties of verb has been examined and it is also worth considering his classification of the uses of the accusative case. Ibn al-Sarraj divides the various uses of the accusative into a class which can be called the maf'ul and a class which can be called the mushabbah bi-l-maf'ul. 28 He draws up a classification of the mushabbah bi-l-maf'ul in a manner which is very much his own, although other scholars do deal with various of the considerations which underlie his classification. The grammarians as a whole did consider that certain uses of the accusative case could be classified as mafa'il, namely the maf'ul mutlag, maf'ul bihi, maf'ul fihi, maf'ul lahu, and the maf'ul ma'ahu. They also considered that other uses of the accusative resembled the maf'ul but they did not go as far as formally putting them into a class called the mushabbah <u>bi-l-maf'ul. The term maf'ul translates into English as "object" and </u> the term mushabbah bi-l-maf'ul could be translated as "quasi-object", just as Ibn al-Sarraj's term mushabbah bi-l-fa il could be translated as "quasi active subject".

According to Ibn al-Sarraj the <u>mushabbah bi-l-maf'ul</u> divides up into two classes: 29

والمنتب بالمفهول بنقسم على قسمين فالقسم الأول قد يكون فيه المنصوب في اللفظ هو المرفوع في اللفظ عو المرفوع في اللفظ في المنصوب في اللفظ فير المرفوع والمنصوب بعن المرفوع والمنصوب بعن المرفوع .

The first of these two classes can be further divided into three varieties: 30

هذا النوع بنقسم على ذلانة أفترب فمنه ط العامل فيه فعل هقيقي , ومنه ط العامل فيه شيء على وزن الفعل ويتصرف تصرفه وليس بفعل في الحقيقة , و سنه ط العامل فيه هرف هامد فير ستصرف

The first of these three types consists of the <u>hal</u> and the <u>tamyiz</u>.

The al-Sarraj does not explain very clearly in the case of the <u>hal</u>

why the accusative in outward form may represent what is nominative in

meaning, and all he writes in explanation is: 31

فأنسبه : ها عبد الله راكبا , فرد، عبد الله رهلا. وراكب هم عبد الله , ليس هم غيره.

Ibn Ya'ish makes the same point much more clearly when he explains why the accusative of the <u>hal</u> cannot be said to be a true <u>maf'ul</u>? 32

وي يدل أنها ليست مفعولة أنها هي الفاط في معنى وليس فيره. فالراكب في : جاء زيد راكبا ، هو زيد . وليس المفعول كذلا بل لا يكون إلا فير الفاعل أو في هكمه نحو: فهرب زيد عمرا.

However, it has to be said that the <u>hal</u> does not neccessarily qualify the subject of the sentence and that the <u>sahib al-hal</u> may be some other term. In the case of Ibn al-Sarraj this objection would appear

to be met by the fact that he states that the class of <u>mushabbah</u>

<u>bi-l-maf'ul</u>, into which the <u>hāl</u> falls, is one in which the accusative in outward form <u>may</u> be (<u>in the land</u>) nominative in meaning. However, the point of classifying the <u>hāl</u> in this manner would seem to be to distinguish the <u>hāl</u> from the <u>mafā'īl</u> proper, rather than to give a universally valid description of it.

The second variety of <u>mushabbah bi-l-maf ul</u> where the accusative in outward form is nominative in meaning and the regent governing it is a verb proper, is the accusative of the <u>tamyiz</u>. This description, however, does not apply to the use of the <u>tamyiz</u> in enumeration and measurement and in the present discussion such usages are specifically excluded. 33 In explanation of his classification Ibn al-Sarrāj writes: 34

الأسماء التى تنتمب بالتهييز والعامل فيها فعل أو صفى فعل والمفعول هو فاعل فى المعنى وذلا قوللا: قد تفقأ زيد شجا و تصبب عرقا و طبت بذلا نفسا واستلا الإناء ماذ وفنقت به فرعا فالماء هو الذى ملا الإناء والنفس هى التى طابت والعرق هو الذى ملا الإناء والنفس هى التى المفعول وهو فى المحنى فاعل وكذلا ما جا فى المفعول وهو فى المحنى فاعل وكذلا ما جا فى معنى الفعل وقام مقامه نحو قولك: زيد أفرههم عبدا ، وهو أهسنهم وجها فالفاره فى الحقيقة هو العبد والحسن هو الوجه ، إلا أن فولا أفره وأهسن فى اللفظ لزيد و فيه فنييره ، والعبد فير وأهسن فى اللفظ لزيد و فيه فنييره ، والعبد فير نيد والوجه إنها هو بعضه إلا أن الحسن في العرب والفراهة للعبد والفراهة المعبد والفراهة للعبد والفراهة المعبد والفراهة للعبد والفراهة المعبد والفراهة والفراهة المعبد والفراهة المعبد والفراهة المعبد والفراهة المعبد والفراهة والفراهة والفراهة المعبد والمعبد والفراهة والفراه والفراه والفراهة والفراه والفرا

Although this analysis of tamyiz as it stands supports Ibn al-Sarraj's way of classifying this particular use of the accusative case, it is an analysis which can appear over-simple when compared with that of later grammarians. Abu Hayyan, for instance, writes in the Manhaj al-salik that the grammarians divide uses of the tamyiz into several categories: 35

... یکون منفولا و مشبها بالمنقول والمنقول و ندن أقسام منفول من فاعل و ذلا نجو: تصب زید عرفا ، واشتعل الرأس شیبا وطاب زید نفسا . أصله تصبب عرق زید ، واشتعل شیب الرأس ، وطابت نفس زید . فأسندنا الفعل الی مجرور وانتصب الفاعل تمییزا ... و منقول من مفعول و ذللا نجو قوله تعالی : و فجرنا الارن فی عیون الارن ، وغرست شجرا ، و هفرت الدار بئرا . اصله : فجرنا عیون الارن ، وغرست شجرا نی الارن ، و فرست شجرا نی الارن ، و فرست شجرا نی الارن ، و فرست شجرا نی الارن ، فعو : زید أحسن منلا وجها . أصله : وجه زید فولهم : هبذا زید رجلا ، و کفی بزید ناصرا ، و داری قولهم : هبذا زید رجلا ، و کفی بزید ناصرا ، و داری فلف دارلا فرسخا .

This passage provides a more highly developed analysis of tamyiz constructions than Ibn al-Sarraj does, although Abu Hayyan does mention that there are certain scholars who do not accept the tamyiz mangul min al-maf'ul and some treat the tamyiz mangul min al-mudaf as having the meaning of an original fa'il which would accord with Ibn al-Sarraj's analysis of tamyiz constructions of this type. 36 Even if tamyiz

constructions are analysed along the lines laid down by later scholars, the same point may be made as was made in the case of the <u>hal</u>, namely, that in the case of the <u>mushabbah bi-l-maf'ul</u> the accusative in outward form may be nominative in meaning. However, there is no evidence that Ibn al-Sarraj would analyse the <u>tamyiz</u> as other than standing in for an implied fa'il.

The two other types of <u>mushabbah bi-l-maf'ul</u> of the class where the accusative in outward form may be nominative in meaning are the predicate of <u>kana</u> and analogous verbs and the subject of <u>inna</u> and analogous particles. <sup>37</sup> It is clear why Ibn al-Sarraj does not consider these to represent true <u>mafa'il</u> and there is no need to discuss them further here but they will be mentioned in a later chapter. <sup>38</sup>

The second class of mushabbah bi-l-maf'ul consists of only one item and this is the accusative used in exception after illa in positive sentences and as an alternative to the badal construction in negative sentences. 39 Ibn al-Sarraj describes this variety of mushabbah ما يكون المنصوب فيه في اللفظ غير المرفوع والمنصوب sas ما يكون المنصوب الرفور . 40 The meaning of the first part of this description is clear, ما يكون المنصوب فيه في اللفظ غير المرفوى , because it serves to distinguish this type of mushabbah bi-l-maf'ul from all other types. Ibn al-Sarraj does not suggest that the exceptive accusative admits of being interpreted as nominative in meaning unlike the other types of mushabbah bi-l-maf'ul. The second part of the description, is somewhat less clear but can be explained بعض المرفوع by using an illustration like إلا زيدا. Here the term زيد in the accusative is to be thought of as part of what is denoted by the in the nominative until it is expressly excepted from it. Ibn Ya'ish takes much the same consideration into account when he explains thy the term in the accusative after illa cannot be treated as a true maf'ul: 41

إنما قلنا أنه مشبه بالمفعول ولم نقل أنه مفعول لأن المستنتى أبدا بعن المستنتى منه والمفعول فير الفاعل.

However, the statement by Ibn al-Sarraj, is only valid if the <u>mustathna minhu</u> is a term in the nominative, but the point of this remark may be to show that there is some affinity between this <u>mushabbah bi-l-maf'ul</u> and the others where the accusative in outward form may be nominative in meaning.

In what has preceded, the work of Ibn al-Sarrāj in classifying and analysing various grammatical usages has been examined and similar to this in many ways is his attention to defining the parts of speech. However, in defining the parts of speech Ibn al-Sarrāj takes into account considerations that are not purely grammatical, but which stem rather from logic. This is because one of the developments which came to take place in grammatical thought was that philosophy and, in particular, logic came to exert an influence on the grammarians in their outlook and the ways in which they explained various matters. <sup>42</sup> One particular area in which logic was influential was in the definition of the parts of speech.

Whether or not Arabic grammar owes to logic its tripartite division of the parts of speech into noun, verb, and particle does not matter for the purposes of this present discussion, but what is of relevance is that the grammarians came to take into account what the logicians had done towards defining the parts of speech. The different types of word which occur in speech were not felt to be something based purely on the analysis of the Arabic language. Al-Mubarrad, for instance, writes: 43

فالكلام كله: اسم فعل وطرف ها المحنى لا يخلو الكلام - عبيا كان أو أكسيا - من هذه التلانة.

In the light of the idea of the universality of the tripartite division of speech it is not unnatural that the grammarians should have drawn on logic in discussing the parts of speech and their definition because logic is a subject which claims universal validity and in it too the parts of speech and their definition are discussed.

In the <u>Kitab Sibawayh does not give a formal definition of the noun but merely gives some examples of nouns:</u>

44

In the <u>Muqtadab al-Mubarrad goes into more detail and offers the following definition of the noun: 45</u>

أما الأسماء فما كان واقعا على معنى نحو: رجل وفرس وزيد وعمرو، وما أننب ذلا. وتعتبر الأسماء بوامدة: كل ما دخل عليه هرف من هروف الجرفهو اسم، وإن امتنع من ذلا فليس باسم.

In defining a noun as that which can be made subject to a preposition, al-Mubarrad produces a definition which is framed in purely grammatical terms and this is a mode of definition of common occurence in works of grammar.

Not all definitions of the noun are of this order. Al-Sirafi, a pupil of Ibn al-Sarrāj, is quoted by Ibn Ya ish as giving the following definition of the noun which may be taken as typical of definitions by later grammarians: المنت دلت على صنى في نسما من يراقران على المناز على المناز

وقرله: يدل على صنى في نفسها، فعل اهترزبه من الحرف لأن الحرف يدل على صنى في غيره، وقوله: من غير اقتران برطان محمّل فعل ظان جمع به المصادر إلى الاسماء وصنع الأفعال أن تدخل في هد الأسماء ولان تدخل في هد الأسماء لأن الأهدان تدل على أزمنة مبلهة إذ لا يكون هدنا إلا في زطان ولالة الفعل على زطان صلوم إما ما في وإما غير طاف.

الاسم ما دل على صنى مفرد وذلك المعنى يكون شخصا الاسم ما دل على صنى مفرد وذلك المعنى يكون شخصا وغير شخص فحد: دجل وفرس وهجر وجلد وعمر وبكر. وأما ما كان فير شخص فنحم :الضرب والأكل والظل والعلم واليوم والليلة والسامة.

Neither here nor later in his discussion does Ibn al-Sarrāj introduce the idea that a noun is المنافق في المنافق المن

وإنها قلت: ما دل على هنى مفرد ، لأفرق بينه وبين الفعل إذ كان الفعل يعل على حمنى وزمان وذلك الزمان إما مافى وإما حاضر وإما مستقبل .

Ibn al-Sarraj does mention that the noun, unlike the verb, does not indicate a notion in in indicate a notion indicate a notion in indicate a notion indicate a no

grammarians) to the generally accepted definition of the noun and he explains the word seems as if it would be unknown in this sense to his readers: 51

فإن قلت: إن في الدّسيا، مثل اليوم والليلة والسابة وهذه أزمنة فيا الفرق بينها وبين الفعل قلنا: الفرق أن الفعل ليس هم زمان فقط فاليوم معنى مفرد للزمان ولم يوضع مع ذلك لمحنى آهر, ومع ذلك أن النعل قد قسم بأقسام الزمان الثلاثة: الماضى والحافر والمستقبل فاذا كانت اللفظة تدل على زمان فقط فهى اسم ، وإذا دلت على معنى وزمان محصل فهى فمل وأدنى بالمحمل الماضى والحافر والمستقبل.

The relation of Ibn al-Sarrāj's approach to the definition of the noun to the ideas of the logicians is not difficult to show. In the Kitāb al-alfāṣ al-musta'malah fī l-mantiq al-Fārābī defines the noun as: كالفظ مفرد دال على الحني من ذير أن يمل بنات الربان المن المنظ مفرد دال على الحني من ذير أن يمل بنات المن المنظمة والمن المنظمة والمن المنظمة والمن المنظمة والمن المنظمة والمن المنظمة والمن المنظمة والمنافعة و

When discussing the definition of the noun in the Idah fi 'ilal al-nahw, al-Zajjāji (d. ca. 949), who was a pupil of both al-Zajjāj and

The al-Sarraj, mentions that the grammarians and the logicians define it in a different manner although some of the grammarians incline to the logicians' view. Al-Zajjājī himself frames his own definition to meet the needs of grammar but admits the validity of the logicians' definition: 55

الاسم في كلام العرب ما كان فاعلا أو مفورلا أو واقعا في ميز الفاعل والمضورل به . هذا الحد اهل في مقايس الني وأوفاعه وليس يخرج عنه اسم البية ، ولا يدخل فيه ط ليس باسم . وإنا قلنا في كلام الحرب لأنا له نقمد وقلبه نتكم ، ولائن المنطقين وبمن الخرين قد مدره مدا خارها عن أوضاء النحر. فقالوا: الاسم صوت موضور دال ماتفاق على معنى غير مقرون برمان. وليس هذا من ألفاظ النحويين ولا أوضاعهم. وإنها هو من كلام المنطقيين، وإن كان قد تعلق به ممانة من النويين . وهو محير على أوماع المنطقين ومذهبهم لأن فرهم فير فرهنا وصفراهم غير مفزانا . وهو عندنا على أوضاء النحر فير محير لأنه يلزم منه أن يكون كثير من الحروف أسهاء لأن من الحروف ما يدل على محنى دلالة غير مقرونة بزمان , نحو: إن ولكن وما أشده ذلك.

This passage confirms that by the time of al-Zajjaji some of the grammarians were taking over the logicians' definitions of the noun although the majority appear to have based their definitions purely on the needs of grammar. Although Ibn al-Sarraj's ideas on the definition

of the noun are influenced by logic, yet they are not so profoundly influenced as the sort of definition based on logic which al-Zajjājā quotes above. However, it does appear from the evidence available that Ibn al-Sarraj was the first grammarian who can be definitely attested as defining the noun along lines laid down by the logicians.

The definition of the verb is easier to deal with because the grammarians tend to discuss it in less detail, as is the case with the definition of the particle as well. However, Sibawayh gives more attention to defining the verb than to defining the other two parts of speech: <sup>56</sup>

وأما الفعل فأمثلة أُهنت من لفظ أهدان الأسماء وبنيت لها منى ولها يكون ولم يقع وما هوكائن ولم ينقطع.

Although Sibawayh's view on the verb was discussed by the scholars and is important for indicating that he held the general view of the Basrans that the verb is derived from the infinitive, nevertheless, it has little direct bearing on the way in which the verb was defined by later scholars.

As typical of later definitions of the verb may be cited that by

The Ya'ish:

57 The al-Sarraj's definition of the verb largely corresponds with this except that, as was the case with the noun, he does not include the expression is to distinguish the verb from the particle:

58

الفعل ما دل على معنى وزمان وذلا الزمان إما ما في واما ها فتر وإما مستقبل وقلنا: وزمان النفرق بينه وبين الدسم الذي يدل على معنى فقط.

This definition by Ibn al-Sarraj also links up with the ideas of the

logicians; al-Farabi, for instance, writes in definition of the verb in the Kitab al-alfaz: عنى المناة مغردة تدل على معنى وعلى زمان. 59

It is significant that Ibn al-Sarraj considers that a verb conveys both a notion and a time. According to Ibn Faris, al-Kisa i had defined a verb as: المناق ا

الحرف ما لا يجوز أن يخبر عنه كما يخبر عن الاسم. ألا ترى أنلا لا تقول: إلى سنطلق كما تقول: الرجل سنطلق ، ولا: عن ذاهب كما تقول: زيد ذاهب. ولا يجوز أن يكون هبرا. لا تقول: عمرو إلى ، ولا: بكر عن . فقد بانأنا لحرف من الكلم الثلاثة هو الذى لا يجوز أن يخبر عنه ولا يكون هبرا.

## Notes to Chapter II

- 1 Al-Sirafi, p. 37.
- 2 Ibn al-Nadim, p. 52; al-Sirafi, p. 39.
- 3 See pp. 10-11 above.
- 4 Al-Qifti, vol. iii, p. 149.
- 5 See p. 13 above.
- 6 For al-Marzubani see Kahhalah, vol. ii, pp. 97-8.
- 7 For the logical term "division" see A. A. Luce, Logic (London, 1958), pp. 31-2.
- 8 Ibn Abī Uṣaybi'ah, '<u>Uyun al-anbā</u>', 2 vols (Cairo, 1882), vol. ii, p. 136.
- 9 Ibn al-Sarraj, vol. i, pp. 61-2, 174-5, 399, 498.
- 10 Ibid., p. 60.
- 11 Ibid., p. 62.
- 12 Ibid., p. 92.
- 13 See pp. 61-3 below.
- 14 Ibn al-Sarraj, vol. i, p. 81.
- 15 Ibn Ya'ish, vol. i, p. 74.
- 16 Ibn al-Sarraj, vol. i, pp. 82-3. In 1. 15 of this passage has been substituted for in the printed text.
- 17 See also al-Mubarrad, vol. iii, p. 188.
- 18 Ibn Ya'ish, vol. vii, p. 89.
- 19 Abu Hayyan, pp. 125-6.
- 20 Ibn al-Sarraj, vol. i, pp. 106-13.
- 21 Ibid., pp. 202-3.
- 22 Ibid., pp. 81-2.
- 23 Al-Mubarrad, vol. iii, p. 188.
- 24 Ibn Ya'ish, vol. vii, p. 62.
- 25 Ibn al-Sarraj, vol. i, pp. 202-3.
- 26 Ibn Ya'ish, vol. i, p. 124.
- 27 See pp. 166-79 below.

- 28 Ibn al-Sarraj, vol. i, pp. 189, 257.
- 29 Ibid., p. 257.
- 30 Ibid., pp. 257-8.
- 31 Ibid., p. 258.
- 32 Ibn Ya'ish, vol. ii, p. 55.
- 33 Ibn al-Sarrāj, vol. i, pp. 189, 374-5. See also pp. 65-6 below.
- 34 Ibn al-Sarraj, vol. i, p. 268.
- 35 Abu Hayyan, pp. 224-5.
- 36 Ibid., pp. 223-5.
- 37 Ibn al-Sarraj, vol. i, pp. 276-7.
- 38 See pp. 70-1 below.
- 39 See pp. 67-70 below.
- 40 Ibn al-Sarraj, vol. i, pp. 257, 342.
- 41 Ibn Ya'ish, vol. ii, p. 77.
- 42 On the general question of the influence of Greek grammatical thought and logic see C. H. M. Versteegh, Greek elements in Arabic linguistic thinking, Leiden, 1977.
- 43 Al-Mubarrad, vol. i, p. 3.
- 44 Sibawayh, vol. i, p. 2.
- 45 Al-Mubarrad, vol. i, p. 3.
- 46 Ibn Ya'ish, vol. i, p. 22.
- 47 Ibid., pp. 22-3.
- 48 Ibid., p. 22.
- 49 Ibn al-Sarraj, vol. i, p. 38.
- إذا has been substituted for الله passage إذا has been substituted for الله in the printed text.
- 51 Ibid., pp. 38-9.
- 52 Al-Farabi, <u>Kitab al-alfaz al-musta malah fi l-mantiq</u> (Beirut, 1968), p. 41.
- 53 Al-Farabi, Sharh al-'ibarah (Beirut, 1960), p. 29.
- 54 Ibn Fāris, Ṣāḥibī fī fiqh al-lughah (Cairo, 1910), p. 51.

- 55 Al-Zajjaji, Idah fi 'ilal al-nahw (Cairo, 1959), p. 48.
- 56 Sibawayh, vol. i, p. 2.
- 57 Ibn Ya'ish, vol. vii, p. 2.
- 58 Ibn al-Sarraj, vol. i, p. 41.
- 59 Al-Farabi, Alfaz, pp. 41-2.
- 60 Ibn Faris, p. 52.
- 61 Al-Farabi, Sharh al-'ibarah, p. 33.
- 62 Sibawayh, vol. i, p. 2.
- 63 Al-Zajjaji, Jumal fi l-nahw (Paris, 1957), p. 17.
- 64 Ibn al-Sarrāj, vol. i, p. 43.

#### CHAPTER III

## REGENTS, THEIR CLASSIFICATION, AND RELATED THEORY

One of the most important of the concepts used by the Arab grammarians is that of the 'amil, a word which can be rendered into English as "regent". In definition of what a regent is Weil writes that it is "to express it in the way of the Arab grammarians a word, which, by the syntactical influence which it exercises on a word that follows, causes a grammatical alteration of the last syllable of the latter, i.e. a change of case or mood." Although the idea of the regent was basic to the thought of the Arab grammarians from the earliest times onward, nevertheless, systematic discussions of this topic are comparatively rare.

Among the various matters which Ibn al-Sarrāj discusses at the beginning of the <u>Usūl</u> is the regent and this discussion would appear to constitute one of the earliest systematic treatments of it. <sup>2</sup> The best known work devoted to the regent is the 'Awāmil al-mi'ah of 'Abd al-Qāhir al-Jurjānī (d. 1078) which was the subject of a number of late commentaries. <sup>3</sup> Prior to this Abū 'Alī al-Fārisī had written a work on the regent although it plainly did not have the same success. <sup>4</sup> In the <u>Ashbāh wa-l-nazā'ir</u> of al-Suyūtī there is a discussion of various points connected with the regent in which the author draws heavily on quotations culled from earlier works. <sup>5</sup>

In his discussion of the regent in the <u>Usul</u> Ibn al-Sarrāj classifies the various regents according to whether they are nouns, verbs,
or particles and he further subdivides these three classes in the manner
shown in the accompanying table. 6 Amongst the particles he even includes

#### TABLE 1

# THE DIVISION OF THE PARTS OF SPEECH INTO REGENTS ACCORDING TO IBN AL-SARRAJ 6

### A Nouns

- 1 The noun in the mubtada' and khabar construction
- 2 Nouns with verbal regimen
  - i The active participle
  - ii The assimilated adjective
  - iii The infinitive
  - iv Nouns with verbal force (e.g. ruwayda)
- 3 Nouns with the regimen of particles (i.e. the <u>mudāf</u> in the construct)

#### B Verbs

## C Particles

- 1 Particles governing nouns
  - i Prepositions
  - ii inna and similar particles
- 2 Particles governing verbs (i.e. entailing the use of the subjunctive and jussive moods)
- 3 Particles without regimen (e.g. interrogative particle a)

#### TABLE 2

# THE "HUNDRED REGENTS" ACCORDING TO AL-JURJANI 7

# A The 'awamil al-lafziyah (98)

- a The 'awamil al-sama'iyah (91 regents in 13 classes)
  - 1 Prepositions (17)
  - 2 inna and similar particles (6)
  - 3 ma and la functioning like laysa (2)
  - 4 Particles governing the accusative (7) (wa, illa, ya, aya, haya, ay, a)
  - 5 Particles governing the subjunctive (4)
  - 6 Particles governing the jussive (5)
  - 7 Particles governing the jussive used in conditional sentences (9)
  - 8 Expressions entailing tamyiz (4) (numerals 11-99, kam, kadha, ka'ayyin)
  - 9 Nouns with verbal force (asma' al-af'al) (9)
    (governing the accusative:- ruwayda, balha, dunaka, 'alayka, bayyahal, ha; governing the nominative:- hayhat, shattan, sar'an)
  - 10 kana and similar verbs (13)
  - 11 Verbs of appropinguation (4)
  - 12 Verbs of praise and blame (4)
  - 13 Verbs of doubt and certainty (7)
- b The 'awamil al-qiyasiyah (7 regents)
  - 1 The verb
  - 2 The assimilated adjective
  - 3 The active participle
  - 4 The passive participle
  - 5 The infinitive
  - 6 The first element (mudaf) in the construct
  - 7 The first element (<u>mumayyiz</u>) in the tamyiz construction

# B The 'amilan al-ma'nawiyan (2 regents)

- 1 Regent of the <u>mubtada</u> and <u>khabar</u>, i.e. <u>ibtida</u>
- 2 Regent of the imperfect tense, i.e. its taking the position of a noun without an express substantive regent

those which do not function as regents at all. For the purposes of comparison a table has also been drawn up listing the "hundred regents" according to al-Jurjani's reckoning so that this scholar's approach may be compared with that of Ibn al-Sarraj. The approach of al-Suyūtī to the regent in the Ashbah wa-l-naza'ir is completely different to that of Ibn al-Sarraj because the latter aims at providing a classification of the regents while the former aims mainly at setting down various rules as to how the regents function.

The 'Awamil al-mi'ah of al-Jurjani is an extremely terse résumé of the grammatical regents in Arabic and is so constructed as to yield a total number of one hundred regents. One of the main points about al-Jurjani's classification is that there are two broad types of regent, the 'amil lafzi and the 'amil ma'nawi and these two terms may be translated as "verbal regent" and "notional regent". In explanation of these two terms Weil writes: "Two kinds of regentia are distinguished, one which can be recognized externally (lafzi) and one which is only to be supposed logically, but which is not expressed (ma'nawi)." Al-Jurjani's further classification of the verbal regents into the two classes of sama'i and qiyasi does not find its way into the standard grammars of Arabic. If a regent is sama'i it is a lexically definable term such as the preposition bi, but if it is qiyasi it represents a class of terms such as the verb whose constituents cannot be exhaustively defined.

One of the main differences between Ibn al-Sarrāj's treatment of the regent and that of al-Jurjānī is that the former makes no reference to notional regents and verbal regents. Later grammarians like al-Zamakhsharī make use of these expressions but they are ones which seem to post-date the time of Ibn al-Sarrāj and are used neither by Sībawayh in the Kitāb nor by al-Mubarrad in the Muqtadab.

All regents except two are classified as verbal regents and these two are ibtida and the regent governing the imperfect indicative verb.

In the <u>Usul ibtida</u> is not specifically called a <u>ma'na</u>, a form of expression which in itself would prepare the way for the formal division of regents into verbal and notional. However, the idea that <u>ibtida</u> is specifically a <u>ma'na</u> does appear in the <u>Jumal</u> of al-Zajjājī, a pupil of Ibn al-Sarrāj: 11

اعلم أن الاصم المستدأ حرفوع والخبر إذا كان اسما واهدا مثله فهو حرفوع أبدا. وذلا قولا: زيد قامً، فزيد حرفوع لأنه مستدأ والابتداء معنى رفعه. وهو مضارعت للفاعل وذلا أن المبتدأ لابدله من هنبر ولا بد للخبر من مبتدأ يسند إليه. وكذلا الفعل والفاعل لا يستفنى أهدها عن صاهبه.

In the commentary on the <u>Jumal</u> by al-Jurjani's contemporary, Ibn Babshadh (d. 1077), <u>ibtida</u>' is discussed using the sort of terms found in al-Jurjani's classification: 12

أما الابتداء فعمنى لا لفظ وهو وصف قائم في المحمول وذلا الوصف احتمامك بالمبتدأ ومملك إياه أولا لثان يكون الثاني خبرا عنه فيهذا تفسير صفة الابتداء والمبتدأ لفظ وهوكل المم بدأت به لتخبر عنه بخيره صورى من الموامل اللفظية . فهذا معرفة الفرق بين الابتداء والمبتدأ والمبتدأ والمبتدأ معمول والابتداء تقريم والمبتدأ معمول . والابتداء تقريم والمبتدأ لفظ

Like the <u>Uşul</u>, the <u>Sharh al-jumal</u> of Ibn Babshadh is a work intended for beginners and, accordingly, it seems probable that Ibn al-Sarraj would have referred to notional and verbal regents if this classification had been current in his day for these expressions did become a basic part of grammatical terminology.

If the concept of the notional regent is taken to be one which only developed after the time of al-Mubarrad and Ibn al-Sarrāj, then statements made by Abu Ḥayyān about views which al-Mubarrad is said to have held must have an element of anachronism about them. One view which al-Mubarrad is said to have held about why the mubtada' is in the nominative is: ارتبع بالتربي با

فأما رفع المبتدأ فبالابتداء . وصفى الابتداد: التنبيد والتعربة عن العوامل غيره , وهو أول الكلام.

If al-Mubarrad had been in the habit of using the expression "verbal regent" as a contrast to "notional regent" one would have expected him to have introduced the expression here.

The other notional regent which the grammarians recognised was the regent governing the imperfect indicative. On this point al-Zamakh-shari writes: 15

هو في الارتفاع بعامل معنوى نظير المبتدأ وخبره. وذلا المعنى وقوعه بحيث يصح و توع الامم كقوللا: زيد يضرب, رفعته لأن ما بعد المبتدأ من مظان معية وقوع الأسهاء.

Ibn al-Sarrāj's explanation of the same point is: 16

الفعل برتنع بسوته موقع الانهاه ... ألا ترى أنك إذا قلت: يقوم زيد، جاز أن تجعل زيدا موضع يقوم فتقول: زيد يفعل كذا، وكذلك إذا قلت: عمره ينطلق ،فإنما ارتفع ينطلق لأنه و قع موقع أ هنوك إذا قلت : زيد أ هنوك.

Ma'na itself does not appear and the same applies to al-Mubarrad's treatment of the point in the Muqtadab. <sup>17</sup> Apart from this Ibn al-Sarraj's explanation is very similar to that of the later scholar, al-Zamakhshari, and for this reason it seems probable that the former would have indicated that the regent is notional if that expression had been current in his day.

In discussing the idea of notional and verbal regents reference has been made to <u>ibtida</u> and the nominal sentence, and this requires some further consideration. In defining the noun's power to act as a regent Ibn al-Sarrāj makes a tripartite division into the noun acting in the <u>mubtada</u> and <u>khabar</u> construction, the noun acting as a verb, and the noun acting as a particle. <sup>18</sup> In the case of the first and the third of these classes the validity of the classification depends on the particular views of Ibn al-Sarrāj to which all grammarians do not subscribe. On the noun in the first of these three classes, that in the mubtada and khabar construction, Ibn al-Sarrāj writes:

الاسم يعمل في الاسم على خلافة فبروب الضرب الأول أن يبنى عليه اسم مخله أو يبنى على اسم ويأخلف باجتماعهما الكلام وينتم ويفقدان المعوامل من فيرها نحوقوللا: عبد الله أخولا. فعبد الله مرتفع بأنه أول المبتدأ فاقد للعوامل ابتدأته لتبنى عليه ما يكون هدينا عنه وأهولا مرتفع بأنه الحديث الحبنى على الاسم الأول المبتدأ.

In this passage Ibn al-Sarraj does not explain exactly how one noun acts as a regent on another in the <u>mubtada</u>' and <u>khabar</u> construction

but, rather, he writes that the <u>mubtada</u>' is put into the nominative by the very act of functioning as a <u>mubtada</u>', which is the concept of <u>ibtida</u>' although he does not call it such here; the <u>khabar</u> is put into the nominative by the act of functioning as a predicate. This, however, does not involve one noun acting on another. A satisfactory explanation of the contention that one noun acts on another in this construction is given when Ibn al-Sarrāj gives special attention to the <u>mubtada</u>' at a later point in the <u>Usūl</u>: <sup>20</sup>

المستماط جردته من عواصل الأسماء ومن الأفعال والحروف وكان القصد فيه أن تجمله أولا لثان مستدا به دون الفعل يكون نانيه خبره ولا يستفى واهد منهما عن ماهيه وهما مرفوعان أبدا فالمستدا رفع بالدبنداء والخبر بهما , نحو قوللا: الله ربنا , وحهد نبينا .

This passage gives an explanation of Ibn al-Sarraj's view that in the <u>mubtada</u> and <u>khabar</u> construction one noun acts as a regent on another because he states here that the <u>mubtada</u>, which is a noun, together with ibtida act as the regent of the khabar.

If this is taken as the explanation of how one noun acts on another in this construction, this type of noun regent only exists providing that it is accepted that the <u>mubtada</u>' acts as a regent on the <u>khabar</u>. This view was rejected by scholars like Ibn al-Anbari and Ibn Ya'ish who held that the <u>khabar</u> was put into the nominative by <u>ibtida</u>' alone, acting, to be sure, through the medium of the <u>mubtada</u>'. In criticism of the view which Ibn al-Sarraj among others takes, and in defence of his own view, Ibn Ya'ish writes: 22

وذهب آخرون إلى أن الابتداء والمبتدأ جميما يعملان في الخبر . قالوا : لأنا وجدنا الخبر لا يقم

إلا بعد المبتدأ والابتدا؛ فوهب أن يعملا فيه. وهذا القول عليه كثير من البصربين ولا ينفلا من ضعف. وذللا من قبل أن المبتدأ اسم والأصل في الأسهاد أن لا تعمل وإذا لم يكون لها تأثير في العمل والابتدا؛ له تأثير، فإهنافة ما لا تأثير له إلى ما له تأثير ، لا تأثير له ويمكن أن يقال إن الشيئين إذا تركبا هدت لهما بالتركيب معنى لا يكون في كل واهد من أفراد ذللا المركب. والذي أراه أن العامل في الخبرهو الابتدا؛ وهده على ما ذكر كما كان عاملا في الحبرهو الابتدا؛ وهده على ما المبتدأ بلا واسطة و عمله في الخبر بواسطه المبتدأ بلا واسطة و عمله في الخبر بواسطه المبتدأ بلا واسطة و عمله في الخبر بواسطه المبتدأ المربد أن المهد عند وجود المبتدأ وإن لم يكن يصل في الخبر عند وجود المبتدأ وإن لم يكن

This passage from Ibn Ya'ish provides an excellent example of how grammatical thinking could become more complex. In the <u>Usul</u> Ibn al-Sarraj simply puts forward his view about the nature of the regent determining the case of the <u>khabar</u> and does not feel compelled to justify it. Later grammarians, however, thought more deeply about the point and were able to offer reasoned critiques of views like that of Ibn al-Sarraj and put forward in their place more subtly formulated views.

As has been mentioned above the existence of the third type of noun regent which Ibn al-Sarraj classifies also depends on the way in which a particular construction is explained. This type consists of nouns with the governing power of particles and in fact this refers to the construct. Ibn al-Sarraj states that the construct either indicates possession and is equivalent to the use of the particle <u>li</u>, or it

indicates of what sort a particular thing is and is equivalent to the use of the particle min. <sup>23</sup> In the course of discussing the latter case Ibn al-Sarraj specifies exactly what the regent is when he explains the relationship of the expressions عن من الماء عن الماء عن

وهذا لا فرق فيه بين إضافته بغير من وبين إضافته بعن وأنما هذ فوا من هنا استخفاظ فلمنا هذ فوها التق الاسمان فخفض أهد مها الاخر أذا لم يكن النان فبما عن الأول ولا صفة له.

However, this type of noun regent does not exist if the view of certain later scholars on the regent governing the genitive in the construct is followed. Ibn Ya'ish, for instance, when analysing the expressions and is and is writes: 25

فالعامل هنا هرف الحرالمقدر والتأثيرة وتقديره علام لزيد و هاتم من فقة . لا ينفلاً كل إضافة هقيقية من تقدير أهد هذين الحرفين ولولا تقدير وهود الحرف الحذكور لها هاغ الجر . ألا ترى أن كل واهد من المضاف والمضاف اليه اسم ليس له أن يعمل في الكاهر لانه ليس عمله في أهدها أولى من العكس . وانها الخففي في المضاف إليه بالحرف المقدر الذي هم اللام أو من وهسن هذفه لنيابة المضاف إليه عنه و ميرورته عوضا عنه في اللفظ وليس أيس في المعمل .

Ibn al-Sarraj follows the view of Sibawayh in holding that the first element in an <u>idafah</u> construction puts the second into the genitive, and he considers that this construction simply conveys the meaning of either of the particles <u>li</u> or <u>min</u>. <sup>26</sup> The view taken by Ibn Ya'ish is,

however, rather more subtle involving the use of <u>taqdir</u> so that the <u>idafah</u> construction is seen not merely as conveying the meaning of either <u>li</u> or <u>min</u> but the term in the genitive itself owes its grammatical case to these notional particles. Again, this is a case of later grammarians putting forward more complex grammatical explanations.

The second major class of regent specified by Ibn al-Sarrāj in the <u>Usul</u> consists of the verb and his treatment of the verb as a regent deserves consideration. In explanation of the verb's power of government Ibn al-Sarrāj writes: 27

اعلم أن كل فعل لا يخلو من أن يكون عاملا وأول عمله أن يرفع الفاعل أو المفعول الذي هو هديت عنه نحو: قام زيد و ضرب عمرو. وكل اسم نذكره ليزيد في الفائدة بعد أن يستفني الفعل مدينا بالدسم المرفوع الذي يكون ذللا الفعل مدينا عنه، فهو منصوب و نصبه لأن الكلام قدم قبل عبيه و فيه دليل عليه.

What is of note about this explanation of the verb's power of government is that Ibn al-Sarrāj explains the principle which determines the use of the accusative case, which is a grammatical case particularly to be explained by the governing power of the verb. When Ibn al-Sarrāj details the various uses of the accusative at a later point in the <u>Uşūl</u> he again refers to what determines the use of this case:

الأسهاء المنصوبات تنتقم قسمة أولى على فترمين:
فالضرب الأول وهوالعام الكثير كل اسم تذكره بمد
أن يستفنى الرافع بالمرفوع وما تبعه فى رفعه إن
كان له تابع ، وفى الكلام دليل عليه ، فهد نصب
والضرب الدخر كل اسم تذكره لفائدة بعد اسم

معناف أو فيه نون ظاهرة أو مضمرة وقد تج بالاضافة والنون , وهالت النون والإضافة بينها ولولاها لصلح أن يضاف إليه . فهو نصب .

The second class of accusative usage listed in this passage is in fact the accusative of the <u>tamyiz</u> used in counting and measuring and it is treated by Ibn al-Sarraj as different because it is not an accusative usage dependent on the verb.

of course, omitting terms in the accusative in this way is purely an analytical device of the grammarians and they can be considered only as "formally" redundant because they are not redundant in any real sense. The principle of the formal redundancy of accusative terms is based on the fact that a verb and its subject are sufficient in themselves to form meaningful utterances, and Ibn al-Sarrāj refers to this by using the expression will be accusative can possibly be considered as

formally redundant because in certain constructions the accusative is used to express what is the subject or predicate of the sentence. This point will be given further consideration in discussing the approach of later grammarians to formal redundancy.

When dealing with the various uses of the accusative individually Ibn al-Sarraj specifically refers to the concept of formal redundancy when dealing with the <u>bal</u> and the use of the accusative in exceptive sentences. The passage relating to the <u>bal</u> is cited later in the chapter when dealing with another point, but it is worth giving some consideration at this point to Ibn al-Sarraj's treatment of the use of the accusative in exceptive sentences because certain aspects are of interest when considering the idea of formal redundancy. In introducing his discussion of exceptive sentences Ibn al-Sarraj writes: <sup>30</sup>

المستننى ينب المنحول إذا أتى به بعد استفناه الفعل بالفاعل و بعد تهام الكلام. تقول: هاه في القوم إذّ زيدا , فجاه في القوم كلام تام وهو فعل وفاعل.

In this passage Ibn al-Sarrāj likens the term in the accusative after <u>illā</u> to the <u>maf'ul</u> and this implies that it is dependent for its grammatical case on the regimen of the verb. In commenting on the expression is the makes this specific: 31

فله جاز أن تذكر زيدا بعد هذا الكلام بفير هرف الاستثناء ما كان إلا نصبا ، ولكن لاحمن لذلا إلّا بنوسط في آهر فلما توسطت إلا هدت معنى الاستثناء ووصل الفعل إلى ما بعد إلّا.

Ibn al-Sarraj here draws attention to the principle of the formal redundancy of the accusative by making the point, which in terms of this principle is valid but in practice is meaningless, that were it possible

to introduce a term like in after policy without also introducing the exceptive particle, then the term would have to go into the accusative. This particular line of argument seems to be peculiar to Ibn al-Sarraj.

The later grammarians followed Ibn al-Sarraj in holding that the term in the accusative after illa owed its case to the regimen of the verb acting through the medium of the exceptive particle. However, this view only seems to have developed in the course of time. In the Kitab Sibawayh deals with the question of the regent by quoting the view of al-Khalil that the regent governing the accusative is the discourse (kalam) preceding the particle illa and there is no mention that it is the verb itself in the preceding discourse which is the regent. This view is in accord with Sibawayh's view that the regent acting on the maf'ul ma'ahu is the preceding discourse, whereas the later Basrans came to the view that it was the verb itself in the preceding discourse that was the regent. 33

However, there were other views on the question of the regent governing the accusative after illa. Al-Mubarrad is recorded as having held the view, which is associated with the Kufan school, that the particle illa itself is the regent on the grounds that it replaces an expression Mugtadab: 34

... لها قلت : جاه في القرم وقع عند السام أن زيدا فيهم ، فلما قلت : إلا زيدا - كانت (إلا) مدلامن قدلك: أكنى زيدا، وأستننى فيمن عاوني زما , فكانت بدلا من الفعل .

Although he holds this view about the regent al-Mubarrad accepts that the accusative after <u>illa</u> is to be explained by formal redundancy: 35 فإيا باب الاستناء - إذا استنى الفعل بالفائل، أو الابتداء يخبره - النصب، إلا أن يصلح الدل، فيكون أجود، والنصب على هاله في الجواز.

The al-Sarraj does refer to the idea that illa can be related to the expression but he simply uses this as an analogy and does not use it as a formal explanation of the use of the accusative: 36

فالمستنى بعن المستنى منه، ألا ترى أن زيدا من القوم من القوم فهو بعملهم فتقول على ذلا: فعرب القوم إلا زيدا ، فكأنك قلت في جميم ذلا أستنى زيدا

One particular variety of exceptive sentence in which the particle <u>illa</u> occurs twice followed by two different terms enables Ibn al-Sarrāj to confirm the general principle of formal redundancy: <sup>37</sup>

فإن أوقعت استناه بعد استناه قلت ما قام أهد إلا زيد الاعمرا , فتنصب عمرا لانه لا يجوز أن يكون لفعل واهد فاعلان مختلفان يرتفعان به بخير هرف عطف . فهذا كا يبصرك أن النصب واهب بعدا ستفناه الرافع بالهرفوع . وللا أن تقول : ما أتاني أهد إلا زيد إلا عمرا , وإلا زيدا إلا عمرا , وألا زيدا إلا عمرو , فتنصب أيهما نئت وترفع الآهز.

This represents a neat way of proving the principle of formal redundancy and it is perhaps indicative of the attention which Tbn al-Sarraj gives to this question that other grammarians do not seem to have put this construction to the same use. Al-Mubarrad, for instance, simply takes the view that after the double <u>illa</u> one of the terms is in the nominative due to the regimen of the verb and the other is in the accusative in

accordance with the principle accepted by the grammarians that the term after <u>illa</u> goes into the accusative unless there is a specific reason why this should not happen: <sup>38</sup>

تقول: ما جابن أهد إلا زيد إلا محرا. وإن نشت قلت: إلا زيدا إلا محرو. فالمعنى فيهما جميعا واهد، وإن اختلف الإعراب؛ لأنك إذا شفلت الفعل بأهدها انتصب الآخر بالاستثناء ولم يصلح البدل؛ لأن المرفوع منهما موجب.

Ibn al-Sarrāj gives particular attention to the idea that the accusative is to be seen as formally redundant but, although this idea was accepted by later grammarians, it is not set out with the same degree of prominence. However, later scholars did give further consideration to this question and coined the term 'umdah to serve as an antonym to the term fadlah which, as has been mentioned, is found in the Usūl. Al-Astarābādhī using these two terms writes: 39

.. والمرفوع عمدة الكلام كالفاطل والمستدا والخبر والخبر والبراقي محمولة عليها والمنصوب في الأمل فضلة لكن يشبه بها بعن العمد كاسم إن وخبركان وخبركان وخبر ما ولا.

The term 'umdah covers those uses of the nominative and the accusative such as the mubtada', khabar, fa'il, subject of inna, and the predicate of kana, which are the basic elements in creating meaningful sentences. The term fadlah covers those uses of the accusative which can be seen as formally redundant and is even used by later authors like al-Astarabadhi and Abū Ḥayyan to cover uses of the genitive.

In explaining the meaning of the term 'umdah al-Astarabadhi mentions that it covers the subject of inna and the predicate of kana because these uses of the accusative have an affinity with uses of the nominative

and cannot be likened to those uses of the accusative which are to be explained by formal redundancy. In defining the term 'umdah in this way al-Astarābādhī introduces a consideration which is not taken into account by Ibn al-Sarrāj because the latter does not make the point clear that not all uses of the accusative can reasonably be explained by formal redundancy.

It has been mentioned that later writers do not seem to give the question of the semantic redundancy of the accusative quite the degree of prominence which it has in the <u>Usul</u>. Al-Astarabadhi, in fact, introduces the idea when discussing why in arranging a grammar the uses of the nominative case are treated before those of the other grammatical cases, and the same applies to Ibn Ya'ish who writes:

اعلم أنه قدم الكلام في الإعراب على المرفوعات لأنها اللوازم للجملة والعمدة فيها التي لا تخلو صنها، وما عداها فضلة يستقل الكلام دونها.

The Ya'ish does subsequently go into the question of formal redundancy in more detail but this is in fact connected with his discussion of why the <u>fā'il</u> is in the nominative, and in the course of this he explains why the <u>strong</u> vowel <u>dammah</u> is used for the nominative and the <u>weak</u> vowel <u>fathah</u> is used for the accusative:

الفاعل إنها اختص بالرفع لقوته والمفعول بالنصب لضعفه والمصفى بقوة الفاعل تحكنه بلزوم الفعل وعدم استفناء الفعل عنه وليس المفعول كذلك بل يجوز سقوطه وهذفه ألا ترى أنلا تقول : فعرب زيد ويكون المكلم مستقلا وإن لم تذكر سفعولا . ولو أهذت تحذف الفاعل ولم تقم مقامه شيئا نحو: فعرب زيدا، من غير فاعل لم يكن كلم ... فنا سبوا بأن أعطوا

## الأقوى الأقرى والأمنوف الأضمف.

It is possible to relate the view of later grammarians like Ibn al-Sarrāj regarding the function of the accusative case to an idea suggested in the <u>Kitāb</u> of Sībawayh. In the <u>Kitāb</u> Sībawayh relates many uses of the accusative case to its use with the numerals 11-99 and he refers to this by the stock phrase '<u>ishrun dirham</u>. The ideas underlying Sībawayh's frequent reference to this expression are outside the scope of this present study, <sup>43</sup> but one particular use of the accusative which he likens to that with '<u>ishrun dirham</u> is treated in a way which can be related to the ideas of the later grammarians about formal redundancy. However, Sībawayh does not give any definite expression to the idea that the use of the accusative case is in general to be explained by formal redundancy. The particular usage in question is as follows:

وأما قولهم: دارى هلف بارك فرسخًا، فانتصب لأن هلف هبر للدار وهو كلام قد عمل بعضه في بعض واستفنى. فلما قال: دارى هلف دارك، أبهم ولم يدر ما قدر ذلك فقال: فرسخا وذراعا وصلا. أراد أن يبين. فيعمل هذا الكلام في هذه الفايات كما يسين. فيعمل هذا الكلام في هذه الفايات كما عمل له عشرون درها في الدرهم ....

The gist of Sibawayh's argument here is that the example رارئ is a semantically complete utterance consisting of <u>mubtada</u>' and <u>khabar</u>, and because of this the additional element which amplifies the meaning is put into the accusative. Here Sibawayh uses the verb which The al-Sarraj also uses in presenting the principle of formal redundancy. Although this particular instance can be used to show how the ideas of later grammarians on the use of the accusative

can be traced back to Sibawayh, nevertheless, there does not appear to be another instance in the <u>Kitab</u> of a similar correspondence. It would appear that the idea of formal redundancy with regard to the accusative case only properly developed after the time of Sibawayh. Al-Mubarrad does make passing reference in the <u>Mugtadab</u> to this idea but unlike Ibn al-Sarraj he lays no great emphasis on it. 45

If the principle of formal redundancy explains uses of the accusative case there is another, but subordinate, consideration. This is, that although the part of a sentence which consists of a verb and its subject is formally complete, nevertheless, it presupposes certain terms which are all put into the accusative. In describing the circumstances in which the accusative is used Ibn al-Sarrāj writes:

وكل اسم تذكره ليزيد في الفائدة بعد أن يستنى النعل مدينا النعل بالاسم المرفوع الذي يكون ذلا النعل هدينا عنه، فهر منصب به ونصبه لأن الكلام قدم قبل عنه، فهر منصب به ونصبه لأن الكلام قدم قبل كينة ونبه دليل عليه . (underlining mine)

Although Tbn al-Sarraj particularly uses here the idea of formal redundancy to account for the use of the accusative case, he also makes the point that a formally complete utterance contains a "suggestion" (dalil) of terms in the accusative.

such a "suggestion". 47

Ibn al-Sarrāj again introduces the notion of a "suggestion" when he discusses the <u>pāl</u>: 48

فأما الذي يسمونه الحال فنحر قرلك: بدا عبد
الله راكبا، وقام أهوك مستصبا، وجلس بكر
متكلنا . فعبد الله عرتفع بجاء وراكبا منتصب
لشبهه بالمفعول لأنه جي به بعد تمام الكلام
واستفناء الفعل بفاعله وإن في الفعل دليلا عليه كها
كان فيه دليل على المفعول . ألا ترى أنلا إذا قلت :
قمت ، فلا بد من أن تكون قد قمت على هال
من أهوال الفعل.

The argument that a verb involves the "suggestion" of a <u>hāl</u> is logically sound but is clearly more abstract than saying that a transitive verb involves the "suggestion" of a direct object. The idea that the verb involves the "suggestion" of terms in the accusative can be found in other authors, and al-Suyūṭī quotes the <u>Sharh al-mufaṣṣal</u> of al-Sakhāwi (1163-1245) on the resemblance of the <u>hāl</u> to the <u>maf'ūl bihi</u> and there the same point is made as is done in the passage above from the <u>Uṣūl</u>: 49

أما شبهها بالمنعول به فلأن في الفعل دلالة على كل واحد منهما فإذا قلت : ضربت ، دل ذلك على مضروب و على هال ، ولأن كل واحد من الحال والمنعول استقلال الفعل بالفاعل.

When dealing with the <u>maf'ul ma'ahu</u>, after having dealt with the <u>maf'ul lahu</u>, Ibn al-Sarraj again draws on the idea of the verb involving a "suggestion" of terms in the accusative: <sup>50</sup>

و بفرق سين هذا الباب والباب الذي قبله أن باب المفعول له إذا قلت: هئتلا طلب الخير، إن في هئتلا طلب الخير، إن في هئتلا دليلا على أن ذلا لشيء . وإذا قلت: ط منعت وأبالا ، فليس في منعت دليل على أن ذلا مع شيء لأن لكل فاعل غرفناله فعل ذلا الفعل ، وليس لكل فاعل معاهب لا بد منه .

Ton Ya'ish introduces this same consideration when he explains why the preposition <u>li</u> which is used with the <u>maf'ul lahu</u> may be omitted but not the conjunction <u>wa</u> which is used with the <u>maf'ul ma'ahu</u>: <sup>51</sup>

وذلا دان دلالة الفعل على مفعول له أقوى من دلالته على المفعول معه. وذللا لائنه لا بد لكل فعل من مفعول له سواء ذكرته أولم تذكره إذ العاقل لا يفعل فعلا إلا لفرض وعلة ، وليس كل من فعل فنيئا يلزمه أن يكون له ضريلا أو معاهب.

In this passage it is worth noting Ibn Ya'ish's remark:

Although this specifically refers to the maf'ul lahu it confirms that in general the capacity of a verb to presuppose certain accusative usages is not affected by their being expressly mentioned cr not.

It is clear that a verb can only involve the "suggestion" of certain types of accusative expression and the grammarians particularly note that the verb does not involve the "suggestion" of a maf'ul ma'ahu. Only a transitive verb can have a "suggestion" of a direct object.

Two other uses of the accusative which the verb presupposes, and which the grammars specifically mention, are the hal and the maf'ul lahu but

other uses could clearly be added. In works of grammar later than the Ugul the idea that the verb presupposes certain uses of the accusative is not given any particular prominence and only appears incidently, whereas in the Ugul Ibn al-Sarrāj refers to it in explaining the verb's power of government and in enumerating the uses of the accusative as well as introducing it as appropriate at other points. Something similar to the later more developed idea of the verb involving a "suggestion" of certain accusative usages can be found in the Kitāb of Sībawayh when the point whether a sentence like is is like one with an accusative of place is discussed: 52

وقد قال بعضهم: ذهبت الشامّ، شبهه بالمبهم المبهم المبهم الذكان كان كانا يقع عليه المكان والمذهب. وهذا شاذ لانه ليس في ذهب دليل على الشأم وفيه دليل على الشأم وفيه دليل على المذهب والمكان.

## Notes to Chapter III

- 1 E.I.<sup>2</sup>, s.v. 'amil (G. Weil).
- 2 Ibn al-Sarraj, vol. i, pp. 55-60.
- 3 See Hajji Khalifah, vol. ii, p. 1179. For al-Jurjani see Kahhalah, vol. v, p. 310. The Arabic text of the 'Awamil al-mi'ah is contained in an appendix to A. Lockett, Mi'ut Amil and Shurhoo Mi'ut Amil, Calcutta, 1814.
- 4 See Ḥājjī Khalīfah, vol. ii, p. 1179.
- 5 Al-Suyuti, Ashbah, vol. i, pp. 241-56.
- 6 This table is based on the material in Ibn al-Sarraj, vol. i, pp. 55-60.
- 7 This table is based on the text of the 'Awamil al-mi'ah contained in the work by Lockett cited in n. 3 above.
- 8 See n. 5 above.
- 2 E.I.<sup>2</sup>, s.v. 'amil (G. Weil).
- 10 See Ibn Ya'ish, vol. i, pp. 83-4: vol. vii, pp. 12-3.
- 11 Al-Zajjaji, Jumal, p. 48.
- 12 I. M. Rejab, "A critical edition of Sharn al-Jumal li al-Zajjājī by Ibn Babshād" (Ph.D dissertation, University of St. Andrews, Dec. 1975), p. 182. For Ibn Babshādh see Kannalah, vol. v, p. 32.
- 13 Abu Hayyan, p. 38.
- 14 Al-Mubarrad, vol. iv, p. 126.
- 15 Ibn Ya'ish, vol. vii, p. 12.
- 16 Ibn al-Sarraj, vol. ii, p. 151.
- 17 Al-Mubarrad, vol. ii, p. 5.
- 18 Ibn al-Sarraj, vol. i, pp. 55-8.
- 19 Ibid., p. 55.
- 20 Ibid., pp. 62-3; see also pp. 65 and 279.
- 21 Ibn al-Anbari, Insaf, p. 23; Ibn Ya'ish, vol. i, p. 85.
- 22 Ibn Ya'ish, vol. i, p. 85.
- 23 Ibn al-Sarrāj, vol. i, pp. 56-7.
- 24 Ibid., p. 57. In 1. 4 of this passage ; has been substituted for in the printed text.

- 25 Ibn Ya'ish, vol. ii, pp. 117.
- 26 Sibawayh, vol. i, p. 209; see also E.I.<sup>2</sup>, s.v. <u>idafah</u>.
- 27 Ibn al-Sarraj, vol. i, p. 58.
- 28 Ibid., p. 189.
- 29 Ibid., pp. 83 and 102; vol. ii, pp. 124, 251, and 252.
- 30 Ibid., vol. i, p. 342.
- 31 Ibid., pp. 342-3.
- 32 Sibawayh, vol. i, pp. 360 and 369.
- 33 Ibid., p. 150.
- 34 Al-Mubarrad, vol. iv, p. 390.
- 35 Ibid., p. 396.
- 36 Ibn al-Sarraj, vol. i, p. 343.
- 37 Ibid., p. 345.
- Al-Mubarrad, vol. iv, p. 424. See also Ibn Ya'ish, vol. ii, pp. 92-3.
- 39 Al-Astarabadhi, vol. i, p. 70.
- 40 Ibid.; Abu Hayyan, p. 130.
- 41 Ibn Ya'ish, vol. i, p. 75.
- 42 Ibid.
- See M. G. Carter, "'Twenty dirhams' in the <u>Kitab</u> of Sibawayhi", <u>Bulletin of the School of Oriental and African Studies</u> 35 (1972), pp. 485-97.
- 44 Sibawayh, vol. i, p. 207.
- 45 Al-Mubarrad, vol. iii, p. 116; vol. iv, p. 396.
- 46 Ibn al-Sarraj, vol. i, p. 58; see also p. 189.
- 47 Ibid., pp. 81-2. Passage quoted p. 36 above.
- 48 Ibid., p. 258.
- 49 Al-Suyūti, Ashbah, vol. ii, p. 190.
- 50 Ton al-Sarraj, vol. i, p. 257.
- 51 Ibn Ya ish, vol. ii, p. 53.
- 52 Sibawayh, vol. i, p. 16.

#### CHAPTER IV

## THE USE OF QIYAS AND TAQDIR

One of the most important terms used by the Arab grammarians in their study of grammar is the term <u>qiyas</u>. As a simple definition of <u>qiyas</u>, which would translate literally into English as "analogy", it may be said that a language has established rules of phonology, morphology, and syntax, and it is the principle of <u>qiyas</u> at work which ensures that in general these rules are followed. It is this idea to which Ibn al-Anbari refers when he writes: 1

اعلم أن إنكار القياس في النو لا يتحقق لأن النحو كله قياس. ولهذا قيل في هده: النو علم بالمقاييس المستنبطة من استقراء كلام الوب فهن أنكر القياس فقد أنكر النحو.

A very simple example from morphology of the use of the term <u>qiyas</u> is found when Ibn al-Sarrāj defines the varieties of grammatical anomaly which occur: <sup>2</sup>

منه ما شذّ من جابه و قباسه ولم بشذ في استمال الوب له نحم استماذ فإن جابه و قباسه أن يحل فيقال استماذ سنل استقام واستعاذ وجميع ما كان على هذا الحيال. ولكنه جاء على الأصل واستمالته العرب كذلك.

The grammarians understanding of <u>qiyas</u> is in fact often best illustrated in their discussions of situations where they consider that

the demands of givas are disregarded in actual usage. Ibn al-Sarraj, for instance, notes that there are certain aspects of the use of conjunctions to which the normal rules of usage do not apply: 3

واعلم أنه قدما في العملن أشياء تخالفة للقياس. في ذلا قولا: حرث برجل قام ابواه لا قاعدين. فقدلا : قاعدين , معطوف على قام وليس في قوللا: قاعدين، نني و يرجع إلى رجل كما كان في قولا: قَامُ أبراه منسير يرجع إلى رجل جماز هذا في المعطوف على غم قياس. هذا لفظ المازني و قول كل من يرمني قدل وكان ينبني أن تقول : حرت برهل قائم ألواه ولا قائد ألواه , وأن لا يحيع الأبوان مفري. ولكنه حكى عن العرب وكنر في كلامهم حتى مار قماسا مستقيما . ويما جاه في المطنى لا يجوز في الأول قدل العرب: كل شاة وسخليه بدرهم ، ولو ممات السخلة تلى كل لم يستم. و سئله: رب رمل وأخده ، فلوكان الأغ على رب لم يجز .ومن كلام العرب: هذا الصاربُ الرهل وزيد، ولوكان زر یکی الفارب لم یکی جرا.

anomalous. However, there is this difference that the form of this verb cannot be applied to other verbs at will, whereas this construction can be used where neccessary. Although this construction may oppose what <u>givas</u> dictates, it can itself be described as <u>givas</u> because it is a construction to be followed or, as the grammarians would say, ..., ...

The grammarians would observe and explain <u>divās</u> at work in the Arabic language but, particularly in the field of syntax, they could also actively put <u>divās</u> to work to determine what was or was not correct in respect of usages and constructions which, although unattested in use, were felt to be consistent with attested usage. By the operation of <u>divās</u> the basic rules of usage derived from the simplest constructions would consistently govern how the language was used in more complex constructions, and this assisted the grammarians in determining what was or was not correct in respect of unattested usages and constructions. This active use of the principle of <u>divās</u> is of particular interest and will be discussed in some detail.

Already by the late 3rd century A.H. the work of the grammarians was to a large extent to explain, rework, and develop the material on the Arabic language contained in the writings of the earliest authorities of whom the most important by a very long way was Sibawayh. No further basic research seems to have been done by way of consulting reputable informants, and the only activity that approximated to basic research was the continued study of early written sources such as the Qur'an and the material recorded in collections of poetry and proverbs, which was undertaken with a view to extracting further information on usage.

The al-Sarrāj himself makes clear that by his time the study of grammar was essentially derivative:

# استقراء كلام العرب حتى وتفوا سنه على الغرض الذي قصده المبتدئون بهذه اللغة

Even if grammatical studies were very solidly based on the work of the early scholars, there was scope for the later grammarians to apply qivas in dealing with usages and constructions which were not covered in the works of the early scholars and on which no information could be found by studying the diction of the Qur'an and early poetry.

On this point Ibn al-Sarrāj makes a remark of general significance when discussing one particular usage: فأذا لم يمنح كام التي التياس في التيا

One of the questions discussed by Ibn al-Sarrāj in which the application of givas is important is that of word order with the verb kana. Ibn al-Sarrāj considers that word order with kana can be related to that which is permissible with the mubtada' and khabar construction:

واعلم أن جميع ما هاز في المستدأ و هنبره من التقديم والتأخير فهو هائز في كان إلّا أن تفصل بينها وبين ما عملت فيه بما لم تعمل فيه . فإن فصلت بفرف ملفى هاز . فأما ما يجمز فقولا : كان منطلقا عبد الله ، وكان أخالا الله ، وكان أخالا ما يما منطلقا اليوم عبد الله ، وكان أخالا ما علامهما قائما . وكذلا أخوات غلامهما قائما . وكذلا أخوات غلامهما قائما . وكذلا أخوات كان قال الله تعالى : وكان خقا عَلَيْنَا نَصْرُ أَنْ فَقَا عَلَيْنَا نَصْرُ الْمُؤْسِنِينَ .

The possible permutations of word order with simple sentences

are quite easy to grasp but the question of possible word order with kāna, when it has more complex predicates, is rather more involved and Ibn al-Sarrāj gives it considerable attention. It is worth noting at this point that the Arab grammarians often deal with questions of word order under the heading of which is an expression best translated into English as "inversion". The Arabic expression itself envisages terms being put either before or after their normal position in a sentence. According to Ibn al-Sarrāj inversion can be carried out when kāna has complex predicates in exactly the same way as it is with simple predicates, as long as the integrity of the predicate is maintained:

والتقديم والتأخير في الأهبار المجتملة كنزلته في الأهبار المخردة ما لم تفرقه تقول: أبره صنطلق كان زيد ابره صنطلق وقائمة عارية يكبها كان زيد ابره صنطلق وقائمة عارية يكبها كان زيد ، ثريد: كان زيد قائمة عارية يكبها و في داره ضربه صحرو خالما كان زيد . فإن قلت : كان في داره ابره ، لم يجز لأن الظرف تريد : كان زيد في داره ابره ، لم يجز لأن الظرف للأب فلبس من كان في شيء وقد مقلت به يستها وبين هبره . ولو قلت : كان في داره ابره ، لم يكن في داره ابره ، ولم قلت . كان في داره وعلى عملته فيمار مثل قدلك : كان منطلقا زيد .

In addition to the sort of change of word order envisaged above, Ihn al-Sarraj further states that the subject of  $\underline{kana}$  can be made to precede the verb and at the same time this can also be done to the predicate:

و تقول: زيد كان منطلقا أبده ، فزيد مبتدأ

وما بعده هنبر له وفي كان منمير زيد وهو اسمها وسنطلقا أبوه هنبره وإن شئت رفعت أبا كان وجولت منطلقا أبوه وتقول: زيد منطلقا أبوه كان , تريد : زيد كان منطلقا أبوه , منثل المسألة التي قبل .

Although he does not mention the point Ibn al-Sarrāj is now discussing changes of structure with reference to a model sentence of the type المون المو

where its predicate is complex is clearly based on the application of givas because many of the alternatives are tortuous and could hardly be supported from attested usage. Even for some of the most simple changes Ibn al-Sarraj has to rely on givas as is the case with فان زيد which he treats as analogous to عنوات با مناسلتا كان زيد a completely acceptable construction. It is while discussing this point that Ibn al-Sarraj makes the general remark about the scope of givas quoted earlier:

وقال قوم: أبوه قام كان زيد، خطأ لأن ما لا يعمل فيه كان لا يتقدم قبل كان. والتياس ما هنبرتك به إذ كان قولك : أبوه قائم في موضع قولك : منطلق ، فهر يمنزلته . فإذا لم يصح سماع الشيء عن الحرب لجئ فيه إلى القياس.

The type of change in normal word order with kana which is discussed here appears to have been a particular concern of Ton al-Sarraj because later grammarians like Ton Ya'ish are more interested in examining kana together with analogous verbs to establish which simple changes of word order are permissible with each verb. 12 However, the only other verb apart from kana which Ibn al-Sarraj discusses in this respect is laysa.

Another example of the application of <u>giyas</u> by Ibn al-Sarraj is found in his discussion of the use with the verb <u>zanna</u> and analogous verbs of the pronoun which the Basrans generally called the <u>damir</u> al-sha'n or <u>damir</u> al-qissah:

و تقول: ظننت زيد قائم، تربد الأمر والخبر. وهذا الندى يسميه الكوفيون المجهول. وتقول: ظننت هند قائمة, فتذكر لأنك تربد الأمر والخبر، وظننته تقوم هند. و يجوز في النياس: ظننتها زيد قائم، تربد القصة , ولا أعلمه مسموعا من العرب.

It is of interest that Ibn al-Sarraj refers to the Kufan technical term majhūl as if there was no corresponding Basran term available to him. <sup>15</sup> The use here of the word <u>qisşah</u> does not seem to have any connection with the technical term <u>damir al-qissah</u> but is used simply as a feminine noun in contrast to the masculine nouns <u>amr</u> and <u>khabar</u>.

Ibn al-Sarrāj's sanctioning of the use of the feminine pronoun in the example والمالية is governed solely by the application of giyas since he concedes that it is an unattested usage. Of course,

in certain constructions a feminine pronoun is normally found as is the case with inna/anna when there is a term following of feminine gender and this is illustrated by the Qur'anic phrase فَإِنْهَا لاَ نَعْمَالُ مُنْ الْمُعَالُ مُنْ الْمُعَالِينَ الْمُعَالُ مُنْ الْمُعَالُ مُنْ الْمُعَالُ مُنْ الْمُعَالِينَ الْمُعَالِينَا اللّهُ الْمُعَالِينَ الْمُعَالِينَ الْمُعَالِينَ الْمُعَالِينَ الْمُعَالِينَ الْمُعَالِينَ الْمُعَالِينَ الْمُعَالِينَ اللّهُ اللّهُ

In tracing the basis of Ibn al-Sarrāj's reasoning it is possible to treat a sentence like in which the nominative in which the pronoun as similar to one like in which the accusative follows the pronoun, because in the latter case there is a view that the feminine pronoun is possible although in this particular instance a slightly different line of argument is used: 17

تقول: طننته أخال قامًا، تريد: طننت الظن فتكون الها كناية عن الظن كأنك قلت: فلننت أخالا قامًا الظن، ثم كنيت عن الظن. وأجاز بعضهم: طننتها أخالا قامًا، يريد الظنة.

However, this passage occurs before Ibn al-Sarrāj turns to the construction and a usage which the Kufans permit, and which Ibn al-Sarrāj mentions directly after proposing this construction, has a more close connection with the application of giyās here: 18

فأمّا الكوفيون فيجيزور تأنيت المجهول وتذكيره إذا وقع بعده المؤنث. يتولون: ظننته هند قائمة.

It is clear why the Kufans permit the use of both the masculine and the feminine pronoun here because the masculine can be seen as representing the basic usage in the construction while the feminine occurs as a result of a very natural attraction of gender. However, Ibn al-Sarrāj's sanctioning of the construction also is somewhat different from this because he holds that the feminine pronounce.

owes its gender to the purely notional term gissah.

While dealing with the same matter in the Sharh al-kafiyah al-Astarabadhi also writes that putting the pronoun into the feminine, even if the following clause does not relate to a term of feminine gender, is supported by givas and is based on the gender of the notional term gissah, although such a usage is unattested. <sup>19</sup> Furthermore, al-Astarabadhi expresses the view that if the clause after the pronoun does in fact relate to something of feminine gender, the pronoun itself is feminine out of regard for the notional term gissah, although in these circumstances an attraction of gender between the pronoun and the feminine term in the following clause is apposite:

The grammarians also took the view which corresponds to this that even where the masculine pronoun relates to a masculine term in the following clause, it still takes its gender from a notional term like khabar.

This last paragraph makes clear the view of the later grammarians on the gender of the damir al-sha'n and, although Ibn al-Sarrāj does not put forward this view in such a precise and explicit manner, it is clear that he held that the gender of the damir al-sha'n is determined by something notional and not with reference to the following clause. Ibn al-Sarrāj does give considerable information on the earlier stages of the discussion of the question of the gender of the damir al-sha'n, and it may well be that he was the first to propose that on the basis of givas the construction

An example of the application of <u>qiyas</u> in a very marked way is provided by relative clause predication. In their grammars both al-Mubarrad and Ibn al-Sarraj give considerable attention to recasting

Relative clause predication does not stand out as a subject which requires a great deal of attention but it was one to which early grammarians in particular applied themselves. Ibn Ya'ish gives some attention to this topic but he does not examine it in the depth which The later grammarian, al-Astarabadhi, does, however, go into some detail on relative clause predication. 24 seems to be the case that later grammars give some attention to this subject but, essentially, it had really been of interest to earlier grammarians, although there remained some residual interest in it. It is evident that an important aspect of the original interest in relative clause predication was the training of students of grammar and, although Ibn al-Sarraj does not expressly say this, he does repeatedly use the phrase بان فيل لا: أخرى which would seem to indicate that it was an exercise which someone might require another to do. 25 Relative clause predication was not a topic with which Sibawayh was concerned but it was rather a subject to which later scholars turned. In the Usul al-Mazini is frequently cited on this topic as if he were someone very much connected with its study.

One of the simplest forms of verbal sentence is one of the type in its and by the process of relative clause predication this can be converted to which is and its and its and its as well

as sentences like this in the singular, sentences in the dual and plural were also considered and the neccessary changes made which this operation dictates. The grammatical analysis which Ibn al-Sarraj gives for the changes which are made to the simple sentence can be taken as similar to that given for equivalent changes made to more complicated sentences:

إذا قبل للا: أخبر من زم طالني قلت : الذي ذهب زر فالذي سندا و دهب صلته وفيه ضحير الفاعل وهو يرجع إلى الذي فقدتم الذي بصلته وهبره زيد فإن قبل للا: أهبر عنه بالألف والام قلت: الذاهب أفنوك فرفعت الذاهب لأنه اس ومعناه : الذي ذهب . ولم يكن بدمن رفعه لأن الألف واللام لا تنتميل من الملة كا نفصال الذي وهي جزء من الاسم ولكن المعنى معنى الذي فإن نست الذي قلت :اللذان قاما أخواك. فإن مملت موضم الذى الألف واللام قلت: القَامَّان أَخْوالا، تُنين الثَّامُّ إِذَ لَم يكن سِيل إلى تنفية الدُّنف والام، والتأويل: اللذان قاما. و يرجع إلى الألف واللم الضمير الذي في القائمين. ... فإن جممت قلت: الذين قاموا إهونك, وبالأنف واللام: القَاعُون إِفُونَكِ وَتَفْسِرِ الْجُمْ كَتْسِيرِ النِّيَّةِ.

Although relative clause predication is in essence a theoretical exercise it does have a practical aspect because there is certainly a difference in emphasis between مالزى ذهب أيد and مالزى ذهب أيد and الزى ذهب أيد and الزى ذهب أيد and الزى ذهب أيد المعادية على المعادية ا

attention to this point and explains how the sentence هنریت زید and الذی ضریته زید differ in emphasis: 28

... إذا قلت: فعربت زيدا , فربها تجاطب به من لا يعرف أن للا مصروبا في الدنيا وربها تخاطب من يعرف شخصا بمضروبيتلا لكنه لا يعرف أنه زيد , وأما قوللا : الذي فعربته زيد , فلا تخاطب به إلّا على الوجه الثاني أي تخاطب من يعرف أن لا مضروبا لأن مضمون الصلة يجب أن يكون معلوما للمخاطب كها ذكرنا ولكن لا يعرف أنه أنه زيد , إذ لو عرف ذللا لوقع الإضار عنه بأنه زيد هنائها . فالجملة الثانية نصت في المحتمل الثاني للجملة الثانية نصت في المحتمل الثاني للجملة الداولي .

Various problems of usage arise in the course of discussion of relative clause predication and these have to be dealt with by the use of grammatical reasoning. One such question which Ibn al-Sarrāj discusses is whether it is admissible to say still on the analogy of the sale of the

ولا يجوز أن تقول: القائم إضوتك، على قول من قال أقائم إضوتك، يجرى بجرى؛ أقائم إضوتك، يجرى بجرى؛ أيقائم إضوتك، يجرى بجرى أيقائم الألف واللام لا يجرى مذا المجرى لانه قد تكمل السا معرفةً والمعارف لا تقوم مقام الأفعال لأن الأفعال نكرات ولكن لا يجوز أن تعمل ما في ميلة الألف واللام وهو قائم.

Because a subject like relative clause predication involves to a large extent discussion of constructions which are grammatically possible but which are often rather barbarous, there is considerable scope for the use of grammatical reasoning to prescribe what is permissible since attested usage is of little help. The reasoning employed can be somewhat involved as is the case with Ibn al-Sarrāj's discussion of the validity of the construction القائان أبوالياً . 30

فتقول: القائم أبره أخول، والقائم أبوها أخواك.
ولا يحوز أن تقول: القائمان أبراها أخوالا , من أجل أن قائم قد عمل عمل الفعل وما تمت الألف واللام بعد بصلته , وما لم يتم فلا يجوز أن يشنى فإذا أعملت ما في صلة الألف ولللام في فاعل امتنمت التشنية وإنها جاز أن تقول : القائمان أخوالا , لأن الاسم قد تم و الضمير الذي في القائم لا يظهر فأشبه ما لا منمير فيه . وإنها احتمل الضمير الاسم إذا كان في صلة ما هم له وجاريا عليه استضاء بعلم السام. وليس جاب الأسماء أن تضمر فيها إنما ذلك للأفعال , فإذا لم يكن اسم الفاعل فعلا في المقيمة للألف واللام أو لها يوصف به أو يكون خبرا له , لم يحتمل الضمير البتة .

When relative clause predication is applied to verbs which govern more than one direct object even the most basic changes raise points of usage. If the sentence من المعالمة أن المعالمة الم

وإن أخبرت من درهم بالذي قلت: الذي أعمل

عبدالله زيدا درهم ، تريد الذي أعطاه عبدالله زيدا درهم ، فعذ فت الها ، و يجوز إثبات ، وللا أن تقول : الذي أعطى عبد الله زيدا إياه درهم ، وهم القياس لأنك جعلت منمبر الررهم في موضعه ألا ترى أنك لو جعلت في هذه المسألة موضع الرهم عمرا ، لم يحسن أن تجعل الضمير إلا في موضع المفعول الثان لائه مليس ، وليس في موضع الذي لا يكون إلا مأخوذا ولا يكون الا مأخوذا ولا يكون الديلون المدا ... فإن قلت ذللا بالأنف واللام قلت : المعطيه عبد الله زيدا درهم . وإن شئت قلت : المعطى عبد الله زيدا إياه درهم . وهو القياس المعطى عبد الله زيدا إياه درهم . وهو القياس المعطى عبد الله زيدا إياه درهم . وهو القياس كما خبرتك .

ومن نصب المصادر إذا كانت نكرة في الحال لم يجز الإخبار عن الحال. يجز الإخبار عن الحال. وإذا كانت المعادر وغيرها أيضا عالا فيها الانن واللام لم يجز أن تخبر عنها نحو: أرسلها الوالا،

والقرم فيها الجماء الففير ورهم عوده على بدئه وما أنشبه هذا صاهباء عالا وهو معرفة وكل ما شد من عابه فليس لنا أن نتصرف فيه ولا نتجاوز ما فكموا به.

The criterion used to reject relative predication with infinitives in <u>hāl</u> constructions is that such usages are basically anomalous and cannot be subject to changes which can validly be applied to regular constructions. If relative clause predication is possible with infinitives there would seem to be no real reason why it should not be admissible when the infinitive is analysed as a <u>hāl</u>, even if those uses of the <u>hāl</u> where the term is indefinite cannot logically be made subject to this process. This confirms that the grammatical methodology which Ibn al-Sarrāj employs does not take into account whether constructions thrown up by relative clause predication are likely to occur in practice but, rather, whether they are admissible on the basis of considerations dictated by purely formal grammatical analysis.

Although the Arab grammarians would explain <u>giyas</u> at work in the Arabic language and would also actively make use of <u>giyas</u>, nevertheless there were constructions and usages which seemed to fall outside the working of <u>giyas</u>. In dealing with such situations the grammarians would employ what is called <u>tagdir</u> and it has, as Weil points out, a strong connection with <u>giyas</u>, although it is used in a variety of ways. 34

In spite of the fact that the idea of <u>taqdir</u> is very important in the thought of the Arab grammarians, no attempt was made by them to define the term and the various aspects of its use. Thus, although it is convenient and legitimate to talk about <u>taqdir</u> as a definable aspect of grammatical methodology, it would be incorrect to say that it was a technical term of the same order as for instance the term 'amil.

The grammarians make use of the actual word tagdir in expressions like ..., "implying...", and ..., "what is implied is...". They do not use the word in the sense of a precisely defined procedure which they would apply in certain situations.

Although it is convenient to talk of "the Arab grammarians" use of tagdir" when discussing their methodology, it must be remembered that while the grammarians certainly had their methodology they contented themselves with simply applying it in practice and they did not feel a need to discuss its techniques in general. For this reason the expression tagdir is not used as a strictly defined methodological term and, indeed, there are many discussions of points in which the word tagdir or its derivatives is not employed but in which the grammatical reasoning could be classified as tagdir.

Many Arabic constructions were treated by the grammarians as being more or less elliptic and by the use of tagdir they could remove this elliptic element through changing the sentence structure and adding explanatory elements, and thereby assisting the process of grammatical analysis. Describing tagdir Weil writes: "Er Endert den Wortlaut der Ueberlieferung, indem er eine Umstellung oder Ergenzung vornimmt, und schnell hat er den neuen Text in Einklang mit irgend einer der erlaubten Analogien gebracht." 35

Where tagdir was of great importance was in the analysing of constructions which occur in poetry and the grammarians used it to seek acceptable explanations for usages which seemed to violate the established rules of grammar. Many such usages could be justified by tagdir although others simply had to be classed as anomalous and defying any grammatical explanation. An example of this use of tagdir occurs when Ton al-Sarraj discusses inversion with the verb kana although he does not employ the word tagdir itself or any of its derivatives:

فها أهزته في المبتدأ والخبر من التقديم والتأخير فأجزه فيها ونكن لا تفصل بينها وبين ما عملت فيه بما لم تعمل فيه ولا تقل . كانت زيدا الحمى تأخذ ، ولا : كان غلامه زيد يضرب . لا تجز هذا إذا كان زيد والحمى اسين لكان . فإن أضمرت في كان الأمر أو الحديث أوالقصة وما أشبه ذلك — وهم الذي يقال له المجلول — كان ذلك المضمر اسم كان وكانت هذه الحملة خبره فعلى ذلك يجوز : كان زيدا الحمى تأخذ . وعلى هذا أنشدوا :

فَأَصْبُواوَالنَّوَى عَلَى مُعَرَّسِهِم وَلَيْسَ كُلَّ النَّرَى يُلِقِى الْسَاكِينُ

كأنه قال: وليس الخبر يلقى المساكين كل النوى ، ونكن هذا المضمر لا يظهر.

According to Ibn al-Sarraj a construction like is admissible if a term like amr, hadith, or gissah is implied after kāna which is a device that the Arab grammarians considered analogous to the explicit damīr al-sha'n. The support of this usage Ibn al-Sarraj also cites a verse into which such a notional term must be introduced to allow a grammatically sound analysis. His approach to this question is rather different from that of Sibawayh in the Kitab. The latter puts forward the idea of a suppressed term with certain usages of kāna so as to explain the syntax of a number of verses of poetry including the one cited above, but he does not consider it

admissible to form sentences like على نيدا الحمى تاني on the analogy of these purely poetical usages. 39

The notion of tagdir can be used in a manner which seems strained even allowing for the methodology of the Arab grammarians. An example of this occurs in Ibn al-Sarraj's discussion of the question of inversion of an attribute and the term which it qualifies: 40

لا يحور أن تقدم الصفة على الموصوف ولا أن تعمل الصفة في قبل الموصوف ولاتقدم شياعا يتصل طالصفة على الموصوف وكذلا السل اذا قلت: مرت برجل فارب زيدا ، لم يحر أن تندم زدا على رجل وكذلك إذا قلت: هذا رجل يضرب زما ، لم يحز أن تقول : هذا زمدا رجل يضرب ، لدُّن الصفة مع الاسم مُنزلة الفني والواهد وكذلك كل ما اتصل بها فإذا قلت: عبد الله رمل بأكل طمامل لم يجز أن تقدم طماملا قبل عبد الله ولا قبل رهل والكوفيون فيرون إلغاء رجل فيحملونه كنزلة ما ليس في الكلام فيقولون: طعامك عبد الله رحل مأكل لا يعتدون يرحل. و نقد يره عندهم: طماملا عبد الله مأكل والماء هذا فير معروف ، و للايفا ، هقرق سنذكرها إن شاه الله. ولكن هذه المسألة تجدر على غير ما قدروا ، وهو أن مجعل رجلا مدلا من عبد الله ترفعه مالاستداء ومجمل يأكل هبرا. لحيسند يملح تقديم . il lalo

In this passage Ibn al-Sarraj first takes the view that it is

inadmissible to change the construction which is inadmissible to change the construction which is inadmissible and he goes back on the general rule which he laid down governing the view that the laid down governing to because this violates the rule that an attribute predicated of a term cannot govern anything preceding that term. The Kufans, however, permit this construction by taking the view that the term is to be disregarded, but Ibn al-Sarrāj considers it incorrect to do this. Nevertheless, he does procede to permit the construction if the word is treated as a badal, although in doing, so he in effect withdraws his previous contention that the construction is inadmissible and he goes back on the general rule which he laid down governing the positioning of terms grammatically dependent on an attribute.

The concept of <u>taqdir</u> is particularly used in the discussion of constructions in which the speaker's intentions are seen as dictating how precisely he expresses himself. This often involves consideration of differences of expression which have little or no effect on meaning. This is illustrated in the discussion, which Ibn al-Sarrāj quotes from al-Māzinī, of the sentence منافر المنافر المنافر

قال المازق في الإهبار عن الدرهم: المعطيه عبد الله زيدا درهم في في الدرهم معلقا بالمعطى لانلا إذا قدرت على الها الم تجي باياه . ألا ترى أنلا تقول: فتربت إياه ، قال: وإن تقول: فتربت إياه ، قال: وإن نئت قلت المعطى عبد الله زيد إياه درهم في موضعه إذا كان مظهرا. في ما من مسن .

قال أبو بكر [ ابن السراج]: وهذا الذي قال المازن إنه مذهب هم عندي الأجود. According to al-Mazini, if in relative clause predication the inseparable pronoun is used attached to the participle this is done on the "implication" of the use of the pronoun object with the verb proper, but if the inseparable pronoun prefixed by the particle <u>iya</u> is introduced later on in the sentence this is done on the "implication" of the position that the substantive for which it stands would have occupied.

Another example of the use of <u>tagdir</u> involving a purely notional distinction is found in Ibn al-Sarrāj's discussion of the expression . From the standpoint of grammatical analysis this construction can be looked upon as either representing the use of the <u>tamyiz</u> or as analogous to the use of the active participle governing the direct object. According to Ibn al-Sarrāj the analysis adopted has a certain grammatical consequence: 42

وأما قدلك: الحسن وجها والكريم أبا, فإن أصحابنا يسبه منه بالضارب رجلاه قد قدست تفسيره في هذا الكتاب. وغير صحتنع عندى أن ينتصد على التمييز أيضا, بل الأصل ينبغى أن يكون هذا وذلك الفرج لانك قد بينت بالوجه الحسن منه كما بينت في قولك: هو أحسنهم وجها. وكذلك يجرى عندى قوله: هو أحسنهم وجها. وكذلك يجرى عندى قولهم: هو الصقور كلبا، وما أشبه ذلك فإذا نصبت على تقدير التمييز لم يجز أن تدخل عليه الألف واللام. فإذا نصبت على تقدير الضارب رجلا، جاز أن تدخل عليه الألف واللام. فإذا نصبت على تقدير الضارب رجلا، جاز أن تدخل عليه الألف واللام.

Here Ihn al-Sarraj draws the rather notional but not inconsistent conclusion that if the accusative in the expression الحسن الوب is used on the "implication" of the tamyiz the expression

is inadmissible, but if the accusative is used on the "implication" of a correspondence with the active participle followed by the direct object the expression is admissible. The approach adopted here seems to be peculiar to Ibn al-Sarrāj.

### Notes to Chapter IV

- 1 Ibn al-Anbari, Luma' al-adillah (Stockholm, 1963), p. 44.
- 2 Ibn al-Sarraj, vol. i, p. 61.
- 3 Ibid., vol. ii, p. 322.
- 4 Tbid., vol. i, p. 37.
- 5 Ibid., p. 101.
- 6 Ibid., pp. 97-8.
- 7 For a discussion of inversion see ibid., vol. ii, pp. 231-65.
- 8 Ibid., vol. i, pp. 100-1.
- 9 Ibid., p. 101.
- 10 Al-Zajjaji, Jumal, p. 55.
- 11 Ibn al-Sarraj, vol. i, p. 101.
- 12 Ibn Ya'ish, vol. vii, pp. 112-5.
- 13 Ibn al-Sarraj, vol. i, pp. 102-3.
- 14 Ibid., p. 219.
- 15 Ibn Ya'ish, vol. iii, pp. 114-8.
- 16 Qur'an 22:16. See also Wright, vol. ii, p. 299B.
- 17 Ibn al-Sarraj, vol. i, p. 218.
- 18 Ibid., p. 219.
- 19 Al-Astarabadhi, vol. ii, p. 28.
- 20 Ibid.
- 21 See M. S. Howell, A Grammar of the Classical Arabic Language, Introduction and 4 pts. (Allahabad, 1883-1911), pt. i, pp. 551-2.
- 22 Ibn al-Sarrāj, vol. ii, pp. 231-65; al-Mubarrad, vol. iii, pp. 89-132.
- 23 Ibn Ya'ish, vol. iii, pp. 156-60.
- 24 Al-Astarabadhi, vol. ii, pp. 44-53.
- 25 Ibn al-Sarraj, vol. ii, pp. 288, 289, 291. See also al-Astarabadhi, vol. ii, p. 44.
- 26 Ibn al-Sarraj, vol. ii, pp. 295, 298, 301, 310, 312, 313.

- 27 Ibid., pp. 288-9.
- 28 Al-Astarabadhī, vol. ii, p. 45.
- 29 Ibn al-Sarraj, vol. ii, p. 289.
- 30 Ibid., pp. 289-90.
- 31 Ibid., p. 295.
- 32 Ibid., p. 311.
- 33 Thid.
- 34 Weil, Introduction, pp. 25-7.
- 35 Ibid., p. 25.
- 36 Ibn al-Sarraj, vol. i, pp. 98-9.
- 37 Ibn Ya'ish, vol. iii, pp. 114-8.
- 38 Sibawayh, vol. i, pp. 36-7, 73-4.
- 39 See Howell, pt. ii, pp. 196-7.
- 40 Ibn al-Sarraj, vol. ii, p. 234. In 1. 2 of this passage the addition in the printed text after فيا ما ما المعامل has been omitted.
- 41 Ibid., pp. 295-6. In 1. 6 of this passage إذا has been substituted for إذا has been substituted
- 42 Ibid., vol. i, p. 272; see also pp. 158-9.
- 43 See Ibn Ya'ish, vol. vi, p. 84.

#### CHAPTER V

## QIYAS AND THE RELATIONSHIP OF ASL AND FARC

One of the effects of what the grammarians saw as <u>qiyas</u> at work in the grammar of Arabic was that grammatical phenomena which can be considered as similar tend to be governed by similar rules. At its simplest this often states little more than the obvious and, for example, is seen in the view of the grammarians that the passive subject is put into the nominative because it resembles the active subject in its function, or in the view that the active participle resembles the verb in its power of regimen because it is derived from the verb. Although at this level this aspect of <u>qiyas</u> is quite obvious, it can be extended and is seen by the grammarians to operate in a more involved manner. An example of this is the view of the grammarians that there is a relationship between the particle <u>inna</u> and the transitive verb due to the fact that both govern the nominative and the accusative. Although this would appear to be a purely coincidental similarity the grammarians did not treat it as such.

When a resemblance between two grammatical phenomena was observed the grammarians would procede to identify which of the two it was that the other had come to resemble. They would describe the basic phenomenon as the asl and the other which had come to resemble it as a far'. This same relationship of asl and far' was also considered to exist in respect of a single phenomenon if it had both a primary and a secondary aspect. For instance, the use of the particle wa as a conjunction proper was seen as its primary use (aşl), whereas its use with the maf'ul ma'ahu was seen as a secondary use (far'). Similarly, the use of the particle

fa as a conjunction was seen as its primary use, whereas its use with the verb in the subjunctive on the implication of a suppressed particle an was seen as a secondary use. When an English translation of the terms asl and far' is required, it is neccessary to resort to expressions like "ground-form" and "by-form" respectively, although in certain instances the terms can be translated as "primary usage" and "secondary usage".

In the grammatical writings particularly of later scholars the terms asl and far are frequently employed and for a discussion of the theory behind their use it is neccessary to turn to the Luma al-adillah of Ibn al-Anbari. In this work he explains that it is giyas which underlies the relationship between asl and far and, indeed, his discussion of giyas deals largely with how it determines the relationship between asl and far and he gives scant attention to other aspects of giyas. This is reflected in the way in which he defines giyas: 2

وه عرف العلماء عبارة عن تقدير الفرم بحكم الأصل وقبل: هو همل فرع على أصل بعل تقتضى إجراء هكم الأصل على الفرع وقبل: هو ربط الأمل مالفرع بجامع وقبل: هو اعتبار الشيء بالنبيء بحامع وقبل: هو اعتبار الشيء بالنبيء بحامع وهذه الحدود كلها متقاربة.

In the Luma' al-adillah much of Ibn al-Anbari's discussion of divas in grammar consists of a very literal application to it of the techniques of divas in figh in all its intricacies. However, this falls outside of the scope of this present study because such an approach is largely irrelevant to actual grammatical practice. The influence of figh can of course be seen in the definitions of divas quoted above. Inspite of the influence of figh, Ibn al-Anbari's discussion, in its more fundamental aspects, of how divas governs the relationship of asl and fare is worth quoting to explain this relationship: 3

لا بد لكل قياس من أربعة أشياء: أصل وفرع و علّة وهكم. وذلك أن تركب قياسا في الدلالة على رفع ما لم يصم فاعله فتقول: اسم أسند الفعل إليه مقدما عليه فعجب أن يكون مرفوعا قياسا على الفاعل. فالأصل هو الفاعل، والفرع ما لم يسم فاعله، والعلة الجامعة هي الإسناد، والحكم هو الرفع والأصل في الرفع أن يكون للأصل الذي هو الفاعل وإنما أجرى على الفرع الذي هو ما لم يسم فاعله بالعلة الجامعة على الفرع الذي هو ما لم يسم فاعله بالعلة الجامعة التي هي الإسناد. وعلى هذا السي تركيب كل قياس من أفيسة النعو.

The particular example given here to illustrate Ibn al-Anbari's explanation of the mechanics of <u>givas</u> is a rather simple and obvious one but the exercise shows how the relationship of <u>asl</u> and <u>far'</u> could be rationalised, although here the influence of <u>givas</u> in <u>figh</u> is strong.

The operation of <u>qiyas</u> is very important where the 'illah involved is what the grammarians called <u>tashbih</u> and the relationship of <u>asl</u> and <u>far'</u> of the verb and the particle <u>inna</u> provides one of the best examples of such a relationship based on <u>tashbih</u> because it is an example to which the grammarians gave particular attention. <sup>4</sup> Ibn al-Sarraj observes that <u>inna</u> and analogous particles resemble the verb because they all govern the nominative and the accusative and he considers it significant that these particles end in an indeclinable <u>fathah</u> like the perfect tense of the verb. <sup>5</sup> Although Sibawayh makes clear the resemblance of these particles to the verb, he does not mention this formal consideration, but this point was picked up by the later grammarians who also considered significant in this respect the use of the <u>nun al-wiqayah</u> with <u>inna</u> and similar particles. <sup>6</sup>

The al-Sarraj deals very briefly with the relationship to the verb of  $\underline{inna}$  and analogous particles: 7

فإن تشبه من الأفعال ما قدم مفعوله محد: طرب زيدا رهبل. وأعملت هذه الأهرف في المستدأ والحسر كما أعملت كان وفرق بين عمليهما بأن قدم المنصوب بالحروف على المرفوع كأنهم هملوا ذلك فرقا بين الحرف الفعل.

The Ya'ish, however, deals with this question much more fully: 8

وشهرت من الأفعال بما قدم مفعوله على فاعله مقولك: إن زيدا قائم ، بمنزلة: ضرب زيدا رجل. وإنها قدم المنصوب فيها على المرفوع فرقا بينها وبين الفعل. فالمعل من حيث كان الأمل في الحصل عبين الفعل. فالمعل من حيث كان الأمل في الحصوب فيها على المنفوع على المنصوب أد كانت رتبة الفاعل مقدمة على المفعول. وهذه المحروف لما كانت في العمل فروعا على الأفعال وصحمولة عليها جعلت دونها بأن قدم المنصوب فيها على المرموع عظا لها عن درجه الأفعال إذ فيها على المرموع عظا لها عن درجه الأفعال إذ فيها على المرموع عظا لها عن درجه الأفعال إد فيها على المواعل المناعل أمل...

In the passage from the <u>Usul</u> Ibn al-Sarraj states that the verb differs from <u>inna</u> because in the case of the latter the term in the accusative must precede the one in the nominative, as if thereby a distinction is drawn between particle and verb. However, in an earlier passage in the <u>Usul</u> he states more definitely that it is the mark of differentiation between them. <sup>9</sup> Ibn al-Sarraj does not introduce the terms <u>asl</u> and <u>far</u> into the above discussion but Ibn Ya ish does establish a relationship of <u>asl</u> and <u>far</u> between the verb and <u>inna</u>.

The grammarians came to hold the view that the <u>far</u> in comparison to the <u>asl</u> is subject to certain restrictions in its usage and al-Suyuţi gathers together a number of examples of this in the <u>Ashbah wa-l-nazā ir</u>. In the case of <u>inna</u> Ibn Ya'ish observes that its inferior status as a <u>far</u> is manifest in the fact that with it the term in the accusative must precede the term in the nominative, inasmuch as with the verb the precedency of the term in the nominative over the term in the accusative embodies the <u>asl</u> whereas the placing of the accusative in front of the nominative embodies a <u>far</u>. Ibn al-Sarrāj simply treats the word order with <u>inna</u> as a mark of differentiation but Ibn Ya'ish treats it as the sign of the lower status of the <u>far</u> with regard to the <u>asl</u>.

Although the use of the terms asl and far to describe such a relationship as was felt to exist between the verb and inna is not found in the <u>Kitāb</u> of Sībawayh, the use of these terms represents a development of certain ideas about grammatical relationships to be found in the <u>Kitāb</u>. Although Ibn al-Sarrāj does not make the point in a definite manner that <u>inna</u> is inferior in status to the verb, a fact which Ibn Ya'īsh attributes to the nature of the relationship of <u>asl</u> and <u>far</u>, this point is made by al-Khalīl in the <u>Kitāb</u>: 11

ورعم الحليل أنها عملت عملين الرفع والنصب من ملت كان الرفع والنصب هين قلت : كان الرفع والنصب هين قلت : كان أخوك أخاك زيد إلا أن تقول : كأن أخوك عبد الله أخوك لأنها لا عبد الله أخوك لأنها لا تصرّف الأفعال ولا يضمر فيها المرفوع كما يضمر في كان ومن أم فرقوا بينها كما فرقوا بينها كما فرقوا بينها المرفوع مين ليس وما فلم مجروها محراها.

In considering the relationship between <u>inna</u> and the verb al-Khalil takes the view that with <u>inna</u> the nominative cannot be made

to precede the accusative because there are certain restrictions on its use in comparison with the verb and these serve as a means of differentiation between them. The idea behind this is that when something resembles something else in the way in which it functions, there is a certain force at play which prevents a complete assimilation with regard to their respective rules of use, and this acts to maintain their separate identity. In this connection al-Khalil compares the verb laysa with mā al-hijāzīyah. In the Hijazi use of the negative particle mā it becomes assimilated to the verb of negation, laysa, because it takes its predicate in the accusative case. However, there are certain restrictions on this use of mā which do not apply to the verb laysa, among which is the condition that the subject must precede the predicate. <sup>12</sup> In short, it may be said that in the passage from the Kitāb quoted here may be seen the basis of the later idea that the fare is inferior in status to its asl.

Although the discussion of <u>inna</u> in the <u>Usul</u> does not give much information on contemporary grammatical theorising about its relationship to the verb, considerable detail on this topic can be found in the slightly later <u>Idah</u> 'ilal al-nahw of al-Zajjājā. This is found in the course of a discussion of the varieties of 'illah which occur in Arabic grammar and this is illustrated predominantly through reference to the particle <u>inna</u>. <sup>13</sup> Al-Zajjājā classifies the varieties of 'illah as being three, of which the latter two have particular relevance to the present discussion. The first variety of 'illah is made up of the 'ilal al-ta'līmīyah which relate to the established rules of grammar which a teacher would set forth to his pupils: <sup>14</sup>

فأما التعليمية فهى التي يتوصل بها إلى تعلَّم كلام العرب لأما لم نسمع نحن ولا غيرنا كل كلامها منها لفظا. وإنها سمعنا بعضا فقسنا عليه

نظيره ... فمن هذا النوع من العلل قولنا: إن زيدا فأعًا إن قيل بم نصبت زيدا ، قلنا : بإنّ لأنها تنصب الاسم و ترفع الخبر للناكذ لك علمنا و نعلمه من نوع التعليم ومه من نوع التعليم ومه من ط كلام الحرب.

The second variety of 'illah arises from the grammarians' explanations of points like the relationship of inna to the verb: 15

فالما العلة القياسية فأن يقال لمن قال: مصبت زيدا بان، في قوله: إن زيدا قالما: ولم وجب أن تسمس إن الاسم . فالجواب في ذلك أن يقول: لأنها وأخواتها ضارعت الفعل المتعدى إلى مفعول فحملت عليه فأعملت أعمال لما صارعته. فالمنسوب بها مشبه بالممعول لفظا والمرفوع بها مشبه بالماعل لفظا، فهي تشبه من الأفعال ما قدم مفهوله على فاعله نحو: ضرب أخاك محمد، وما أشبه ذلك.

It has been mentioned above that Ibn al-Anbari particularly defines givas in terms of how it underlies the relationship of asl and far' and it is of interest that al-Zajjāji should describe as an 'illah givasiyah the 'illah which he considers to govern what is in fact a relationship of this type, although he does not expressly mention this here.

The third variety of 'illah takes up as it were where the second variety leaves off: 16

وأمَّا العلدُ الجدلية النظرية فكل ما يعدل به في باب إن بعد هذا حنل أن يقال: فمن أي

مهة شادهت عده الحروف الاصال. وبأى الأصال سنه سهد ما الحاصية أم المستقبلة أم الحادثة في الحال أم المترافية , أم المنتصية بلا مهلة . (وهين) شيهتموها بالأفعال لأى شيء عدلتم بها إلى ما قدم مصوله على فاعله محد مرب ربدا عمره وهلا شبهتموها بما قدم فاعله على مضعوله لأن قدم فاعله على مضعوله لأنه هو الأمل وذلا فرع قان . فأى علق مضعوله لأنه هو الأمل وذلا فرع قان . فأى علق دعتال إلى إلحاقها بالعروع دون الأصول وأى قياس اطرد للا في ذلك.

In itself al-Zajjājī's discussion of this 'illah is an indication of the high stage of development which grammatical theory had reached by the first half of the 10th century. In this passage al-Zajjājī uses the terms asl and far' not with reference to the relationship itself of inna to the verb but to the fact that placing the subject before the object in a verbal sentence represents the asl, whereas placing the object before the subject represents a far'. This point was also made by Ibn Ya'ish in the passage from the Sharh al-mufassal quoted earlier. Al-Zajjājī indicates that he will answer the questions which the passage quoted above and its continuation raise but unfortunately he does not in fact do so. It would have been of some interest to know his precise answer to his own question:

وملا شبهتموها بها قدم فاعله على مفعوله لأنه هو الأصل وذلك فرع ننان. فأى علمة دعتك إلى إلحاقها بالفروع دون الأصل و أى قياس اطرد في ذلك.

However, it seems clear that the answer would have been very similar to that which could be given by consulting the passage quoted from

The Ya'ish: that through this is manifest the inferiority of the far' to the asl.

Before considering further the sort of relationship which can be described as that of <u>asl</u> and <u>far</u>, it is of value to consider what use Ibn al-Sarraj makes of these two terms occurring together. In analysing the construction , which has already been discussed, Ibn al-Sarraj uses the terms <u>asl</u> and <u>far</u>: <sup>17</sup>

وأما قولك الحسن وجها، والكريم أبا ، فإن أصحابنا يشبهونه بالصارب رجلا وقد قدمت تفسيره في هذا الكتاب وغير ستنع عندي أن ينتصب على التمييز أيضا، بل الأصل ينبغي أن مكون هذا وذلك فرع لأنك قد بينت بالوجه الحسن منه كما جينت في قولك : هو أحسنهم وجها.

Ibn al-Sarraj considers the analysis of the construction المن ومع as a tamyiz to represent the asl, whereas the analysis based on an affinity to the construction الصارب رملا , to which the former construction has been assimilated, represents a far'. This use of the terms asl and far' does not involve the strict technical meaning discussed in this chapter.

Another instance of the use of the terms asl and far is found when Ibn al-Sarraj discusses possible alternative word order for the sentence وأبوها والمرابعة وأبوها والمرابعة على المرابعة المر

فإن قلت: مند أبوما قائم ومنطلقة , هاز والأهمن عندى أن تقدم منطلقة لأن الأمل للمعرد والحملة فرع. ولا ينبغى أن يقدم الفرع على الأمل إلا في صرورة شعرهم.

Here a predicate which consists of a single term is seen as more basic

than one which consists of a sentence complete in itself. Although Ibn al-Sarraj uses the terms <u>asl</u> and <u>far</u> in this connection, these terms again are not used in a strict technical sense.

Ibn al-Sarraj also uses the terms usul and furus in the course of a discussion of the rather rare use in positive statements of the far al-sababiyah: 19

ألا ترى أن الشاعر، إذا اضطر معطف على الفعل الواجب الذي على فير شرط بالفاء وكان الأول سب اللثاني نصب كما قال:

سَأَتُرُكُ مُنْزِلِي لِمِن يَمِيمٍ وَأَنْفَ بِٱلْجَازِ فَأَسْتَرِيحًا

هدل لحاق بالحجار سبما لاستراحت فتقديره لما نصب كأنه قال: يكون لحاق ها ستراحة . وقد ها مثله في الشعر أبيات لقوم فصاء إلّا أنه قبيح أن تنصب وتصطف على الواجب الذي على غير شعر آلائه متد جعل لهذا المعنى آلات عكان حق الكلام أن يقول لوكان في غير شعر ]: وأني حق الكلام أن يقول لوكان في غير لشعر ]: ألحق أسترع . ومع ذلك فإن الإيحاب على غيرالشرط أصل الكلام . وازالة اللفظ عن جهته في الووع أمس صدها في الأصول لأنها أدل على المحاني .

The meaning of the general remark on <u>usul</u> and <u>furu</u> as it applies in this particular instance is that the use of the particle <u>fa</u> purely as a conjunction represents the <u>asl</u>, whereas its use with the subjunctive represents a <u>far</u>; and it is better to go beyond the normal use with the

fa al-sababiyah because the use of the subjunctive clearly marks its function, whereas <u>fa</u> used simply for conjunction does not have an analogous accompanying marker of its purpose. In this context the terms <u>asl</u> and <u>far</u> are used as strict technical terms unlike their use in the two preceding instances cited.

From an examination of the occasions on which Ibn al-Sarraj uses the expressions asl and far' it is clear that he makes little use of them in the discussion of relationships which later grammarians would define in terms of an asl and far'. However, in his discussion of certain questions he does refer to a point which the grammarians took to be a principle involved in the relationship of asl and far'. It has been mentioned above that the view was taken that a far' is inferior in status to its asl and this inferiority is manifest in certain restrictions on the use of the far' which do not apply to the asl. In the Usul there are some examples of this line of thought, although it is not expressly put forward in terms of the relationship of asl and far'.

When discussing the position of the  $\underline{\text{maf'ul ma'ahu}}$  in the sentence Ibn al-Sarraj writes:  $^{20}$ 

ولا يجوز التقديم للمفعول في هذا الباب. لا تقول: والخشية استدى الهاء , لأن الواد أهلها أن تكون بعد تكون للعطف , وهق المعطوف أن يكون بعد المعطوف عليه , كما أن هق الصفة أن تكون بعد بعد الموصوف عليه , كما أن هق الصفة أن تكون بعد بعد الموصوف . فقد أ هرجوا الواد في هذا الباب عن هده الذى كان له وص شأنهم إذا أفرجوا الشى ، من هده الذى كان له ألرسوه هالا واهدة .

In the Ashbah wa-l-naza ir al-Suyūţi quotes Ibn 'Uṣfur on the same

question in the course of discussing the point that  $\underline{\text{furu}}^{\epsilon}$  are inferior in status to  $\underline{\text{usul}}$ :

وقال ابن عصفور في ( شرح الجول) لما كان جعل الواو بحوني مع في المفعول محه فرعا عن كونه عاطفة , لم يتصرفوا في الاسم الذي بعدها ولم يقدموها على العامل ، وإن كان متصرفا , ولا على الفاعل . لا يقولون : والطيالسة جاء البرد , ولا : جاء والطيالسة البرد , ولا : جاء والطيالسة ما تحتمل من التصريف ما تحتمل من التصريف ما تحتمل الأصول .

Although Ibn al-Sarraj uses the term asl in the passage from the Usul he does not introduce the term far' to describe the special use of the conjunction wa with the maf'ul ma'ahu. However, the word asl is used in a way which would lead on to the use of the word far', as the quotation from Ibn 'Usfur shows. This scholar uses these expressions to refer to the primary and a secondary function of the conjunction wa.

The al-Sarraj takes the view that the position of the maf'ul ma'ahu in the sentence is very firmly defined because the particle wa is basically used for conjunction and the element which it is used to add on, naturally, follows what it is added on to. In dealing with the same point Ibn 'Uṣfur states that there are restrictions put on the position of the maf'ul ma'ahu because the accompanying particle wa functions as a far' in relation to its use as a conjunction proper which represents the asl. Accordingly, there are such restrictions because furu' do not have the range of permitted use which characterises usul.

Although Ibn al-Sarraj gives a practical explanation of why the maf'ul ma'ahu should be restricted in the position it may take in the sentence, it is apparent that he does take what is tantamount to the

more theoretical view of Ibn 'Uşfur but he expresses it in somewhat different terms:

وقد أخرجوا الواوفي هذا الباب عن هده الذي كان له . ومن شأن الحرب إذا أخرجوا الشيء عن هده الذي كان له الراموه ها لا واهدة.

The general principle stated here that a usage which is to be seen as an extension of a more basic usage is subject to restrictions which do not apply to the basic usage, is again taken up by Ibn al-Sarrāj when he discusses the use of the <u>fā' al-sababīyah</u> and the subjunctive:

وص سأن الحرب إذا أرالوا الكلام من أصله الى شىء آخر فيروا لفظه وهذفوا منه شيئا والرموه موضعا واهدا، إذا لم يأنوا محرف بدل على هذا المحنى، ولم يصرفوه وجعلوه كالممثل ليكون ذلك دليلا لهم على أمهم ها لعوا به أصل الكلام.

It has been mentioned earlier that the use of the particle <u>fa</u> with the subjunctive is considered by Ibn al-Sarraj to be an extension (far') of its basic use (aṣl) as a conjunction and in this context the general remark about such extended usages indicates that the verb following <u>fa</u> must be in the subjunctive because this alone marks the special function of the particle.

In the above passage from the <u>Usul</u> the phraseology used in describing usages which are extensions of more basic usages is of interest. The use of the expression الربوء صوفا واهدا may be compared with that of the expression " used by used by Ibn al-Sarraj in connection with the <u>maf'ul ma'ahu</u>. The verbs <u>sarrafa</u>, tasarrafa, and their derivatives are commonly used by the grammarians

In discussing the relationship of asl and far', and in the above passage The al-Sarraj uses the verb sarrafa in making the point that a usage which is an extension of a more basic usage has certain restrictions placed on it which do not apply to the basic usage ( لم صوره ). Although questions of asl and far' are not discussed as such in the Kitab of Sibawayh, the verb sarrafa is used in the sense referred to here because al-Khalil states in a passage quoted earlier when he deals with the point that with inna and analogous particles the nominative cannot be made to precede the accusative: الأفعال على المنافعة على المنافعة

By the 10th century the grammarians were discussing relationships of asl and far' but it is unclear in what precise stages the grammarians developed their ideas on this subject. As has been mentioned, the subject is not one which is dealt with in the <u>Kitāb</u> although it did develop out of ideas present in that work. In the <u>Uşūl</u> Ibn al-Sarrāj does give some attention to what are relationships of asl and far' but he rarely uses these terms themselves, and al-Mubarrad in the <u>Muqtadab</u> shows no interest at all in this topic. It seems likely that at the time of Ibn al-Sarrāj relationships of asl and far' were treated as a rather theoretical subject and discussion of it did not intrude to any great extent into general works of grammar. Indeed, it is from a theoretical work from this period, the <u>Idah</u> 'ilal al-nahw of al-Zajjājī, that some idea can be obtained about the stage of development of grammatical theory in this field as it affects the particle <u>inna</u>.

### Notes to Chapter V

- 1 Ibn al-Anbari, Luma', pp. 42-58.
- 2 Ibid., p. 42.
- 3 Told.
- 4 Al-Suyūtī, Iqtirab, p. 48.
- 5 Ibn al-Sarraj, vol. i, p. 278.
- 6 Sibawayh, vol. i, p. 279; Ibn al-Anbari, Insaf, p. 82.
- 7 Ibn al-Sarraj, vol. i, p. 278.
- 8 Ibn Ya'ish, vol. i, p. 102.
- 9 Ibn al-Sarraj, vol. i, p. 113.
- 10 Al-Suyūti, Ashbah, vol. i, pp. 260-4.
- 11 Sibawayh, vol. 1, p. 280.
- 12 See Wright, vol. ii, p. 104A; Ton Ya'ish, vol i, pp. 108-9.
- 13 Al-Zajjaji, Idah, pp. 64-6. See also E.I.<sup>2</sup>, s.v. 'illah.
- 14 Al-Zajjaji, Idah, p. 64.
- 15 Ibid.
- 16 Ibid., p. 65.
- 17 Ibn al-Sarraj, vol i, p. 272. See also pp. 98-9 above.
- 18 Ibn al-Sarraj, vol ii, p. 63.
- 19 Ibid., pp. 189-90. In 11. 9-10 of this passage the material in brackets has been supplied from the quotation in al-Baghdadī, vol. iii, p. 600. See also Wright, vol. ii, pp. 30-2, esp. p. 32B.
- 20 Ibn al-Sarraj, vol i, p. 256. In l. 4 of this passage المعطوف عليه has been substituted for المعطف عليه in the printed text.
- 21 Al-Suyuti, Ashbah, vol. i, p. 262. For Ibn 'Usfur see E.I.2, s.v. Ibn 'Usfur.
- 22 Ibn al-Sarraj, vol. ii, p. 189.
- 23 See p. 106 above.
- 24 Ibn al-Sarraj, vol. i, p. 116: vol. ii, p. 349.

#### **OHAPTER VI**

## THE TAMYIZ AND THE VERB OF WONDER

In the preceding chapters attention has been concentrated on the techniques and procedures of the Arab grammarians but in this chapter a different approach will be followed by way of contrast and an examination will be made of the way in which two specific topics of Arabic grammar are treated. As it would be impossible to examine every topic dealt with in the grammars, attention will be given in this chapter to the tamyiz and the verb of wonder. Although in the previous chapters various grammatical topics have been touched on when dealing with the techniques and theory of the grammarians, nevertheless, by a more detailed examination of specific topics a different perspective on the work of the grammarians can be obtained.

## Tamyiz

It is curious that although there are a number of uses of the accusative which are clearly analogous and which are generally known as the tamyiz, Sibawayh does not have a term to cover such uses of the accusative, inspite of the fact that he does discuss several of them.

Indeed, in the Kitab there is no unified discussion of usages which could be seen as involving tamyiz. Although there is no term for tamyiz in the Kitab, the term mufassir is used in the Ma'ani l-Qur'an of al-Farra' who was a pupil of Sibawayh's Kufan contemporary, al-Kisa'i. In the Muqaddimah fi l-nahw of the Basran scholar, Khalaf al-Ahmar (d. 796), the expression is used for the noun functioning as a tamyiz with the numbers 11-99 and this term is clearly

based on the fact that a noun in the singular is here used for what is plural in meaning. 3 However, this expression, which is limited to one particular use of the tamyiz, was not generally accepted and by the time of al-Mubarrad the familiar and more comprehensive expression tamyiz was in use together with other less important terms like tabyin and tafsir. 4

Sībawayh's basic discussion of constructions which would be recognised as being examples of the tamyīz is actually found in his chapter on the assimilated adjective (sifah mushabbah) and this deserves some explanation. Sībawayh considered that the assimilated adjective could be treated as similar to the active participle because both could be connected with another noun through annexation or through governing it in the accusative as in the expressions ما المعارف على ال

It is worth mentioning that when Ibn al-Sarraj deals with the tamyiz in the Usul he adopts a format different from that found in other grammars because his discussion of the tamyiz is split into two separate parts which do not run consecutively. One chapter on the tamyiz deals with constructions where the regent is a verb or its equivalent and these are the majority of tamyiz constructions. The other chapter deals with the use of the tamyiz in enumeration and measuring. This division is based on the fact that in the former case the tamyiz is dependent on a formally complete utterance like many other uses of the accusative, as has been explained in an earlier chapter, 8 whereas in

the latter case the <u>tamyiz</u> is dependent simply on the noun which precedes it.

In dealing with tamyiz the later scholars tried to define exactly what it is and al-Zamakhshari, for instance, begins his discussion of it by giving a comprehensive definition of it:

وحور رفع الإبهام في جملة أو صفرد بالنق على أهد مستسلات فمثال في الجملة طاب زيد نفسا ، وتصنّب عرفا ونفقاً شما ... وسئاله في الصفرد ؛ عند ي راقود مند ، ومنا ، وقفيزان برّا ، وعشرون درعينا .

Al-Mubarrad, on the other hand, defines the <u>tamyiz</u> in a far more cursory way and only illustrates his definition in terms of its use with the numbers 11-99: فعدناه أن يأق صبينا عن نرعه و ذلك قرلك:عندى أن يأق صبينا عن نرعه و ذلك قرلك:عندى أن تربًا وتلاتون تربًا مسرون درويًا وتلاتون تربًا not attempt to provide a real definition of the <u>tamyiz</u>.

One of the points which needs some attention in discussing the tamyiz is the question of the use of singular and plural nouns.

Al-Mubarrad turns to this point after discussing tamyiz constructions of the type أنت أفره الناس عبداً when he moves on to discuss the related expression أنت أفره الناس عبداً : 11

و مجوز أن تقول - وموهس جدا - أنت أفره الناس عبيدا ، وأجود الناس دورا . ولا مجوز عندى: عشرون دراهم يا فتى .

والفصل بسنهما أسك إذا قلت : محشرون ، ققد أتيت على العدد علم يحتج إلّا إلى ذكر ما يدلّ على الجنس فإذا قلت : هو أفره الناس عبدا ، هاز أن تعنى عبدا واهدا فسن أمّ هسن . واهتير ، إذا

أردت الحماعة ، أن تقول : عبيدا . قال الله عزّ وجلّ : قُلْ هُلْ أَنْبُنْكُمْ بَالْأَفْسَرِينَ أَعْمَالًا . وقد يجوز أن تقول : أفره الناس عبدا ، فنعنى جماعة العبيد فحد التعبير والحمو أبين إذا كان الأول غير محطور العدد .

After his own discussion of this point Ibn al-Sarraj quotes from this passage but his own explanation of the occasions on which it is possible to use the plural in the <u>tamyiz</u> is based on using the regent as the criterion: 12

واعلم أن الاسماء التي تنصب على النمييز لا تكون إلا مكرات تعدل على الأهماس وأن العوامل فيها إذا كن أفعال كنت فيها إذا كن أفعال كنت بالخيار في الاسم المميز إن نشت جمعته وإن سنئت وهديه تقول: طبتم بذلك نفسًا، وإن ننت أنفسًا، قال الله تعالى: فإن طبت كم عن شنى إسنه نفسًا، وقال الله تعالى: فإن طبت كم عن شنى إسنه نفسًا، وقال تعالى: فأن طبت كم عن بألا طبر إلى منا الموافرة الناس عبدًا، وأهود الناس دورًا.

Here Ibn al-Sarraj lays down somewhat more incisively the principle which is to be followed although he adds the quotation from al-Mubarrad to provide further detail.

In his supplementary section on the <u>tamylz</u> with a verbal regent The al-Sarraj introduces further discussion of this question when he quotes al-Mabarrad on a point of grammar in the Qur'an:

وقوله : فإن طبئ لكم عن شي منه نفسًا ، إن النيبز إدا لم يتم عدد مطوم كالعشرين والثلاثين جاز تبيينه بالواهد للدلالة على الجنس وبالحصيه إذا وقع الإلماس وللإلباس في هذا الموسع لقوله ، فإن طبئ لكم ، ولقوله : في يخرجكم . وقال : قد قال قوم : طفلًا حال ، وهذا أحسن ...

Although the question of number with the <u>tamyiz</u> is in essence relatively simple to understand, the grammarians still took a lot of trouble to explain it exactly. 14

Another point which comes up in discussions of the <u>tamyiz</u> is under what circumstances a term in the accusative functioning as a <u>tamyiz</u> may be replaced by the same term, but in the genitive governed by the preposition <u>min</u>. Sibawayh refers briefly to this usage but does not supply any specific rules: 15

(هذا عاب ما بنتصب انتهاب الاسم معد المقادير) ودلا قدلا وي رهد رهد رهد وي وسبلا به رجد رهد قدت وي من رجد رود المقادير) رجد رود المقادير وي من رجل والله در من رجل وتله در من رجل وتدخل من من رجل وتد در من رجل وتدخل من من رجل وتد در من رجل وتدخل من منها كدهولها في كم توكيدا.

Al-Mubarrad, on the other hand, attempts to provide a precise rule for when the preposition min can be used:

ومی النمییز وجه رحلاً ولله دره فارسًا و هسبلا به سخاعًا ، إلا آنه إذا کان فی الأول ذکر منه هشن أن تدخل من تحکیدا لذلا الذکر ، فتقول وهه من رجل ، ولله دره من فارس و هسبلا به من شجاع ، ولا بحوز : فشرون من درهم ، ولا : هو أفرهم من

## عبد، لأنه لم سذكره في الأول.

In his section on the use of the <u>tamyiz</u> in measurements Ibn al-Sarraj appears to borrow from this passage but adds an explanation of what precisely is meant: 17

... وادا كان في الأول ذكر منه حسن أن تدخل من تعكيدا لذللا الذكر . تقول: وجه من رجل والله در زيد من فارس وحسبلا به من شجاع . ولا يجوز عشرون من درهم ، ولا: هو أفرهم من عبد الأنه لم يذكره في الأول . وصفى قولهم : ذكر منه ، أن رجلا هو الها في وجه ، وفارس هو زيد ، ولا الأفره لأن الأفره خبر زيد .

In the <u>Usul</u> Ibn al-Sarraj also twice quotes a passage by al-Mubarrad on the use of <u>min</u> but it would appear to be derived from a work other than the <u>Muqtadab</u>. This passage does not lay down a rule for the use of <u>min</u> but attempts to explain the reason for its use: 18

وقال ابوالعباس (المبرد) - رحمه الله - فأمّا قولهم:

حسباد برید رجلا، وأکرم به فارسًا، وما أشه ذلك،

م تقول: حسباد به من رجل، وأکرم به من فارس،

ولله دره من شاعر، وأنت لا تقول: عشرون من

درهم، ولا: حر أمره منك من عبد، فالعصل بيمها

أن الأول كان يلتس فيه النمييز بالحال فأدهلت

من لخلصه للتسييز. ألا ترى أنلا لوقلت : أكرم

به فارسل وحسباد به خطيبا، لحاز أن تعنى في هذه

الحال، وكد للا إذا قلت كم ضربت رحلا، وكم ضربت

من رجل، جاز ذلاد لأن كم قد يتراهي عنه صيره،

فإن قلت : كم ضربت رجلا ، لم يدر السام اردت كم مرة ضربت رجلا واحدا أم كم ضربت من رجل، فد فول من قد أزال السلك.

In the <u>Mufassal</u> al-Zamakhshari makes no reference to the use of min instead of the <u>tamyiz</u> and what Ibn Ya'ish writes is not particularly useful. <sup>19</sup> The latter does make the point that min can separate the <u>bal</u> from the <u>tamyiz</u> and in doing this he incorporates without attribution into his own work part of the above passage which Ibn al-Sarraj quotes from al-Mubarrad. Later discussion of the use of <u>min</u>, such as is found in the <u>Manhaj al-salik</u> of Abū Hayyan, cannot be directly related to what is found in earlier works because, as has been mentioned in an earlier chapter, the grammarians eventually classified uses of the <u>tamyiz</u> according to how the various constructions could be analysed, and this influenced their discussion of the use of <u>min</u>. <sup>20</sup>

Al-Mubarrad was well-known for his readiness to depart from the grammatical views of Sibawayh and one of the points where he does so is in permitting the placing of the <u>tamyiz</u> in front of the verb which acts as its regent: 21

وأعلم أن التبيين إذا كان العامل فيه فعلا جاز تقديمه لنصرف العمل فقلت: تفقأت شحاً تفقأت، وعرقا عرقًا، فإن سنت قدمت فقلت: شحا تفقأت، وعرقا تصببت. وهذا لا يجيزه سيبويه لأنه يراه كقوللا: عشرون درها، وهذا أفرهم عبدًا. وليس هذا بمنزلة ذللا لأن عشرين درهًا إنها عمل في الدرهم ما لم يقذد من فعل. ألا ترى أنه يقول: هذا زيد قائمًا ، ولا يجيز: قائمًا هذا زيد، لأن العامل غيرفعل. وتقول: واكبًا جا زيد، لأن العامل غيرفعل. وتقول: واكبًا جا زيد، لأن العامل غيرفعل.

تقديم التمييز إذا كان العامل فعلا. وهذا رأى أبي عنهان المازني.

In his Nagd kitab Sibawayh, which is quoted by Ibn Wallad, al-Mubarrad criticises Sibawayh's view for being inconsistent and considers his own view to be supported by giyas and attested usage: 22

وقد أجاز (سيبويه) في الحال التقريم إذا كان العامل فعلا . وإنها الحال عنده وعند غيره بمنزلة التمييز فيلزمه هذا أن يجيز تقديم التمييز إذا كان العامل فعلا وإثلا ترك قوله في الحال ... وهاء في التعر تصديق حذا القياس وحرقوله:

# أَمَّهُ رُكُولًا لِلْفِرَاقِ مُبِيبَهَا وَمَا كَانَ نَفْسًا بِٱلْفِرَاقِ تَطِيبُ

In his discussion of <u>tamyiz</u> Ibn al-Sarraj mentions the view of al-Mubarrad on this question but he does not make his own view clear at this point. However, he rejects this usage later on in the <u>Usul</u> in the chapter on inversion where he bases his view on Sibawayh's analysis of the nature of the <u>tamyiz</u>: 23

وقياس بابه أن لا يجوز لأن عامل في الحقيقة وهو مخالف للمعمولات ، ألا ترى أنه إذا قال: تفقأت سخيًا ، فالشمر هو المتفق ، كما أنه إذا قال : هو طير عبدا ، فالحد هو طير .

The Ya'ish takes the same view but expresses himself in a rather more technical manner: 24

ألا ترى أن التصب في قولك تصبب زيد عرقا. وتفقأ شحيل في الحقيقة للعرق والتفقؤ للشعم ، والتقديم تصبب عرق زيد , وتفقأ شحمه . فله قد صفاها الأوقمناها صوقعا الا يقع فيه الفاعل الأن الفاطل إذا قد صفاه عرج عن أن يكون في تقدير فاعل نقل عنه الفعل ، إذ كان هذا موضعا الا يقع فيه الفاعل .

When dealing with the question of placing the <u>bal</u> at the beginning of the sentence Ibn al-Sarraj writes that the Basrans treat it like the <u>temyiz</u>, and this would indicate that the general view of the Basrans of this period was that the <u>tamyiz</u> could be placed at the beginning of the sentence when the regent is a verb. <sup>25</sup> Ibn al-Sarraj, however, does not take this view and later scholars like Ibn al-Anbari and Ibn Ya'ish consider the view which he takes to be the true Basran one. <sup>26</sup> In fact, there was always a division of opinion among Basran scholars on this point. <sup>27</sup>

#### Verb of wonder

The particular approach of the grammarians to the verb of wonder (fi'l al-ta'ajjub) led them to examine certain theoretical questions which their approach itself entailed. No single grammarian treats the theoretical questions raised by the verb of wonder in an exhaustive manner and there is a varying emphasis in the works of grammar on the different questions. It is of interest to see how al-Mubarrad and Ibn al-Sarraj deal with this subject and to compare their approach with that of later grammarians.

In dealing with the theoretical aspects of the grammarians' approach to the verb of wonder it is particularly worth remembering that the approach of the Arab scholars was rather different from that which a scholar today might adopt. The latter might look at the verb of wonder in a manner similar to Wright who, in explanation of the

examples المُفْضِلُ بريد and مَا أَفْضَلُ بينا writes: 28

The first formula literally means: what has made Zeid excellent? can anything make him more excellent than he is? The second: make Z. excellent (if you can,—you cannot make him more excellent than he is); or, more literally: try (your ability at) making excellent upon () Zeid.

The Arab grammarians did not attempt to interpret an expression such as علم أحسن نبدا as having the literal sense of "What has made Zayd excellent?", but representing, in fact, a standard formula for expressing wonder or astonishment. Rather, the majority of the Basrans followed and expanded the very briefly expressed view of al-Khalil which Sibawayh cites in commenting on the expression علم أحسى عبد 29

Although this view was accepted by the majority of the Basrans it is worth noting that there was a minority view held by the Kufan, al-Farra', and the Basran, Ibn Durustawayh, that the term ma used in expressions like المان زيدا is, in fact, interrogative and this would appear to be a more reasonable explanation as has been mentioned.

Al-Mubarrad begins his discussion of the verb of wonder by explaining how the example الحسن ريد is to be parsed: 31

فعا اسم مرتفع بالابتداء وأهسن خبره وهوفعل وزيد مفعولا به فتقديره: شي أحسس زيرا، إلّا أن سعى التعبيب دخله مع ما ولا يكون ذلك في شيء غير ما.

The explanation that <u>ma</u> replaces the term <u>shay</u> was not accepted by those who held that <u>ma</u> must have a clause dependent on it (silah) when used in senses other than its interrogative and conditional senses, but

al-Mubarrad argues that this view is mistaken: 32

قال قائل: هل رأيت ما تكون اسما سغير صلة إلا في المحراء والاستفهام؟ قيل له :إنها كانت في الجزاء والاستفهام بغير صلة إذا قلت مجازيا : ما تصنع أصنع، أو مستفهما : ما تصنع يا فتى ، لأملا إنما تستفهم محما تتنكر ، ولوكنت تحرف كنت مخبرا لا مستخبرا ، والعلة تعرّفه . وكذلك الجزاء إذا قلت : ما نصنع أصنع ، لأملا أبهمت ولم تقصد إلى شىء واحد بعينه . عالمحنى من الإبهام الذى يكون في الجزاء والاستفهام كذلك هو التعبد لأنك إذا قلت : ما أحسن زيدا , فقد أمهمت ذاك فيه ولم تخصى .

That the term ma is to be interpreted as having an understood <u>silah</u> is a view that was held by certain of the Kufans and is, according to Ibn Ya'ish, the most generally accepted view attributed to al-Akhfash. 33 Al-Mubarrad does explain this view but he does not specify who held it: 34

وقد قال قوم: إن أحسن صلة لما والخبر محدوقه وليس كما قالوا. وذلا أن الأهبار إنها عدف إذا كان في الكلام ما حدل عليها. وإنها هرموا من أن خكون ما وهد ها اسها. فتقد يرهم: الذي حسّن زيدًا شئ. والقول فيها ما بدأنا به من أنها عرى سفير صلة فالقول فيها الاستفهام والجزاء في الإجهام.

Al-Mubarrad also takes up the question of why an expression of the form ما أحسن نيدا should convey the idea of wonder and astonishment: 35

فإن قال قائل : فإذا قلت : ما أمسن زيدا ، يمنزلة :

شى؛ همت ريدا، فكيف دخل صفى التعبيب، وليس دلا في قوللا: شى أحسن زيدا؟ قيل له: قد يدخل المحتى في اللفظ ولا يدخل في نظيره. محمل ذلك قولهم: علم الله لأفعلن. لفظه لفظ: رزق الله ، ومعناه القسم. ومن ذلك قولهم: عفر الله لزيد. لفظه لفظ الخير ومعناه الدعاء. ومن ذلك أخك تقول: تالله لأفعلن، ومعناه الدعاء. ومن ذلك أخك تقول: تالله لأفعلن، فتقسم على صفى التعبيب. ولا تبدخل الناء على شىء من أسهاء الله فير هذا الملاسم لأن المحنى الذي يوجب التعب إنها وقع هاهنا. وكل ما لزمه شيء على صفى لم يتصرف لأنه إن تصرف بطل ذلك المعنى ومار التا عمرة الله عمرة الله عمرة الله عمرة الله عمرة الله عمرة الله على الله عمرة الله عمرة الله المعنى ومار المعنى ومار الدفعال التي تحرى على أصولها ، ولم يدهنها من المعنى أكثر من ذلك.

In this passage al-Mubarrad makes the point that the verb of wonder is essentially a formulaic expression with a special meaning associated with it. Although Sibawayh does not go into detail on questions of theory connected with the verb of wonder, the line of argument which al-Mubarrad employs in the above passage can be found in the <u>Kitāb</u> used in other circumstances. The same point Ibn al-Sarrāj makes use of an analogy often employed by the grammarians because he likens the verb of wonder to proverbs which, as set expressions, have an affinity with the formulae used for expressing wonder.

When he deals with theory connected with the verb of wonder Ibn al-Sarraj is much more concise than al-Mubarrad and, although he holds the same views, there are differences in the arguments which he uses. In explanation of how the verb of wonder is to be interpreted he writes: 38

فإذا قلت: ما أحسن ريدا , فما اسمستما وأحسن حمره وفيه ضمير الفاعل وزيد صحول به وما منا اسم تام غير موصول فكأنلا قلت: شيء حمّن زيدا , ولم تصف أن الذي حمّنه شيء بعينه ، فلذللا لزمها أن تكون صبهمة غير مخصوصة كما قالوا: شيء هادك ، أي: ما هادك إلّا شيء ، وكذللا : شرّ أحرّ ذا فاجى ، أي : ما أحرّه إلا شرّ ، ونظير ذللا : فرا أن أفعل ، يربد : إلى من الأمر أن أفعل . فلما كان الأمر مجهولا جملت ما بغير صلة . ولو فلما كان الأمر مجهولا جملت ما بغير صلة . ولو لوصلت لصار الاسم معلوما ، وإنها لرم الفعل المامي وحده لأن التجب إنها يكون مها وقع وتبت ، ليس مها يسكن أن يكون وجمكن أن لا يكون .

In this passage Ibn al-Sarrāj introduces several analogies to support the contention that the term mā is indefinite when used with the verb of wonder. The analogy drawn from the expression إن الله أن أن عامل is also discussed by al-Mubarrad who goes into more detail. 39

The al-Sarrāj by way of analogy also makes use of the expression المناف عن الله عاملاً عنه الله عاملاً عنه الله عاملاً عنه الله عاملاً عنه عاملاً عاملاً عنه عاملاً عاملاً عنه عام

فتقول: ما أحسن زيدا... فما اسم مبتدأ في صوفع الرفع وهي هنا اسم فيرموسول ولاسوسوف عمنى النفىء كأنك قلت منىء أحسن زيدا , ولم ترد شيئا بعينه . إنها في صبهمة كما قالوا: شيء جاء بل أي: ما حاء بل إلا شيء .

In a later passage Ibn Ya'ish also makes use of the proverb which Ibn

al-Sarraj cites, سر أهر ذاناب , "Tis an evildoer that makes a dog growl." 41

The view of the Arab grammarians on the nature of the verb of wonder has been compared above with that which scholars today would take and a point which Ibn al-Sarraj makes throws particular light on the view of the majority of Arab scholars: 42

وإنها لزم معل النجب لفظا واحدا ولم يُصرَف ليدلّ على التعجب ولولا ذلا لكان كسائر الأخبار لأنه على التعجب ويدلّ على أن خبر أنه يجوز للا أن تقول فيه: صدق أو كذب.

Ibn al-Sarraj takes the view that a sentence which contains a verb of wonder is a proposition admitting of truth or falsity. However, according to the modern interpretation a sentence like ما أحسن زيدا would be seen as having the outward form of a question and could not admit of truth or falsity. Even if the meaning of such a sentence is taken into account, which in English idiom would be "How excellent Zayd is!", the sentence is rather an exclamation or an ejaculation and not a proposition. However, for Ibn al-Sarraj there is no question that a sentence like المسمن زيد which clearly admits of being true or false.

Ibn Ya'ish takes this same consideration into account when discussing how the verb of wonder formed on the pattern نازم المعنى المعنى

و عددى أن أسهل منه مأخذا أن يقال إنه أمر لكل أهد بأن يجمل زيدا كريما ، أى بأن يصفه بالكرم.

Ibn Ya'ish, however, criticises this view on several counts and this includes the consideration that a sentence containing a verb of wonder forms a proposition: 45

إنه وإن كان بلفظ الأمر فليس مأمر وإنما عو خبر كتر وإنها للصدق والكذب فيصح أن يقال في هوابه: صدقت أوكذبت لأنه في صنى : همن زيد هذًا.

One of the rules which the grammarians lay down in dealing with the verb of wonder is that it can only be formed from verbs with a simple triliteral root and, accordingly, cannot be formed from quadriliterals and from augmented forms of the verb. Although both al-Mubarrad and Ibn al-Sarraj set down this rule, nevertheless, they have to account for expressions like ما أولاه ما أولاه للراهم بالصورة , where the verb of wonder is based on the sense of fourth form verbs. Sibawayh does mention briefly that the verb of wonder can be formed from the fourth form of the verb but he does not elaborate on this. 46 Later scholars, however, did not accept that the verb of wonder could be formed at will from the fourth form of the verb. 47

Neither al-Mubarrad nor Ibn al-Sarraj satisfactorily account for expressions like the two quoted above. Ibn al-Sarraj, for instance, writes: 48

فإن قال قائل : فقد قالوا : ما أعطاه - وهو أعطى يصطى - وما أولاه مالخير ، قبل هذا على هذف الروائد لأن الأمل عطا يعطو اذا تناول ، وأعطى غيره إذا ناوله . وكذللا ولى وأولى غيره .

Al-Mubarrad expands on this approach but his whole argument based on analogy is rather weak and not really to the point: 49

فإن قيل: فقد قلت: ما أعطاه للدراهم، وأولاه بالمعروف، وإنما هو من أعطى وأولى، فهذا — وإن كان قد خرج إلى الأربعة - فإنها أصله الثلاثة والهمزة في أوله زائدة. وعلى هذا جاء: وَأَرْسَلْنَا الرّبِيَاحَ لَوَاقِحَ ، ولو كان على لفظه لكان ملاقح ، لانه يقال: ألقيت فهى ملقة ولكنه على هذف الزوائد.

يَخْرُجْنَ مِنْ أُجْوَازِ لَيْلِ غَاضِى ﴿ وَإِنْهَا هُو مُنْفَى ِ وَاسْتَمْعَلَ بِحَدْفَ زَيَادَتَهُ. وَمَثْلَ ذُلِكَ قَوْلُهُ:

ذلك قوله:

تَكْشِفُ عَنْ جَهَّاتِهِ مُلُّو الدّالُ

يريد المعلى. ومن ذلك هذفك جميع الزوائد إذا
احتجت إلى هذفها في تصغير أوجه أو اضطر إليه
الشاعر , كما قال العجاج :

وَمَهْمَةٌ مَا لِأَوْ مَنْ تَحَرَّجُا إنها حومهللا في بعض الأقاويل.

Here al-Mubarrad bases his argument on unsatisfactory analogies which are drawn from what are really debatable points of lexicography. <sup>50</sup>

Both al-Mubarrad and Ibn al-Sarraj attempt to explain anomalous formation of the verb of wonder but they offer no consistent explanation of this. Ibn Ya'ish on the other hand puts forward a rather simple and obvious explanation: <sup>51</sup>

وقد قالوا على أعطاه الدرهم وأولاه للخير . فهذا و نحوه مفصور على السماع عند سيبويه ، لا يجيز

سنه إلا ما تكلمت به العرب. فالتعبب من فعل فياس مطرد ومن أفعل مسموع لا مجاوز ما ورد عن العرب.

Although Ibn Ya'ish cites Sibawayh in support of his view, it seems clear that the latter does permit the verb of wonder to be formed regularly from the fourth form of the verb and al-Astarabadhi states specifically that this is the view of Sibawayh. 52

Although al-Mubarrad, Ibn al-Sarraj, and Ibn Ya'ish admit the possibility of forming the verb of wonder from the fourth form of the verb, a later grammarian like Abū Ḥayyan also lists examples of the verb of wonder formed anomalously from augmented forms of the verb other than the fourth: 53

وأما إن كان تلاتيا مريدا ... إن كان على ورن غير ورن أفعل فلا يجوز أن يصاغ منه أفعل ولا أفعل. ورن أفعل فلا يجوز أن يصاغ منه أفعل ولا أفعل ورند من ذلك ما أغناه وما أفقره وما أتقاه وما أقومه وما أمكنه وما أملاه وما آمله وما أشده وما أهوله وما أهماه وما أرهمه أهوله وما أهماه وما أرهمه من استفنى وافتقر واتقى واستقام وتمكن وامتلأ وتأبل واشتر واهتال واختصر واشتهى واستيا

The appearance of such usages in later works may be due to the fact that further study of early poetry had revealed them, or they may represent usages current in the language but which were so anomalous in the eyes of earlier grammarians that they ignored them.

From time to time in works of Arabic grammar there are discussions of points which are in a broad sense theological in nature and in

al-Mubarrad's chapter on the verb of wonder certain such points are discussed in detail. An examination of his treatment of such questions provides excellent examples of his rather discursive style and his tendency to digress. The standard Basran explanation of the verb of wonder is capable of producing some theological argument and al-Mubarrad deals with this point: 54

فإن قال قائل : أرأيت قولك: ما أحمن زيدا اليس في التقدير والإعمال — لا في التعبب — عنزلة قولك: شي حسن زيدا . فكيف تقول هذا في قولك: ما أعظم الله يا فتي ، وما أكبر الله ؟ قيل له : التقدير على ما وصفت لك والحين : شيء عظم الله يا فتي ، وذلك الشيء الناس الذين يصفونه بالعظمة ، كقولك : كترت الشيء الناس الذين يصفونه بالعظمة ، كقولك : كترت كبيرا ، وعظمت عظيما . فإن قال قائل : فينتهب هذا من حيث انتهب زيد ، قيل له : لا شيء من الأفعال يستصب على معنى الدخر بأكثر من من الأفعال يستصب على معنى الدخر بأكثر من الفاعل والمفعول به . ألا ترى أخك تقول : شتمت زيدا ، وأكرمت عمرا ، فالفعل الناصب هنس واهد والمعنى صفاله عن صفاله الناصب هنس واهد

وليس شيء يحبر به عن الله - عزّ وجلّ - إلّا على خلاف ما تخبر به عن غيره في المعنى، وجنس الفول واحد في الإعمال. ضمن ذلك ما أذكره لك ليدلّ على سائره إن شاء الله. وهو نحو تحولك: رحم الله الناس، ورحم زيد عمرا، فالرحمة من زيد رقّة وتحيّن ، والله يجبل عنها. وكذلك علم الله ، وهو عالم بنفسه . وتقول: علم زيد علما ، وإنها ذلك علم حمول علم الله علم حمول المناس ، وتقول: علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول: علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول: علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول: علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول: علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول: علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول : علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول : علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول : علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول : علم زيد علما ، وإنها ذلك علم حمول بنفسه . وتقول : علم ويد علما ، وإنها ذلك علم حمول بنفسه . وتقول : علم ويد علما ، وإنها ذلك علم حمول بنفسه . وتقول : علم ويد علما ، وإنها ذلك علم ويد علما ، وإنها ذلك علم حمول بنفسه . وتقول : علم ويد علما ، وإنها ذلك علم ويد علما ، وإنها ذلك علم حمول بنفسه . وتقول : علم ويد علما ، وإنها ذلك علم ويد علما ، وإنها ذلك علم ويد علما ، ويد عل

قيه و أدب اكتسبه . وكذلك جميع ما تخبر به .
و إذا كان زيد مفعولا قلت : لقيت زيدا ,
و دأيت عسرا . و تقول : ذكرت الله فإنا تعنى أن
ذكرك كان لهذا الاسم ، وكذلك دعوت الله .
محفارج الدُّفعال واهدة في الإعمال والمعانى محتارج الدُّفعال واهدة في الإعمال والمعانى التقدير فيها ذكرت لكو .

Although al-Mubarrad does not say so the point which gives rise to the above discussion is in fact one of the Kufan objections to the Basran explanation of the verb of wonder. 55

Another point which raises difficulties of a theological nature, and which al-Mubarrad discusses, is the apparent use in the Qur'an of the verb of wonder with reference to the state of mind of God himself. When discussing the verb of wonder of the pattern which is al-Mubarrad writes: 56

وص هذا الماب قول الله عرّ وجلّ ؛ أُصْبَعْ بِهِمْ وَأُسُمِرُ ولا يقال لله حرج على ولا يقال لله حرج على كلام العباد , أى هؤلا، صن يحب أن يقال لهم : ما أسمهم وأبصرهم في ذلك الوقت. وصنل ذلك قوله : فَقُولًا لَهُ قَوْلًا لَيْنَا لَعَلَّهُ مَنَا لَعَلَّهُ مَنَا لَا لَهُ عَوْلًا لَهُ قَوْلًا لَيْنَا لَعَلَّهُ مَنَا لَا لَهُ عَوْلًا لَهُ قَوْلًا لَيْنَا لَعَلَّهُ مَنَا لَعَلَّهُ مَنَا لَا لَهُ وَلَا لَهُ عَوْلًا لَهُ قَوْلًا لَهُ قَوْلًا لَهُ الله ولا يقال مَنْ ولك إنها هي للترجي ولا يقال دلك الله ولكن المعنى - والله أعلم - اذهبا على رهائكما وقولا العول الدي ترجوان به ويرجو به المخلوقون شفر منا طالبوه .

وأما قوله: فَمَا أَصْبَرَهُمْ عَلَى ٱلنَّارِ، عليس

من هذا ونكمه - والله أعلم - التقرير والتوسيخ. وتقديره: أى شيء أصبرهم على النار، أى دعاهم السيما واضطرهم إليها، كما تقول: صبرت ريدا على القتل، ونهى رسول الله صلى الله عليه وسلم أن بصبر الروح، ومثل ذلك قوله:

فُلْتُ لَهُ: أَصِرْهَا دَائِنًا أَمْنَالُ بِسَطَامٍ بَنِ فَيْسٍ قَلِيلٌ

فهدا محاره , ولا يقال لله عز وجل لأده إنها يعجب من برم عليه ما لا يعلمه ولا يقدره فيستعب كيف وقع مثله وعلّام الفيوب يجلّ عن هذا.

#### Notes to Chapter VI

- 1 Sibawayh, vol. i, pp. 104-8, 298-300.
- 2 Al-Farra, vol. i, pp. 79, 225-6.
- 3 Khalaf al-Ahmar, Mugaddimah fi l-nahw (Damascus, 1961), p. 58. For Khalaf al-Ahmar see Kahhālah, vol. iv, p. 104.
- 4 See Ibn Ya'ish, vol. ii, p. 70.
- 5 See art. cited p. 78, n. 43 above.
- 6 See pp. 98-9 above.
- Ibn al-Sarraj, vol. i, pp. 268-76, 375-82; see also pp. 189, 374-5.
- 8 See pp. 65-73 above.
- 9 Ibn Ya'ish, vol. ii, p. 70.
- 10 Al-Mubarrad, vol. iii, p. 32.
- 11 Tbid., p. 34.
- 12 Ibn al-Sarraj, vol. i, p. 269.
- 13 Ibid., pp. 274-5.
- 14 See al-Astarabadhi, vol. i, pp. 216-22.
- 15 Sibawayh, vol. i, p. 299.
- 16 Al-Mubarrad, vol. iii, p. 35; see also p. 67.
- 17 Ibn al-Sarraj, vol. 1, pp. 378-9.
- 18 Ibid., pp. 273-4, 376-7.
- 19 Ibn Ya'ish, vol. ii, p. 73. See also p. 181, n. 21 (d) below.
- 20 Abu Hayyan, p. 226. See p. 43 above.
- 21 Al-Mubarrad, vol. iii, pp. 36-7.
- 22 Ibid., p. 36, n. 2.
- 23 Ibn al-Sarraj, vol. ii, p. 238. See also Sibawayh, vol. i, p. 105.
- 24 Ibn Ya'ish, vol. ii, p. 74.
- 25 Ibn al-Sarraj, vol. i, p. 261.
- 26 Ibn al-Anbari, Insaf, pp. 351-3; Ibn Ya'ish, vol. ii, pp. 73-4.
- 27 See Abu Hayyan, pp. 228-30.
- 28 Wright, vol. i, p. 98C.

- 29 Sibawayh, vol. i, p. 37.
- 30 Abu Hayyan, p. 370; see also Ibn Ya'ish, vol. vii, p. 149.
- 31 Al-Mubarrad, vol. iv, p. 173.
- 32 Ibid.
- 33 Abu Hayyan, p. 370; Ibn Ya'ish, vol. vii, p. 149; Ibn al-Sarraj, vol. i, p. 116.
- 34 Al-Mubarrad, vol. iv, p. 177.
- 35 Tbid., p. 175; see also p. 177.
- 36 See Sibawayh, vol. i, p. 419.
- 37 Ibn al-Sarraj, vol. i, p. 116.
- 18 Ibid., p. 115. In 1. 10 of this passage الزم has been substituted for الزم in the printed text.
- 39 Al-Mubarrad, vol. iv, pp. 174-5.
- 40 Ibn Ya'ish, vol. vii, p. 142.
- 41 Ibid., p. 146.
- 42 Ibn al-Sarraj, vol. i, pp. 114-5.
- 43 See Abu Hayyan, p. 371; Ibn Ya'ish, vol. vii, p. 148.
- 44 Ibn Ya'ish, vol. vii, p. 147.
- 45 Ibid., p. 148.
- 46 Sibawayh, vol. i, p. 37.
- 47 See Ibn Ya'ish, vol. vii, pp. 146-7.
- 48 Ibn al-Sarraj, vol. i, p. 116.
- 49 Al-Mubarrad, vol. iv, pp. 178-80.
- 50 See ibid., editor's notes.
- 51 Thn Ya'ish, vol. vii, p. 144.
- 52 Sibawayh, vol. i, p. 37: vol. ii, pp. 251-2; al-Astarabadhi, vol. ii, p. 308.
- 53 Abu Hayyan, p. 347. See also Wright, vol. i, p. 99B.
- Al-Mubarrad, vol. iv, pp. 176-7. Il 8-10 of this passage are somewhat difficult to understand although the general point is quite clear.
- 55 See Ibn al-Anbari, Insaf, p. 67.
- 56 Al-Mubarrad, vol. iv, pp. 183-4.

#### CHAPTER VII

#### THE UŞUL AND THE KUFAN SCHOOL OF GRAMMAR

#### The Kufans

The <u>Usul</u> of Ibn al-Sarrāj represents the earliest extant Basran source of information on the Kufan school. Although the Kufan school was in existence by the time of Sibawayh there is no reference in the <u>Kitāb</u> to the views of Kufan scholars and the next major Basran grammar, the <u>Muqtadab</u> of al-Mubarrad, only mentions the Kufans by name once. 

However, in the <u>Usul</u> the grammatical views of the Kufans and their leading scholars, al-Kisā'i and al-Farrā', are frequently mentioned and commented upon. Although the <u>Usul</u> is the earliest Basran source of information on the Kufans, Kufan views on many questions of grammar can be obtained from an actual Kufan source, the <u>Ma'ānī l-Qur'ān</u> of al-Farrā'. 

However, this does not detract from the value of the <u>Usul</u> as an early source for Kufan grammatical thought because it provides much information that cannot be found in the <u>Ma'ānī l-Qur'ān</u>.

In his Introduction to the <u>Kitab al-insaf</u> Weil observes that the points at issue between the two schools of Basrah and Kufah are portrayed in this work as they appeared to the grammarians of Ibn al-Anbari's day. The season the study of much earlier works provides information about the differences between the Basrans and Kufans when the issues were still being debated. In fact, the <u>Usul</u> of Ibn al-Sarraj is a product of the period when the first monographs were composed which dealt with the points at issue between the schools, and to which the later works of al-'Ukbari and Ibn al-Anbari are almost certainly

indebted. <sup>4</sup> These first monographs seem to have been completely lost and for this reason the <u>Usul</u> is the only work from the same period which deals with points at issue between the schools.

However, the <u>Usul</u> is a work primarily intended for students of grammar at an early stage in their studies and for this reason it does not provide the detailed information on disputes between the Basrans and Kufans which specialised treatises would have provided. Nevertheless, it is still of value to compare the information about the views of the Kufans which is found in the <u>Usul</u> with that which is to be found in later sources.

An examination of the material about the Kufans contained in the Usul also permits a check to be made on the accuracy of later sources, although it must be said that the simple facts concerning the main points where the Kufans differed from the Basrans were too well known for that to be inaccurately recorded. The principal sources of information on the Kufan school have been the Kitab al-insaf of Ibn al-Anbari and the Sharh al-mufassal of Ibn Ya'ish but to a large extent these two works only cover the main questions on which the Kufans had their own view. However, the Usul provides information on the views of the Kufans and their leading scholars, al-Kisa'i and al-Farra', on many points which the two former works do not deal with. It is also worth mentioning in this connection that the Manhai al-salik of Abu Hayyan is another work which records the views of the Kufans on many points which are not usually mentioned in other works, but it is a very much later work than the Usul.

The attitude towards the Kufan school of scholars who lived long after its heyday is well-known from works like the <u>Kitab al-insaf</u> of Ibn al-Anbari in which criticisms of the Kufans and their ideas abound. In the <u>Iqtirah</u> al-Suyūṭī gathers a selection of views on the Kufan school which are more or less severely critical but he does begin with

a balanced appraisal of the strengths and weaknesses of the Basrans and Kufans respectively based on the consensus view of the scholars:  $^{5}$ 

النفوا على أن المصربين أصح قياسا لأمهم لا يستون إلى كل صموع ولا يقيسون على الشاذ والكوفيون أوسم رواية . قال ابن جنى الكوفيون علامون بأشعار العرب مطلعون عليها . وقال ابو حيان ... ولسنا متعبدين باتباع مذهب البصريين بل نتبع الدليل . وقال الأحداسي في شرح المفقل : بل نتبع الدليل . وقال الأحداسي في شرح المفقل : الكوفيون لوصموا جيئا واهدا فيه جوازشي عالم محلوه أصلا وبوروا عليه بخلاف البصريين . قال . وحما افتحروا البصريون على الكوفيين أن قالوا ، نحن نأخذ اللعة من هرشة الشعاب وأكلة البرابيم وأنتم تأخذ ونها من أكلة الشواريز وباعة الكواميخ .

As this passage shows the Arab grammarians were often critical of usages which the Kufans allowed and in this Ibn al-Sarraj was no exception. For Ibn al-Sarraj and for all the Basran grammarians the twin foundations on which grammatical studies were based were sama (attested usage) and givas (analogy), and with regard to these two principles Ibn al-Sarraj criticises usages which the Kufans allow.

In discussing the use of the damir al-shan in sentences of the type مُلْنَتُهُ رَبِّ عَامُ The al-Sarraj observes that the Kufans put the participle, which they call the fi'l da'im, into the accusative if it is placed adjacent to the pronouns

والكوفيون إجيرون إذا ولى هذه الهاء فعل دائم النصب، فيقولون : طنست قامًا زيد. ولا أعرف

## لدلك وجها في القياس ولا السماع من الحرب

Unfortunately, Ibn al-Sarraj does not give any reason why the Kufans permit this usage and he abruptly rejects it as inconsistent with gives and not founded upon sama and he offers no further explanation of this.

Another Kufan usage which Ibn al-Sarraj rejects for the same reasons is the use of the definite article with the hal in a construction of the type مرت بريد واقعا —a construction which the Kufans parse in a different way than the Basrans do: 7

وتقول: مدت تزيد واقفا، فتنصب واقفا على الحال، والكوفيون بحيرون نصبه على الحبر و بحملونه كنصب طبر كان و طبر طن، و بحيرون فيه إدخال الذّلف واللام .... وهذا الذى أجازوه فير معروف عندى من كلام العرب ولا موجود ي ما يدجبه القياس.

On another occasion Ibn al-Sarrāj attacks the Kufans for not being able to differentiate between the parts of speech and for not having a proper appreciation of what is common in speech and what is unusual. This criticism is made after Ibn al-Sarrāj has given a Kufan list of prepositions and it is worth quoting the list to show how unfair on occasions Basran criticisms of the Kufans could be:

واعلم أن الاسياء التي بسميها البصريون ظروفا بسميها الكسائي صفة والقراء بسميها محال ويخلطون الاسماء بالحروف فيقولون: طروف الحفض أمام وقدام وهلف وقبل وسعد وتلقاء وتجاه وهذاء وإزاء ووراء محدودات ومع وعن وفي وعلى ومن والي ... وصبتاء وسيداء وللعني ولعد محدود وسنامقصور عنزلة هذاء

ولدى. فيخلطون الحروف والأسها والشادّ بالشائع وقد تقدم تبيين الغرق بين الاصم والحرف وبين الشاذّ والمستمل.

The al-Sarraj criticises the Kufans here because the list purports to be one of particle-prepositions (huruf al-khafd) but contains many noun-prepositions (zuruf) and he implies that the Kufans do not know the difference between particle and noun. However, there is every evidence that the term harf was used in a more general sense than just to mean particle and could apply as here to a noun or even to a verb. 9

The strictest of Basran grammarians may have avoided using the term harf loosely for any part of speech but it is not a sign of incompetence as a grammarian not to do so. The second criticism which Ibn al-Sarraj levels against the Kufans on the basis of the above list is that they confuse what is rare and unusual in speech with what is normal and current. Although the list may contain some expressions of infrequent occurrence they are lexically sound and would merit inclusion in any account of Arabic prepositions which aims at completeness.

One point at issue between the Kufans and the Basrans was that the former permitted the formation of the elative from the roots <u>b-y-d</u> and <u>s-w-d</u> whereas the latter did not and treated them like all other roots denoting colour. <sup>1C</sup> To justify their view the Kufans would cite evidentiary verses but the Basrans considered that such verses constituted no authority for the usage but merely represented a poetic licence which was not to be followed. In the <u>Usul</u> Ibn al-Sarraj mentions one such verse and he relates al-Mubarrad's view both on the verse and on the sort of scholars who use such verses as grammatical evidence:

و قد أنشد بعق الناس يَا كَيْتَنِي مِشْلَكِ فِي النِّياضِ أَنْيُصُ مِنْ أُمْنْتِ بَيِي إِبَاض قال أبو العماس (المبرد): هذا معمول على فساه. وليس البيت الشاذ والكلام المحفوظ مأدنى إسناد هجة على الأمل المجموع عليه في كلام ولا محو ولا فقه وإنها يركن إلى هذا ضعفة أهل النحو ومن لا هجة معه. وتأويل هذا وما أشبهه في الإعراب كتأويل ضعفة أصحاب الحديث وأتباع القصامى في الفقه.

Although this particular criticism is not directed against the Kufans by name it is very much in the spirit of the criticisms which the Basrans made of the Kufans, and the usage in particular which leads to the general criticism of unsound scholarship is one allowed by the Kufans.

As has been shown, in the <u>Usul</u> Ibn al-Sarraj relates and criticises usages allowed by the Kufans but he does not usually provide detailed information on the issues between the Basrans and Kufans of the type which can be found in the <u>Kitab al-insaf</u> of Ibn al-Anbari. However, this is not always the case as is the case with the dispute over whether sentences of the type

whether sentences of the type

admissible. When dealing with the particle <u>inna</u> Ibn al-Sarrāj writes:

12

والفراء يجيز: إنّ هذا وزيد قاعًان ، وإنّ الذي عندك وزيد قاعًان ، وإن الذي الم إن وزيد قاعًان ، إذا كان اسم إن لا يستبين فيه الإمراب شوهذا وما ذكرناه في هذه المسائل . وعلى ذلك ينشدون هذا البيت

مَنْ يَلُ أُمْسَى بِأَسْدِينَةِ رَمْلُهُ فَإِنَّى وَقِيالٌ بِهَا لَعْرِيبُ

فيرفع قيارا وينصب، وكذلك لو قال: العربان, فإفراد الفعل وتثنيته في هذا عندهم سواء.

والكسائ يحيز الرفع في الاسم الثاني مع الظاهر والمكى. فإن نعت اسم إن اثو أكدته أو أبدلت منه فالنف عندنا على عندنا لا يجوز غيره وإما الرفع جاء عندنا على الفلط.

From this passage it emerges that al-Kisa'i from among the Kufans supports a usage of the type ان نبا وعمرو قائان.

Ibn al-Sarraj undertakes to refute the types of usage supported in this passage at a later point in the <u>Usul</u> in the course of his chapter on العطف على الموضع . After discussing the alternative constructions العطف على العامل and العامل ولاقاعد he turns to the construction وكمرو : إن زيد في الدار وحمرو

ومن ذلك: إن زيدا في الدار و عمرا ، ولو أسقطت فيه إنّ لكان : زيد في الدار و عمرو ، فإنّ مع ما عملت فيه في موضع الرفع . و بنبغي أن تعلم أنه ليس للا أن تعطف على الموضع الذي فيه حمرف عامل إلا بعد تمام الكلام من قبل أن العطف نظير التثنية والجمع ألا ترى أن معنى قوللا قام الزيدان ، إنما هو :قام زيد وزيد فلما كان العاملان مشتركين في الاسم تقول :قام زيد و عمرو ، فالوام نظير التثنية وإنما تدخل إذا لم تكن التثنية . فلما لم يكن يجوز أن يعمل في التثنية الرفع والنصب ولا الرفع والخفف ولا أن يعمل في التثني عاملان ، كذللا لم يجز في المحطوف والمعطوف عليه . فإذا تم الكلام عطفت في العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيده على العامل الأول وكنت مقترا إعادته ، وإن لا تفيد

في اللفظ لأنك مستفى عنه ، ألا ترى أنه لا يجوز أن تقول الى زيدا وعمره منطلقان ، لما هبرتك به ولأن قدلك منطلقان ، يصير خبر لمرفوع ومنصوب وهذا مستحيل ، فإذا قلت ، إن زيدا منطلق وعمره ، صلح لأن الكلام قد ق ورفعت لأن الموضع للابتدا وإن زائدة فعطفت على موضع إنّ وأعملت الابتدا وأن رائدة فعطفت على موضع إنّ وأعملت الابتدا وأضرت الخبر وهذفته اجتراء عأن الأول بدلّ وأضرت الخبر وهذفته اجتراء عأن الأول بدلّ عليه .

Although Ibn al-Sarrāj gives considerable attention to refuting Kufan usages of the type الْمَانُ وَعُمَرُو قَالَانَ , there are various other aspects to this issue but in his actual refutation Ibn al-Sarrāj is very thorough.

Another of the points at issue between the Basrans and Kufans was whether it is permissible to say the first of the first of the same of the Basrans and Kufans of the was whether it is permissible to say the first of the first of the same of the

والم الكوفيون فيجيزون: طعامك ما زيد آكلا، ينسبهونها (ما) بلم ولن، وأباه البصريون. وهجة البصريين أنهم لا يوقعون المفعول إلّا هيت يصلح لناصبه أن يقعه. فلما لم يجز أن يتقدم الفعل على ما الم يجز أن يتقدم ما عمل فيه الفعل. والفرق بين ما وبين لم ولن أن لم ولن لا يليهما إلّا الفعل فصارنا مع الفعل بمنزلة حموف الفعل.

Although Ibn al-Anbari devotes a little more space to this question

he in fact adds little to what Ibn al-Sarraj writes but rather expands his subject matter to suit his format in presenting the Basran-Kufan controversies.

It seems clear that Ibn al-Sarraj did take an interest in developing the sort of detailed arguments against Kufan views which can be found recorded in the <u>Kitab al-insaf</u>, although the <u>Usul</u> itself does not give any detail on this. One of the most prominent controversies between the two schools was whether the verb is derived from the infinitive or vica-versa. The Basrans took the former view and the Kufans the latter. In the <u>Idah</u> 'ilal al-nahw of al-Zajjājī there is a section devoted to this question and he attributes one of the arguments in favour of the Basran position to Ibn al-Sarraj: 16

دليل آخر للبصريين كان أبوبكر بن السراج يستدل به قال : لو كانت المعادر مأخوذة من الأفعال هارية عليها لوهب ألّا تختلف كما لا تختلف أسماء الفاعلين والمفعولين الجارية على الأفعال شحو مارب مصروب وشام ومشوم ومكرم ومكرم وما أشبه ذلك ما لا ينكسر ووأينا المصادر مختلفها أكثر ما جاء سنها على الفعل كقولنا شربا وشربا وشربا ومشربا ومشربا وشرابا ، وعدل عن الحق عدلا وعدولا ، وما أشبه ذللا علمنا أنها غير عدولا ، وما أشبه ذللا علمنا أنها غير عادية على الأفعال وأن الأفعال ليست بأمولها .

It is of interest by way of comparison to cite Ibn al-Anbari's treatment of the same line of argument in defence of the Basran position: 17

وسنهم من تحسك بأن قال الدليل على أن المصدر ليس مشتقا من الفعل أنه لوكان مشتقا منه لكان

يحد أن يحرى على سَنَن في القياس ولم يختلف كما لم يختلف أسماء الفاعلين والمفصولين. فلما المتلف المصدر المتلاف الاحتاس كالرجل والثوب والتراب والماء والزيت وسائر الأجناس دل على أنه غير منتق من الفعل.

In his treatment of the point Ibn al-Anbari does not mention the name of Ibn al-Sarrāj and this illustrates the tendency of later writers to obscure the individual contribution of earlier scholars to grammatical thought. It is also of interest to note how the later scholar treats the argument. With Ibn al-Sarrāj the line of argument is based purely on linguistic considerations but with Ibn al-Anbari it assumes a pseudo-philosophical veneer because he draws on logic to liken the various forms of the infinitive to the genera of the logicians.

#### The Baghdadis

As well as making reference to the famous grammatical schools of the Basrans and the Kufans, the Arab writers on grammar also refer from time to time to a group called the Baghdadis. <sup>18</sup> From the evidence available it is clear that if the expression "the Baghdadis" is not to be understood merely as an alternative name for the Kufans, it must refer to a group of scholars closely connected in outlook with the Kufan school. Evidence can be brought forward for the view that the expression "the Baghdadis" is no more than an alternative name for the Kufans and this evidence may seem conclusive, but on further examination the identification of the Baghdadis does not appear to be such a simple matter.

In the <u>Muqtadab</u> al-Mubarrad does not mention the Baghdadis at all and, indeed, a solitary reference to the Kufans is the only occasion

on which he mentions by name a party of grammarians other than the Basrans. <sup>19</sup> Ibn Qutaybah, an exact contemporary of al-Mubarrad, mentions the Baghdadis four times in the Adab al-kātib but the Baghdadis are the only group of grammarians mentioned in this work other than the Basrans and, from Ibn Qutaybah's use of the term, it must be understood as a simple alternative name for the Kufans. <sup>20</sup> Since the leading Kufan scholars were by residence intimately associated with Baghdad and its intellectual life, it is not unnatural that they should also take their name from that city.

Two of the occasions on which the Baghdadis are mentioned by Ibn Qutaybah are of particular interest:  $^{21}$ 

ورنأت فلانا ، إذا فلت فيه مرتبة . هذا قول البصريين ، الأهفس وغيره . وأما الفراء وغيره من البغداديين فيجعلونه من غلطهم مثل : هذات السويق .

and:

قال التصريون تقدير الانسان فعلان .... وقال بعض التعدادين، أصل فيه الانسيان على زنة إفعلان هُذُذ فت الياء استخفافا لكثرة ما يجرى على السنام المنتهم.

In the first passage one of the leading Kufan scholars, al-Farra', is expressly described as one of the Baghdadis which would mean that those Ibn Qutaybah calls the Baghdadis are in fact the Kufans. In the second passage the derivation of the word <u>insan</u> is discussed and certain of the Baghdadis are said to oppose the Basrans, but in the discussion of this point in the <u>Kitab al-Insaf</u> it is the Kufans who oppose the Basran view.

If the Adab al-katib is a work which supports the view that the

Kufans and the Baghdadis are one and the same group of scholars, there is information in other works which would lead to a rejection of this view. In the <u>Ugul</u> there are a number of references to the Baghdadis apart from the much more frequent references to the Kufans and it does appear from this work that some distinction is drawn between the two groups. <sup>23</sup> A discussion of a number of the occasions on which the Baghdadis are referred to provides a further opportunity to consider the treatment in the <u>Ugul</u> of the views of grammarians outside of the Basran tradition.

On one occasion when Ibn al-Sarraj mentions the Baghdadis he does identify them with the Kufan school. The grammarians do not consider sentences in which there is a use of two relative pronouns following each other to be supported by attested usage, although such sentences are constructed as an exercise. Dealing with this point Ibn al-Sarraj writes: 24

In this passage Ibn al-Sarraj mentions "the Baghdadis who follow the Kufans". How this is to be interpreted is not quite clear for either it could refer to a group of Baghdadis who follow the Kufans or it could mean that the Baghdadis in general follow the Kufans.

On another occasion in the  $\underline{\text{Usul}}$  the Kufans and the Baghdadis are mentioned side by side:

باللام يقولون: هي بمنزلة ما وإلّا وقد قال الفراء: إنها محسزلة قد وتدخل أبدا على آخر الحكلام شحو قولك إنْ زيدا لقائم، تريد: ما زيد إلا قائم. وقد قيل إن زيدا لقائم، تريد وكذلك: إنْ ضرب زيد لعصرا، وإن أكل زيد لطحاملا. وكان الكسائي يقول: هي مع الأصاء والصفات – يعني بالصعات الظروف – إن المشقّلة فيقفت ومع الأفعال محنى ما وإلّا.

The point dealt with here is the grammatical analysis of sentences containing what the grammarians call in al-mukhaffafah which is followed by lam al-farigah.

26 At the beginning of the passage the Kufans and the Baghdadis are grouped together as if they were two distinct parties of grammarians with similar views on the point under discussion. The only two scholars mentioned by name are al-Kisa'i and al-Farra' who are in fact Kufan scholars.

On certain occasions Ibn al-Sarraj does follow the Baghdadis in their views. When discussing the verb of wonder one of the questions which the grammarians turn to is the use with it of various auxiliary verbs. In the <u>Usul</u> Ibn al-Sarraj discusses admissible use of certain auxiliary verbs after the verb of wonder: <sup>27</sup>

ولا يجوز : ما أحسن ما ليس زيد ، ولا : ما أحسن ما زال زيد ، كما جاز للا ذلك في كان . و لكن يجوز : ما أحسن ما لا يزال أحسن ما ليس يذكرك زيد ، وما أحسن ما لا يزال بذكرنا زيد ، وهذا مذهب البنداديين .

Unfortunately, Ibn al-Sarraj does not give any more information on the usage which he and the Baghdadis permit. The use of kana alone from among verbs of its type after the verb of wonder is a usage which is

generally recognised by the grammarians, and on this Ibn Ya'ish writes:  $^{28}$ 

وقد قالوا، ما أحسى ما كان زيد. ترمع زيدا حنا لا غير وكان تامّة هنا وزيد عاعل. ومامع العمل مصدر والتقدير: ما أحسن كون زيد.

Another example of a Baghdadi view being accepted is seen in a discussion of the vowelling of i, where Ibn al-Sarraj quotes al-Mubarrad who gives the Baghdadi view which he accepts as being based on givas: 29

قال ابوالصاس (المسرد) - رحمه الله: والمتداريون مقولون والله أن زيدا منطلق فيفتون أن . وحد سندى قياس لأنه قسم فكأنه قال : أحلف بالله على ذاك , أشهد أنك منطلق.

On this point Abu Hayyan has some interesting information in the Manhajal-salik; in commenting on the view of Ibn Malik that after an oath , when unaccompanied by the particle la, may be vowelled both as inna and anna, he writes: 30

والمذاهب في ذلك أرسة (۱) إجارتهما واختيار الفتح وهو مذهب الكمائي والمفداديين (۲) و إجارتهما واختيار الكسر (۲) و وجوب الكسر (۲) و وجوب الفتح و حو مذهب الفراء (٤) و وجوب الكسر و حو الذي صحيحه أصحابنا و حو القيامي و به ورد السماء و حو مذهب البصريين.

It should be noted that although this passage from the Manhaj al-salik expressly deals with it unaccompanied by the particle la, the same is the case with the passage from the Usul as the example shows, even if this is not expressly stated. When it was used in

oaths and the particle la followed the vowelling inna was accepted without controversy. Although Abu Hayyan mentions the views of the two leading Kufan scholars, he does not mention the Kufans as a group but he gives the view of the Baghdadis as does the <u>Usul</u>. Ibn al-Sarraj, quoting al-Mubarrad, merely states that the Baghdadis favour the use of <u>anna</u>, but Abu Hayyan adds that they found the use of <u>inna</u> acceptable. Although al-Mubarrad favours the Baghdadi view, Ibn al-Sarraj writes earlier in the <u>Usul</u> that the vowelling <u>inna</u> is to be used at all times in oaths, and this would agree with what is the best Basran view.

Although al-Mubarrad accepts the Baghdadi view on this occasion,

The al-Sarraj does quote al-Mubarrad being highly critical of the

Baghdadi view of the nature and power of government of the exceptive

particle illa: 32

وال ابو العباس (المبرد) - رحمه الله: يرعم البغداديون أن قولهم إلا في الاستثناء إنها هي إنّ ولا ولكنهم حقّفوا إنّ تكثرة الاستعمال ويقولون: إذا قلنا: بها هادني أهد إلّا زيد فإنها رمعنا ريدا بلا وإن نصنا فيإنّ ونحي في ذلك مخترون في هذا لأنه قد احتمع عاملا، إنّ ولا ، فحن نحمل أيهما شئنا . وكذلك يقولون عاملا، إنّ ولا ، فحن نحمل أيهما شئنا . وكذلك يقولون عماء في القوم إلا زيد وإلا زيدا . ولا يحرفون ما نقول نحن أن رفعه على العصف في صفى غير . فيلزمهم أن يقولوا ؛ ما جائي إلا زيدا ، إذا أعملوا إنّ وهم لا يقولون به . حائي إلا زيدا ، إذا أعملوا إنّ وهم لا يقولون به . فيالناهم : لم ذلك ، فقالوا لأن أهد مضمرة ، قلت : فيال أهدر أن يجوز أن النصب كما يحوز إذا أظهرت في وهذا فاسد من كل وجه ذكرنا إياه يجمل له حظا فيما وهذا فاسد من كل وجه ذكرنا إياه يجمل له حظا فيما

ملتفت إلى ويحب على قولهم أن تنصب النكرات بلا ى الاستثناء ملا تنوين لأن لا تنصب النكرات بلا تنوين.

The view quoted here on the nature of <u>illa</u> and its power of government is described in later works like the <u>Kitab al-ingaf</u> as a Kufan view held by al-Farra in particular. <sup>33</sup> Al-Mubarrad's attack on this position is based on the fact that it is quite obviously inconsistent, whereas in the <u>Kitab al-ingaf</u> Ibn al-Anbari's attack is rather more formal and theoretical: <sup>34</sup>

... كل هرفيين ركب أهد مما مع الآخر فإنه يبطل حكم كل واهد منهما عمل كان عليه في حالة الإفراد ويحدث لهما بالتركيب حكم آخر ومار هذا بمنزلة الأدرية المركبة من أشياء محتلفة فانه يبطل حكم كل واهد منهما في حالة الإفراد و يحدث لهما بالتركيب حكم آخر. وهو (الفراء) لا يقول في إلا كزلال بل يرقم أن كل واهد من الحرفيين جائ على أصل و عمله بعر التركيب واهد من الحرفيين جائ على أصل و عمله بعر التركيب كما كان قبل التركيب.

### Kufan influences on the Uşul

In a previous chapter a passage has been cited from Yaqut in which he quotes al-Marzubani who considers that Ibn al-Sarraj derived the contents of the Usul from the Kitab of Sibawayh, "although he relied in it on the Masa'il of al-Akhfash and on Kufan ideas and opposed Basran principles in many matters...."

The idea that Ibn al-Sarraj is indebted in the Usul to Kufan grammatical thought is patently untrue because a study of this work leads to no other conclusion than that Ibn al-Sarraj was a scholar firmly within the Basran tradition.

However, there are indications that Ibn al-Sarraj was not uninfluenced by Kufan grammatical thought.

when demonstrative pronouns are used to begin sentences and are followed by a noun a further element may be added and treated as a hal as in المعناء ألمان , although the final element may equally correctly be put into the nominative. The al-Sarraj's discussion of sentences of this type seems on certain points to be influenced by the Kufans' approach to this topic. After explaining why it is neccessary to use an accusative of the hal alone in a sentence of the type seems in which a proper noun follows the demonstrative, Ibn al-Sarraj continues: 37

وإن كان الاسم ليس معلم ولكنه واحد ليس له تان كان الخبر أيضا منصوبا كتولك: هذا الضر منيرا وهذه الشمس طالعة وكذلك إن أردت بالاسم أن تجمله بعم الجنس كله ويكون إخبارك عن واحده كإخبارك عن جميعه ، كان الخبر منصوبا كقولك: هذا الأسد مهيبا وحذه المعقرب مخدّقة ، إذا لم ترد عقربا تراحا ولا أسدا نشير إليه من مائر الأسد.

The particular attention given to the sort of sentences discussed in this passage and the prescribing of the accusative in them seems to stem from the Kufan approach to sentences introduced by the demonstrative pronoun. The use of the accusative in the examples in the above passage was called tagrib by the Kufans. Tagrib is the use of the demonstrative pronoun with the same governing force as kana and, according to Thaclab, the use is so called because the demonstrative pronoun is made to "approximate" to the verb kana. 38 As tagrib is a purely Kufan concept and was not recognised by the Basrans, it is worth quoting a discussion of it by Abu Hayyan although he does not mention the sort of sentence

to be explained by <u>tagrib</u> which are mentioned in the above passage from the <u>Usuls</u> 39

فقال الفرا، والكسائى: يقال: هذا زيد قائمًا على أن قائمًا هنر التقريب الذى يشبه فيه هذا كان . هين يقال : كيف تخاف الظلم وهذا الخليفة قادمًا ، وكيف تحد البرد وهذه الشمس طالعة ، يقرب هذا وهذه قدوم الخليفة وطلاح الشمس . ولم يكن هذا وهذه قدوم الخليفة وطلاح الشمس . ولم يكن هذا في المعنى إشارة لأن الخليفة لا يحمل ولا يُشلا فيه فقعر في الأشارة . وكذلك الشمس قد غنيت فيه فقعر في الإشارة . وكذلك الشمس قد غنيت بشهر تهاعن الإشارة التي تحددها و تصينها .

In the passage from the Usul quoted previously the first type of sentence for which Ibn al-Sarraj prescribes the use of the accusative are ones like هذه النص طالعة and هذه النص طالعة in which the subject is a unique entity. From the standpoint of normal Basran grammatical analysis such sentences would not be mentioned for special consideration because the last element could be put into the nominative or accusative at will depending on whether it is treated as a hal or not. 40 The second type of sentence mentioned by Ibn al-Sarraj for which he prescribes the use of the accusative are those introduced by the demonstrative pronoun in which something is affirmed in one particular instance but is generally applicable to the class into which the thing it is affirmed of falls, as in the examples مدا الدارية على المدارية على مدارية العالمية والعالمية والعالمية

Ibn al-Sarraj's special treatment of the above two types of sentence seems to have its basis in the Kufan concept of tagrib as a

discussion of the demonstrative pronoun by al-Farra in the Macani 1-Quran shows: 41

واعلى أن هذا إذا كان بعده اسم فيه الألف واللام حمرى على خلاف معانى : أهدها أن ترى الاسم الذي دمد هذا كما ترى هذا فقعله هينيذ مرفوع كقولك هذا للحمار فاره عملت الحمار نعيّا لهذا إذ كانا عاضرت , ولا محوز ها منا النصب . والوره الآخر أن يكون ما بعد هذا واهدا يؤدى عن طميو جنسه فالفعل حيننذ صصوب كقولك: ما كان من السباع فير مخمّ في فهذا الأسد محوّفا. ألا ترى أنك تخبر عن الأسدكلها عالحوف. والمعنى الثالث أن مكون ما معد هذا واحدا لا نظير له فالففل حينية أيضا سنصوب. وإنها نصبت الفعل لأن هذا ليس بصفة للأسم إنما دخلت تقرسا .... وأما معنى التقريب فهذا أول لم أخبركم عنه. فلم يحدرا مدًا ص أن يرفعوا حذا مأسد وهنره سنتظر والما سنفل الأسد بمرافعة هذا نصب فعله الذي كان يرافعه كاوته ومثله: والله غفور رهيم. فإذا أدخلت عليه كان ارتفو به والخبر ممتظر يتم به الكلام فنصبته كخلوته.

أما مصبهم فعل الواهد الذي لا مظير له مثل قولك: هذه الشمس ضياء للعباد، وهذا القر نعرا، فإن القر لا نطير له خكان أيضا عن قوللا هذا مستقنيا. ألا ترى إذا قلت: طلع القر , لا عند حب الوهم إلى قائب فتعتاج أن تقول هذا لحضوره

It is clear from the above passage that Ibn al-Sarraj makes use of Kufan grammatical discussion although he does not mention the term tagrib in this connection. However, he does mention the concept of tagrib when discussing the construction to where the particle ha is followed first by a personal and then by a demonstrative pronoun, and he also gives an indication of what tagrib is: 42

وقال قوم: إن كلام الحرب أن يجعلوا هذه الأسماء المكنية حين ها وذا وبنصبون أخبارها على الحال فيقولون: ها هوذا قائماً, وها أنذا هالما، وها أحت ذا ظالماً. وهذا الوهم يسميه الكوفيون التقريب. وهو إذا كان الاسم ظاهرا هاء بعد هذا مرفوعاً, ونصبوا الخبر محرفة كان أو مكرة. فأما البصريون فلا ينصبون إلّا الحال.

Another of the concepts to which the Kufan grammarians make reference to explain certain points of grammar is the notion of garf or khilaf. This explanation is used on various occasions by al-Farra in the Ma'ani l-Qur'an and when he introduces the notion of garf for the first time he writes: 43

.. فان تلت ؛ وما الصرف ، قلت : أن تأتى بالواو معطوفة على كلام فى أوله ها د ثنة لا تسيقيم إعاد تها على ما عطف عليها , فإذا كان كذلك مهو الصرف ، كقول الشاعر :

لَا تَنْهُ عَنْ خُلُقٍ وَتَأْتِي بِثَلَهُ

كَارٌ عَلَيْكَ, إِذَا فَعَلْتَ, عَظِيمٌ

ألا ترى أنه لا يجور إعادة لا في تاتي

سناه ، ملذلك سمى صفا إذكان سعطونا ولم يستقم أن يعاد فيه الحادث الذى قبله وسئله من الدّماء التى نصبتها العرب وهى معطوفة على صرفع عولهم: لو تركت والأسد لأكلك ولو خليت ورأيك لضللت لتا لم يحسن في الناني أن تقول : لو تركت وثرك رأيلا لضللت تهيبوا أن يعطفوا هم فا لا يستقيم فيه ما هدت في الذى قبله.

For the Kufans a common explanation based on meaning could be found for certain uses of the accusative and subjunctive and they called this factor <u>sarf</u> or <u>khilaf</u>. 44 However, in such cases the Basrans did not resort to the abstract idea that it was the meaning which governed the use of the subjunctive or accusative and they produced rather more concrete explanations of what the regent is.

The al-Andari, for instance, in the <u>Kitab al-insaf</u> firmly rejects the idea of <u>sarf</u> in all its applications. 45

The al-Sarraj makes no reference to the term <u>sarf</u> when discussing the main types of usage which the Kufans explained by this concept. <sup>46</sup>

For instance, he explains the use of the subjunctive after conjunctions like <u>wa</u> and <u>fa</u> as being due to the action of an understood, but unexpressed particle <u>an</u> and this fully accords with the normal Basran explanation. <sup>47</sup>

However, in discussing conditional sentences The al-Sarraj refers to a particular use of the subjunctive and states that the Kufans call it <u>sarf</u> and he himself goes on to make specific use of the term in explaining further similar uses of the subjunctive:

و تعدل إن تم و فسر كتلا، تربد إن تحمع مع فيامك إحسانا كتك، وكذلك: إن تقم

تُحسنَ آتك، ترد: إن تقم مُحسنا، ولم ترد: ان تقم وال تحسن آتك. وهذا النصب يسميه الكوميون الصرف لانهم صرفوه على النستى إلى معنى غيره. وكدلك في الحواب تقول: إن تقم آتك وأحسى السك وإلى تقم آتك فأحسن إلىك وإدا قلت: أقوم إن تقم ، فنسفت معمل علمها فان كان من شكل الأول رفعته وان كان س شكل الناني فضه تلاثة أومه الجزم على النسق على إنْ والنصب على الصرف والرفوعلى الاستئناف فأما ما شاكل الأول فقولك بالمحمد إن تأمر بالمعروف وتؤجر لأنه من شكل تحمد. فهذا الرقع فيه لا فير وأما ما يكون للثاني فقولك: "كحمد أن تأسر بالمحروف وتنه عن المنكر. فيكون فيه ملائة أوهه , فإن نسقت بمعل يصلح للأول ففيه أربعة أوجه: الرفه من جهتين , نسقا على الأول وعلى الاستثناف, والجزم والنصب على الصرف.

In this passage two similar uses of the subjunctive are introduced, one related to a verb forming part of the protasis of a conditional sentence and the other related to a verb linking up with the apodasis. Of the first use Ibn al-Sarrāj gives two examples, إِنْ تَعْمُ وَنَّ مَا اللهُ عَلَى اللهُ وَهُمُ اللهُ عَلَى اللهُ اللهُ عَلَى اللهُ اللهُ عَلَى اللهُ اللهُ عَلَى اللهُ اللهُ

to be conveyed. However, this usage can be related to another which is mentioned in Arabic grammars.

In his discussion of the subjunctive Wright states that it is employed with the conjunction we "when the governed verb expresses an act subordinate to, but simultaneous with, the act expressed by the previous clause." 49 Among the examples given are some which have a strong resemblance to the present usage:

and s

and describes as <u>sarf</u> occurs in sentences like and describes as <u>sarf</u> occurs in sentences like and describes as <u>sarf</u> occurs in sentences like and the subjunctive because when the verb in the protasis of a conditional sentence is in the jussive and there follows another verb connected to it by <u>fa</u> or <u>wa</u>, this second verb can be put into the subjunctive instead of the jussive. The al-Sarraj's treatment of this particular topic was of sufficient interest to al-Astarabadhi for him to quote the passage with certain slight changes when discussing in the <u>Sharh</u> al-kafiyah the possible moods of a verb joined by a conjunction to a conditional sentence.

Although Ibn al-Sarraj adopts the term <u>sarf</u> it would appear that he does so only because it is a convenient term and he does not put forward the Kufan view that <u>sarf</u> is a concept which explains uses of the subjunctive. In the passage quoted from the <u>Usul</u> it may well be

that Ibn al-Sarraj adopts the term <u>sarf</u> because in this instance he is particularly indebted to a Kufan source.

#### Notes to Chapter VII

- 1 Al-Mubarrad, vol. ii, p. 155.
- 2 See p. 26, n. 71 above.
- 3 Weil, p. 48.
- 4 Al-'Ukbari, Masa'il al-khilaf, Aleppo, n.d. The present writer has been unable to obtain a copy of this work for consultation.
- 5 Al-Suyuti, <u>Igtirah</u>, p. 84. In 1. 12 of this passage الشوارير has been substituted for the incomprehensible البتوا of the original. For this change see A. M. Salman, <u>Al-Suyuti al-nahwi</u> (Baghdad, 1976), p. 238.
- 6 Ibn al-Sarraj, vol. i, p. 219.
- 7 Ibid., p. 267.
- 8 Ibid., pp. 246-7.
- 9 See Khalaf al-Ahmar, pp. 41-2, 65, 67-8; al-Zajjaji, Jumal, p. 53.
- 10 Ibn Ya'ish, vol. vi, pp. 93-4: vol. vii, pp. 146-7; Abū Ḥayyan, p. 376.
- 11 Ibn al-Sarrāj, vol. i, pp. 122-3. Passage quoted by al-Suyūtī, Iqtirāh, p. 29.
- 12 Ibn al-Sarraj, vol. i, pp. 312-3.
- 13 Ibid., vol. ii, pp. 64-5.
- 14 See also Tbn al-Anbari, <u>Insaf</u>, pp. 85-7; al-Astarabadhi, vol. ii, pp. 354-5; Tbn Ya'ish, vol. viii, pp. 66-70; al-Farra', vol. i, pp. 310-1.
- 15 Ibn al-Anbari, <u>Insaf</u>, pp. 79-80; Weil, pp. 136-7; Ibn al-Sarraj, vol. ii, p. 244.
- 16 Al-Zajjaji, Idah, p. 59.
- 17 Ibn al-Anbari, Insaf, p. 104.
- 18 See A. I. Shalabi, Abu 'Ali al-Farisi' (Cairo, 1958), pp. 445-7.
- 19 See n. 1 above.
- 20 Ibn Qutaybah, Adab al-katib (Leiden, 1900), pp. 299, 390, 514, 637.
- 21 Ibid., pp. 390, 637.
- 22 Ibn al-Anbari, Insaf, pp. 341-2.
- 23 Ibn al⇒Sarrāj, vol. i, pp. 126, 313, 316, 339, 352, 367, 370, 396: vol. ii, p. 374.

- 24 Tbid., vol. ii, p. 374. Passage quoted by al-Baghdadi, vol. ii, p. 530.
- 25 Ibn al-Sarraj, vol. i, p. 316.
- 26 See Wright, vol. i, p. 283B: vol. ii, p. 81C.
- 27 Ibn al-Sarrāj, vol. i, pp. 126-7. In 1. 1 of this passage ند has been substituted for زيد in the printed text.
- 28 Ibn Ya ish, vol. vii, p. 150.
- 29 Ibn al-Sarrāj, vol. i, p. 339.
- 30 Abu Hayyan, p. 75.
- J1 Ibn al-Sarraj, vol. i, p. 318. See also al-Zajjaji, Jumal, pp. 70-18 Wright, vol. ii, p. 175; Howell, pt. iii, pp. 392, 398.
- 32 Ibn al-Sarraj, vol. i, pp. 367-8.
- 33 Ibn al-Anbari, Insaf, pp. 118-22; Ibn Ya'ish, vol. ii, pp. 76-7.
- 34 Ibn al-Anbari, Insaf, p. 121.
- 35 See p. 28 above.
- 36 Sibawayh, vol. i, pp. 256-61; Ibn Ya'ish, vol. ii, pp. 56-9; Wright, vol. ii, p. 278A.
- 37 Ibn al-Sarraj, vol. i, p. 180.
- Tha lab, Majalis Tha lab, 2 vols. (Cairo, 1948-9), vol. ii, pp. 227-8.
- 39 Abu Hayyan, pp. 198-9. See also ibid., p. 53; al-Suyuti, Ham<sup>c</sup> al-hawami<sup>c</sup>, 2 vols. (Beirut, n.d.), vol. ii, p. 113; Mahdi Makhzumi, Madrasat Kufah (Cairo, 1958), pp. 320-1.
- 40 See Sibawayh, vol. i, pp. 258-60; Ibn Ya'ish, vol. ii, p. 58.
- Al-Farra', vol. i, p. 12. In this passage the word fi'l appears in the sense of the Kufan expression fi'l da'im which is their term for the participle. In this passage al-Farra' also refers to the Kufan view that the <u>mubtada</u>' puts the <u>khabar</u> into the nominative and <u>vice-versa</u>.
- 42 Ibn al-Sarraj, vol. i, p. 181. Al-Farra' does not mention this use of tagrib in the passage quoted above but Tha lab does mention it in his Majalis, vol. i, p. 54.
- 43 Al-Farra, vol. i, pp. 33-4. See also ibid. pp. 276, 292.
- 44 See M. G. Carter, "Sarf et hilaf, contribution à l'histoire de la grammaire arabe", Arabica 20 (1973), pp. 292-304.
- 45 Ibn al-Anbari, <u>Insaf</u>, pp. 108-12, 229-32.

- 46 Ibn al-Sarrāj, vol. i, pp. 253-7: vol. ii, pp. 159-61.
- 47 Ibid., vol. ii, p. 159.
- 48 Ibid., pp. 197-8.
- 49 Wright, vol. ii, p. 32.
- 50 Tbid., pp. 40-1.
- 51 Al-Astarabadhi, vol. ii, p. 261.

#### CHAPTER VIII

## IBN YACISH AND THE USUL OF IBN AL-SARRAJ

The <u>Mufassal</u> of al-Zamakhsharī like so many of the concise works of the Arab grammarians calls for a commentary to assist those using it so as to clear up any problems which the very terseness of the work may create and also to provide valuable and often neccessary additional information on topics which are only briefly mentioned. In the case of the <u>Mufassal</u> this need was met by many scholars who undertook to write commentaries on it, but the one which has found most favour is that of Abū l-Baqā' Ibn Ya'īsh (1158-1245). Although its publication both in the Middle East and in Europe have led to its wide use in recent times, its long-standing popularity in the mediaeval Islamic world is clear from the constant references to it in the pages of the <u>Ashbāh wa-l-nazā'ir</u> of al-Suyūṭī and of the <u>Khizānat al-adab</u> of 'Abd al-Qādir al-Baghdādī. 2

Although Ibn Ya'ish's commentary is such a famous work it is not one whose quality is indisputable and many who use it would agree with J. W. Fück's judgment on its author that "his style is verbose and sometimes slovenly." <sup>3</sup> If the quality of this work has been called into question, its claims to originality have also been challenged. When comparing the Sharh al-kafiyah of al-Astarabadhi with the Sharh al-mufassal of Ibn Ya'ish, H. Fleisch writes: "Il est plus difficile a comprendre qu'Ibn Ya'is. Mais quand on connaîtra les sources de celui-ci, il est probable qu'il apparaîtra comme un diligent copieur, peu original." <sup>4</sup> As a concrete example of this Fleisch refers the reader to a study by G. Troupeau of al-Sīrafi's commentary on the

chapter of Sibawayh's <u>Kitab</u> dealing with phonetics. In giving the results of his study of this material Troupeau writes: "Ces renseignements nous étaient parvenus, en partie, dans le commentaire tardif d'Ibn Ya'is qui les avait repris à son compte, sans mentionner sa source:...."

If the <u>Usul</u> of Ibn al-Sarraj is studied together with the <u>Sharh</u> al-mufassal of Ibn Ya'ish, it becomes clear that the former work is one of the sources used in writing the latter because Ibn Ya'ish incorporates into his own work a number of passages from the <u>Usul</u> exactly as they stand and does not acknowledge the fact. The discussion in this chapter of some of the parallel passages not only serves the purpose of establishing exactly how Ibn Ya'ish used the <u>Usul</u> but it also gives an opportunity to examine parts of it which a later grammarian found to be of interest and which are often good examples of Ibn al-Sarraj's thought.

In this study of Ibn Ya'ish's use of the <u>Usul</u> the parallel passages have been presented side by side with, on occasions, additional material preceding or following to make their respective contexts clear, but this extra material has been separated from the adjacent column by a double vertical line. The break between any such additional material and the parallel passage which it precedes or follows has been shown by sets of dots. Where the parallel passages diverge slightly they have both been underlined with a broken line but where the divergence is more substantial they have been underlined with a continuous line. If either of the texts omits material contained in the other the omission has been shown by square-bracketing the consequent gap in the other text, and the additional material has been underlined in accordance with the principle just mentioned. Errors or omissions affecting the sense in the printed text of the <u>Usul</u> have been corrected

from the relevant passage in the Sharb al-mufaggal. Such corrections or additions have been enclosed in arrow-shaped brackets and are explained in the notes.

One topic where Ibn Ya'ish has made particular use of the Usul is the hal. Ibn al-Sarraj begins his discussion of the hal by stating that the hal together with the tamyiz constitute the class of mushabbah bi-l-maf'ul whose regent is a true verb. 6 He then explains why the hal is put into the accusative and this consists of showing how the general theory of accusative usages which he has explained earlier relates to the hal. Ibn Ya'ish works this explanatory passage into his treatment of the question why the hal is not a true maf'ul but

only resembles its 7

ولوكات الحال مفعولة الفاها الذي يسمونه الحال لحار أن تكون حرف ونكرة ال ضعد غولك عدد الله راكبال مفعولة وإذ قد تبت أنها الله عاء والممنى عاء عبد الله في ليست معمولة فهى تشبه العده الحال، و راكب مستقب ا لشبه مالمفعول

كسائر المخمولين. فلما لفتست ال وقام أهولا منشبا، وجلس بالنكرة دل على أنهاليمت الكرمتكيًا . فعيد الله مرتمع المفدول ...

تمام الكلام واستفناء الففل تمام الكلام حواستفناء الففل إذا قلت : قمت ، فلا يد إذا قلت : قمت ، فلا يد

... من هيٺ أنها نجي بعد ... لأنه هي به بعد بفاعله وأن في الفعل دليلا مفاعله > وأن في الفعل دليلا علیما که کان فیه دلیل علیه که کان فیه دلیل على المفعول. ألا ترى أمك على المفعول ألا ترى أمك [ ] أن تكون قد قيت من أن تكون قد قمت

على هال من أهوال المعل. راكبال ] ، خرب عبد الله رجلا ....

في هال من الأهوال [ فأشبه قوللو: ها ديدالله فأشبه []: ها عبدالله راكبا ، <u>ق</u>َولكِ: خرب عبدالله

ا عنمترما بإطال كيف على المنمل والفاعل. تقول : كمف جاء عبد الله ، فيكون الجواب: داكما.

مثله

Immediately after this passage from the Usul Ibn al-Sarraj explains thy the hal is so called and Ibn Ya'ish makes use of this passage at the beginning of his commentary on al-Zamakhshari's treatment of the <u>hal</u>. Preceding this Ibn Ya'ish gives his own explanation of what the hal is: 8

اعلم أن الحال وصف عبيّة الفاعل أو المفمول عذلك نحو: جاء زيد راكبا ، وأقبل حد صرعا ، وضربت عبد الله ماكيا ، و لقيت الأسير عادلا. والممنى: ها عبد الله في عند الحال ولمّيت الأميم في حذه الحال. واعتباره بأن يقو في حواب كيم . فإذا قلت : أقبل مد الله ضاحكا ، فكأن ساملا سأل: كيف أقبل ، فقلت : أقبل ضامكا كما يقو المفعول له في جواب لم فعلت...

To this explanation Ibn Ya'ish adds the passage from the Usul: 9 ... وإنما صى عالا لأنه لا وإنما صميت الحال لأنه لا

(Shark) بحور أن مكون اسم الفاعل محرر أن مكون اسم العامل ( الدل ) تطاول الوقت أم قصر ولا تطاول الوقت أو قصر ولا وانقطو ولا لها لم بأت من وانقطو ولا لها لم بات س الأفعال [ ] إذ الحال الأفعال ويبتدأ بها. والحال ذلا الفيل الحسر مه عنه.

العور أن يكون لما مفى المحور أن يكون لما مفى إنها مي حيثة الفاعل أو إنها هي حيثة الفاعل أو الممول وصفته في وقت المفعول أوصفته في وقت ذلك الفط

After this explanation of the term hal Ibn al-Sarraj goes on to define what sort of description the hal may provide and he states that it may not be an innate quality but only a transient one. Ibn Ya'ish incorporates this point into his commentary on al-Zamakhshari's remark that the hal has an affinity to the zarf and he himself states more precisely that it is particularly related to the zarf of time. 10 In taking over this passage Ibn Ya'ish compresses the examples given in it by making the various hals apply to one subject whereas Ibn al-Sarraj has three different subjects: 11

طعمه الشبه بظرف الشبه بظرف المشه المسمة المسمة المسمة الرسان لأن الحال لا تبقى المسمة المسمة المسمة المسمة المسمة منسمة منسمة المسمة المسمة منسمة المسمة المسمة منسمة المسمة منسمة المسمة منسمة المسمة ببقى ويخلفه غيره ولذلا

... لا يجور أن خكون الحال ... ملا يجوز أن خكون [ طلقةً ملا مجوز ]: خلقة. لا مجوز أن تقول: جه في زيد أحسر ,و لا [ ] جه في زيد أحمر ، ولا أحوك

﴿ أُهُولَ > ، ولا هاءني عمره أهول وولا آ طويلا ، فإذا قلت ستحاولا طويلا . فإن قلت ستحاولا أو ستطاولا عار لأن ذلك أو ستطاولا ، جاز لأن ذلك شىء يفعله وليس عليه ... ننى يفعله وليس علقة.

## ... صحم انتقاله

Ibn al-Sarraj next turns to the question of the indefiniteness of the hal and he explains that the hal must be indefinite because it simply serves to convey extra information, whereas if the definite article were prefixed to it, it would become an epithet differentiating the noun it qualifies from something else. Ibn al-Sarraj then goes on to specify in detail the difference between the hal and the adjective and Ibn Ya'ish incorporates a considerable part of this discussion into his own treatment of this point. This occurs in the broader context of his discussion of the verb as the regent governing the accusative of the <u>hal</u>: 12

( <u>nin</u> ) مرتفع بانه قاعل وصاهلاهال الوانها تفيد الماكل والمحدث منه. العامل فيهما الفعل الفيرما يعرف فإن أدهلت الحال صفة من جهة المعنى الدسم المعرفة وفرقا بينه وبين غره. والفيق بين الحال والصفة ...

ما كال إلَّا فكرة العامل إدا كان عملا ال ولا عكون الحال إلَّا فكرة الهذكور الذي هو جاء لأن | الدُّلف واللام صارت صفة ولذلك اشترط فيها ما يشترط في الصفات من الاشتقاق تحوضارب ومضروب وشيههما. فكما أن الصفة يعمل فعمل

عامل الموصوف فكذلا الحال يعمل فيه العامل في صاحب الحال إلَّا أَنْ عمله في الحال على سبيل الفضلة لأنها حارية صرى المفعل وصله نى الصفة على صبيل الحاجة اليها , إذ كانت صنية للموصوف فجسرت جرى هرف التعريف. وهذا أهد النوق بين الصفة والحال وذلك...

... أن الصفة تُفرَّق بين اسمين مشتركين في اللفظ والحال زبادة في الفائدة والخبروان لم مكن الاسم مشاركا في لفظه. ألا ترى أَدْكِ إِذَا قَلْتَ: صررت مزيد القائح ، فأخت لا تقول ذرك إلا وى الماس رجل آخر اسمه زيد وهوغيرقائح. ففصلت بالقائم بينه وبين ففصلت بالقام بينه وبين من له هذا الاسم وليس الفرزدي فيره ، [

... د أن الصفة > مفرق بين الممين مستركين في اللفظ والحال زبادة في الفائدة والخبر وإن له يكن للاسم مشاركٌ في لفظه. ألا ترى أنك إذا قلت: مرت بزيد الفّائح , فأخت لا تقول ذلك إلا وفي الماس رجل آخر اسه زيد وهو غير قائم. من له هذا الاصم وليص بقائم. وتقول: صررت بالفرزدق مقائم. وتقول: صررت بالفرزدق قائما، وإن لم يكن أهد اسمه قاعًا، وإن لم يكن أهد اسمه الفرزدى غيره، فقولك قائما

فضممت [] إلى الإضار إنما ضممت به إلى الإضار بالمرور خبرا آخر متصلابه بالمرور خبرا آخر متصلا به مفیدا.

به سفیدا ...

... فيهنا فرق ما بين الصفة الوالحال وهو أن الصفة لا فيه لمسين أو لمعان والحال قد يكون للاسم المشترك والاسم المفرد.

... إلَّا أن الخبر بالسرور على سبيل اللزوم لأنه به انمندت الجملة والإخبار | تكون الله لامم منترك بالقيام زبارة يجبوز الاستغناء منه.

After completing his discussion of this question Ibn al-Sarraj turns to the point that the hal may refer both to the subject and the object of a sentence. Here Ibn Ya'ish again draws on the wording of the Usul when he criticises what he considers to be a weakness of expression in al-Zamakhshari's treatment of the point: 13

( Usul)

ملك) «والحال مكون بيانا لهيئة العالم أن الحال محوز أن ( الفاعل أو المفعول» فتقول: الكون من المفعول كما تكون ( الفاعل الماعل ا ها زيد قامًا ، فتكون ميانا ال من الفاعل . تقول: ضربت لهيئه الفاعل الذي هوزيد. النيا قاعًا , فتجمل قاعًا لزيد وتقول: ضربت زيدا قامًا، ال ويجوز أن تكون الحال ص فتكون بيانا لهيئة المفعل الناء في ضربت إلَّا أنك ... و قعله لا تجمله مالا من أيهما نشت الله يعنى أفله إذا قلت ضربت زيرا قامًا , إن شئت جملته

هالا من المفعول الذي هو ذيد. وهذا فيه تصحّ وذلك أنك إذا جملت الحال من الناء وجب أن تلاصقه فتقول: فربت قا يُا زما ....

وكان اطلاقه فاسدا

... فإذا أزلت الحال عن ... إذا أزلت الحال عن صاهبه فلم تلاصقه الم صاهبه فلم تلاصقه لم يجز ذلك لهافيه من اللبس بجز ذلك [ إِنَّا أَنْ يَكُونُ السَّامِعِ يَعْلَمُ ۚ إِلَّا أَنْ يَكُونُ السَّامِ يَعْلَمُهُ كما تعلمه [ ] . فإن كما تعلمه أنت . فإن كان غير معلوم لم يجز .... كان غير معلوم لم يجز 11

One of the standard questions which the grammarians discuss in رأيت زيدا مصمدا dealing with the hal is the ambiguous sentence . Ibn Ya'ish's treatment of this point consists of little more than a repetition of the relevant passage from the Usul: 14

(U.il) وس كلام العرب: رأيسا وأما قولهم: رأيت observation of-mufacul) زردا مصمرا سعدرا. ورأيت زيدا ساشيا راكما درأيت زيده حاشيا راكبا - إذا كان أهدها مصميا -إذا كان أهدها ماشا والآمر سعدرا وأهدها مانيا والآهر راكيا وأهدها مصعدا والآخر سنجدرا. تعنى أنلا والآخر راكبل فالهما د إذا قلت : رأيت زيد مصمدا

المصد وزيد المحدر، المصمد وزيد المحدر، المصمد وزيد المحدر، المحدر، فيكون مصمدا حالا للقاء فيكون مصمدا حالا للقاء وسعدرا حالا لريد. وكيف وسعدرا حالا لريد. وكيف قدرت بعد أن يعلم المحاطية قدرت بعد أن يعلم المحاطية قدرت معد أن يعلم السامع المحدر، فإنه من المحدر، فإنه للمحدر، فإنه من المحدر، فإنه للمحدر، فإنه المحدر، في المحدر، في

Although Ibn Ya'ish's debt to the <u>Usul</u> is particularly noticeable in his treatment of the <u>hāl</u>, other borrowings of material can be noted as, for instance, in his discussion of exceptive sentences. Ibn al-Sarraj starts his chapter on exceptive sentences by discussing the use of the accusative case in positive exceptive sentences when it follows the particle <u>illā</u>. <sup>15</sup> After this he attempts to define what <u>illā</u> resembles in its function and when Ibn Ya'īsh explains the nature of exception he incorporates this material from the <u>Usul</u>. First of all Ibn Ya'īsh writes:

اعلم أن الاستثناء استعفال من ثناه عن الأمر يتنيه إذا صرفه عنه فالاستثناء صرف اللفظ عن عصمه بإهراج المستثنى من أن يتناوله الأول. ومقيقة تخصيص صفة عامة. على استثناء تخصيص وليس كل تخصيص استثناء فإذا قلت قام القوم وليس كل تخصيص استثناء فإذا قلت قام القوم إلّا زيدا , تبيّن بقوللا : إلّا زيدا , أنه لم يكن داخلا تحد الصدر . إما ذكرت الكل وأدن تربد بعض مدلوله مجازا . وهذا معنى قول النحويين : الاستثناء إهراج بعض من كل , أى إحراجه من أن يتناول

On to this passage Ibn Ya'ish grafts the extract from the Usul: 17

(Usil) وإتد تخرج النانى حا

دخل في الأول. فهي تشبه هرف النفي إذ ا فلت: قام القوم إلَّا زيدا ، فالهمي: ظم القوم لا زمر إلَّا أن الفرق مين الاستثناء والعطف أن الاستثناء لا يكون إلَّا بعضًا من كل و المصطوف مكون

غير الأول. ويجوز أيضا في المصطوف أن تعطف على واهد نحد قولك :قام ربد لا عمرو ولا إجوز أن تقمل في الاستثناء: قام زيد إلّا محرا ....

ا...ولا يكون المستثنى من أشياء ولا إنها تأتى لتننى عن الناني ما وجب للأول ، و إلَّا تَعْرِجِ النَّاني ما دطل فيه الأول موهبا كان أو منفيا, سعناها

(Shorh) ... فا إِنَّا تَخْرِجِ النَّانِي كَا ( اودیه اس او دخل فی الأول فلی شبه مرف النفي . فقولنا : قام القوم إلَّا زيدا , بمنزلة: قام القوم لا زيم الد أن الفرق بين الاستثناء والمطف أن الاستناء

لا بكون إلَّا بعضا من كل والمعطوف يكون غير الأول. و يحوزل

] أن يعطف على واحد نحوقولك: قام رْسد لا عمرو ، ولا يجوز في الاستشاء أن تقول: قام زید الا مسا ....

... والمستثنى منه والمستنني جملة واحدة التك بعضا من كل وشيئا وميا...

الإستنان: والاسم المستنى منه موط ... aio e :: " |

... منزلة اسم مفاف. ألا ترى أنك إذا قلت: جاءني قومك إلا علىلا منهم ، فهو ممنزلة قولك: منهم ، فهو ممنزلة قولك: جاءبي أكثر قومك فكأنه حامق أكثر قومك فكأنه بالاضافة.

... بمنزلة اسم مضاف... ] فإذا قلت: جاءني قوملا الاقليلا ا سم سفاف لا يتم إلا اسم مفاف لا يتم إلا بالإصافة.

Ton Ya'ish also draws on the Usul for material on the points of issue between the Basrans and Kufans and this is seen in his discussion of the dispute over the regent of the khabar of inna and similar particles: 18

( <u>Usūl</u>) و ذهب الكوفيون إلى فإن قال قائل: إنّ إنّ الله <u>al-mufossal</u>) أن هذه الحروف لم تعمل إنها عملت في الاسم فقط في الخبر الرفع وإنها تصل الفنصبية وتركت الخبر...

ى الاسم لا فير. وإنَّها الخسر صرضوع ...

... على حال كما كان مو

... على حاله كما كان مع

...وهو فاسد و ذلا من الله فعدل الكوفيين، قمل له: الدليل على أنها

# ا هي الرافعة للخبر ...

... أن الاستداء قد زال وبه وبالمبتدأ كان يرتفع الخبر. وبالمبتدأ كان يرتفع الخبر. فيه ، ومع ذلك فإنّا وحدنا فيه ، ومع ذلك إنّا وجدنا كل ما عمل في المستدار كلما عمل في المبتدأ رفط ] عمل في هنبره , أو نصا عمل في هنبره . ألاترى إلى طلنت في المستدأ عملت في الخبر في المستدأ عملت في هبره ، وكدلك كان وأخواتها ...

...أن الابتداء قد زال وبه فلمّا زال الحامل بطل فلمّا زال الحامل بطل أن يكرن هذا محمولا أن يكون هذا محمولا محو فلندن وأضواتها الماعمات وأضواتها الماعمات وكذلك كان وأخواتها ...

...لما عملت في المبتدأ ... فكما جاز لك في المبتدأ ... عملت في المبتدأ والخبر جاز مع إنّ ، لا أ فرق بيسما في ذلك إلَّا أن الذي كان ستمأ بستصب بان و أخواتما.

In drawing on this passage from the Usul Ibn Ya'ish is led to contradict himself by reproducing a passage which contains a view held by certain Basrans which he has previously rejected. about because Ibn al-Sarraj amongst others holds the view that the khabar is put into the nominative by the joint action of ibtida and the mubtada and the above passage from the Usul confirms this. 19 However, Ibn Ya'ish rejects this view as unsound when he deals with the mubtada' and the khabar and prefers the view that it is <u>ibtida</u>' alone which is the regent, although acting through the medium of the mubtada'.

The fact that Ibn Ya'ish incorporates material from the <u>Usūl</u> into his own work is in itself a somewhat oblique testimony to the place given to Ibn al-Sarrāj's grammatical writings, although the use made of it by writers like al-Suyūţī and 'Abd al-Qādir al-Baghdādī, with all due acknowledgement, provides a more open testimony to the worth of the scholarship of Ibn al-Sarrāj. If Ibn Ya'īsh had drawn on the <u>Usūl</u> to a greater extent than he in fact does, it would have been possible to say that he did so simply to save himself work in writing his commentary on the <u>Mufagsal</u>, but the fact that his use of the <u>Usūl</u> is more selective leads one to assume that he made use of it because of its intrinsic merits. However, there are clearly passages which have been copied simply to suit the ease of the writer and not because they offer particularly fine grammatical analyses or material not readily obtainable elsewhere.

Although extensive reading of the <u>Mufassal</u> commentary of Ibn Ya'ish and the <u>Usul</u> specifically with a view to discovering further parallel passages will undoubtably yield further results, the preceding survey gives an idea of how Ibn Ya'ish is indebted to the <u>Usul</u>. 21

#### Notes to Chapter VIII

- In addition to the Cairo edition used in preparing this present study (see p. 25, n. 53 above), there is also Ibn Ya'ish, Sharh al-mufassal, 2 vols., ed. G. Jahn, Leipzig, 1882-6.
- See al-Suyuti, Ashbah, vol. i, pp. 28, 30, 49, 52, 55, 61, 62, 68, 70, 85, 89, 90, 95, 96, 99; vol. ii, pp. 22, 27, 33, 34, 37, 40, 45, 59, 61, 74; Maiman, s.v. Mufașsal, sharh Ibn Ya'ish.
- 3 E.I.<sup>2</sup>, s.v. Ibn Ya'ish (J. W. Fück).
- 4 Fleisch, pp. 41-2, n. 2.
- 5 G. Troupeau, Le commentaire d'al-Sirafi sur le chapitre 565 du Kitab de Sibawayhi, Arabica (5) 1958, p. 179.
- 6 See pp. 40-5 above.
- The al-Sarraj, vol. i, p. 258; The Ya'ish, vol. ii, p. 55. In 11. 9-10 of the passage from the Usul استفناه الفعل بفاله and this agrees with the text of the Mulassal and with The al-Sarraj's usual terminology as in the Usul, vol. i, pp. 58, 189, 342, 345. In 1. 14 of the passage from the Usul يكون has been changed to the تكون of the Mulassal.
- 8 Ibn Ya'ish, vol. ii, p. 55.
- g Ibn al-Sarraj, vol. i, p. 258; Ibn Ya°ish, vol. ii, p. 55.
- 10 Ibn Ya'ish, vol. ii, p. 55.
- Ibn al-Sarraj, vol. i, pp. 258-9; Ibn Ya'ish, vol. ii, pp. 55-6.
  In 1. 7 of the passage from the Usul the term أصرل which is absent from the printed text has been added to make sense of the passage.
- 12 Ton al-Sarraj, vol. i, p. 259; Ton Ya'ish, vol. ii, p. 57. In 1. 6 of the passage from the <u>Usul</u> the words أن الصنة are absent from the printed text.
- 13 Ibn al-Sarraj, vol. i, pp. 259-60; Ibn Ya'ish, vol. ii, p. 56.
- 14 Ibn al—Sarraj, vol. i, p. 264; Ibn Ya'ish, vol. ii, p. 56. In 1. 5 of the passage from the Usul أمدكا has been substituted for أمدكاً.
- 15 Ibn al-Sarraj, vol. i, pp. 342-3.
- 16 Ibn Ya<sup>c</sup>ish, wol. ii, pp. 75-6.
- 17 Ibn al-Sarraj, vol. i, p. 343; Ibn Ya'ish, vol. ii, p. 76.
- 18 Ibn al-Sarraj, vol. i, pp. 278-9; Ibn Ya'ish, vol. i, p. 102,
- 19 Ibn al-Sarraj, vol. i, p. 63. See also pp. 61-3 above.
- 20 Ibn Ya ish, vol. i, p. 85.

- 21 For further parallel passages observed in the preparation of this present study see:
  - a Ibn al-Sarrāj, vol. i, pp. 67-8 ...فبرالحبتدأ ... لتسند إليه ...

    Ibn Yaʿīsh, vol. 1, p. 87 ...فبرالحبتدأ ... لتسند أليه ...
  - / وإنما لزمت البه ... وجهه أوعيث. 9=118 The al-Sarraj, vol. i, pp. 118 وإنما لزمت البه ... عدنه أو وجهه. The Ya'ish, vol. vii, p. 148

  - d Ibn al-Sarrāj, vol. i, pp. 273-4 ...اكان بلتبس... ق مده اكال... Ibn Ya'ish, vol. ii, p. 73 .... ق مده اكال...

Although Ibn Ya'ish draws material from the <u>Usul</u> without reference to its source, he does on occasions mention the name of Ibn al-Sarraj when recounting the latter's views on various matters, e.g. <u>Sharh</u> <u>al-mufassal</u>, vol. i, pp. 22, 129; vol. ii, p. 54; vol. vii, p. 99; vol. viii, p. 3; vol. ix, p. 104.

#### SELECTED BIBLIOGRAPHY

Abu Hayyan Athir al-Din. Manhaj al-salik. New Haven, Connecticut, 1947.

Abu Tayyib al-Lughawi. Maratib al-nahwiyin. Cairo, 1955.

'Adimah, M. A. Introduction to the <u>Muqtadab</u>, see under al-Mubarrad.

Al-Asterabadhi. Sharh al-kafiyah. Turkey (Istanbul ?), n.d.

Al-Baghdadi, 'Abd al-Qadir. Khizanat al-adab. 4 vols. Beirut, n.d.

Encyclopaedia of Islam. 2nd ed.

Al-Farra'. Ma'anī l-Qur'an. 2 vols. published. Cairo, 1955- .

Fleisch, H. Traité de philologie arabe. 1 vol. published. Beirut, 1961- .

Hajji Khalifah. Kashf al-zumun. 2 vols. Turkey, 1941-3.

Howell, M. S. A grammar of the Classical Arabic Language. Introduction and 4 pts. Allahabad, 1883-91.

The al-Ambari. <u>Kitab al-ingaf</u>. Edited and with an Introduction by G. Weil. Leiden, 1913.

\_\_\_\_\_\_. <u>Luma' al-adillah</u>. Stockholm, 1963.

Ibn Khayr, Fahrasah, Saragossa, 1894-5.

Ibn al-Nadim. Fihrist. Leipzig, 1872.

Ibn al-Sarraj. <u>Usul fi l-nahw</u>. Ed. A. H. al-Fatli. 2 vols. published. Vol. i, Najaf, 1973. Vol. ii, Baghdad, 1973.

Ton Ya'ish. Sharh al-mufassal. 10 pts. Cairo, n.d.

Kahhalah, U. R. Mu'jam al-mu'allifin. 15 pts. Damascus, 1957-61.

Khalaf al-Ahmar. Muqaddimah fi l-nahw. Damascus, 1961.

Maiman, M. A. Iglid al-khizanah, Lahore, 1927.

Al-Mubarrad. Mugtadab. Ed. and with an Introduction by M. A. 'Adimah.

4 vols. Cairo, 1965-8.

Al-Qifti. Inbah al-ruwah. 3 pts. Cairo, 1950-5.

Sibawayh. Kitab. 2 vols. Bulaq, 1316-7 A.H.

Al-Sirafi. Akhbar al-nahwiyin al-Başriyin. Cairo, 1955.

Al-Suyuti. Ashbah wa-l-naza ir. 4 pts. Hyderabad, 1359-61 A.H.

. Iqtirah. Hyderabad, 1359 A.H.

Troupeau, G. Lexique-Index du Kitab de Sibawayhi. Paris, 1976.

Weil, G. Introduction to the Kitab al-insaf, see under Ibn al-Anbari.

Wright, W. A grammar of the Arabic language. 3rd ed. 2 vols. Reprinted Beirut, 1974.

Yaqut. Mu'jam al-udaba'. 7 vols. Leiden and London, 1907-31.

Al-Zajjaji. Idah fi 'ilal al-nahw. Cairo, 1959.

. Jumal fi l-nahw. Paris, 1957.

Al-Zamakhshari. Mufassal, see under Ibn Ya'ish.

Al-Zubaydi, Tabaqat al-nahwiyin, Cairo, 1954.